UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): July 19, 2021

LSB INDUSTRIES, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation) 1-7677 (Commission File Number) 73-1015226 (IRS Employer Identification No.)

3503 NW 63rd Street, Suite 500, Oklahoma City, Oklahoma (Address of principal executive offices) 73116 (Zip Code)

Registrant's telephone number, including area code: (405) 235-4546

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

Dere-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

Dere-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, Par Value \$.10	LXU	New York Stock Exchange
Preferred Stock Purchase Rights	N/A	New York Stock Exchange

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§ 230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§ 240.12b-2 of this chapter).

Emerging growth company \Box

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Item 1.01 Entry into a Material Definitive Agreement.

On July 19, 2021, LSB Industries, Inc. (the "Company") entered into a Securities Exchange Agreement (the "Agreement") with LSB Funding LLC (the "Holder"). Pursuant to the terms of the Agreement, the Holder has agreed with the Company to exchange all of the outstanding Series E-1 and Series F-1 Redeemable Preferred Stock ("Series E-1 Preferred" and "Series F-1 Preferred") held by the Holder into common stock, par value \$0.10, of the Company (the "Common Stock"), by dividing the liquidation preference payable therefore, in the case of the Series E-1 Preferred, and the redemption price payable therefore, in the case of the Series F-1 Preferred, by \$6.16 (collectively the "Transaction"). The exchange price of \$6.16 is equal to the volume weighted average price ("VWAP") per share of the Common Stock over the 30-day period ended July 16, 2021. In connection with the Transaction, existing Company common stockholders, and holders of the Series B 12% Cumulative, Convertible Preferred Stock and the Series D 6% Cumulative, Convertible Preferred Stock, in each case, on an as converted basis, will receive a special dividend of 0.30 shares of Common Stock for every 1 share of Common Stock, or on an as converted basis, as applicable, owned as of the record date (the "Special Dividend"); provided, however, that the aggregate number of shares of Common Stock issuable pursuant to the Transaction will be reduced by the number of shares of Common Stock that the Holder will receive in respect of the payment of the Special Dividend on the Common Stock held by the Holder on the record date. The terms and conditions of the Transaction were established by arm's length negotiations between an independent special committee of the board of directors (the "Board") of the Company (the "Special Committee") and the Holder. The Special Committee and the Board each unanimously approved the Transaction, including, in the case of the Board, for the purpose of rendering inapplicable to the Agreement and the Transaction the provisions of Section 20

Consummation of the Transaction is subject to customary closing conditions as well as (i) the approval of the Transaction and the Agreement by Company stockholders holding greater than 50% of the voting power of the outstanding shares of capital stock of the Company held by stockholders other than Holder and affiliates of the Holder (collectively, the "Disinterested Stockholders") and entitled to vote thereon, in a vote of the Disinterested Stockholders, (ii) the approval and adoption by holders of greater than 50% of the voting power of the outstanding shares of capital stock of the Company entitled to vote thereon of (x) an amendment to the Company's Amended and Restated Certificate of Incorporation, dated January 21, 1977, as amended on August 27, 1987 (the "Charter") to increase the total number of authorized shares of Common Stock to 150,000,000 shares and (y) the Transaction and the Agreement pursuant to Section 312.03 of the New York Stock Exchange Listed Company Manual, and (iii) the approval by holders of greater than 50% of the voting power of the outstanding shares of capital stock of the Company entitled to vote thereon and greater than 50% of the outstanding shares of Series E-1 Preferred entitled to vote thereon of an amendment to the Charter to permit the Transaction and to eliminate the right of the Series E-1 Preferred to participate in the Special Dividend (items (i) through (iii) collectively, the "Company Stockholder Approval"), (iv) termination of certain related party contracts between the Company and the Holder, (v) an amendment to the Board Representation and Standstill Agreement by and among the Company, the Holder, and the other parties thereto, dated December 4, 2015, as amended, to amend Section 3 thereof with respect to the share ownership limitations to give effect to the issuance of the shares of Common Stock in connection with the Transaction (the "Issued Shares") and to make certain other changes, (vi) approval of the Issued Shares for listing on the New York Stock Exchange subject to official notice of issuance and (vii) expiration or termination of any applicable waiting period under the Hart–Scott–Rodino Antitrust Improvements Act of 1976 relating to the Transaction.

The Agreement includes customary representations, warranties and covenants including, among others, a covenant to cause a special meeting of stockholders to be held to consider the Company Stockholder Approval. The Agreement provides for mutual indemnification by the Company and the Holder for breaches of representations, warranties and covenants, subject to certain customary limitations. In addition, the Company has agreed to provide indemnification to the Holder with respect to any losses, including costs and expenses, arising out of any litigation related to the Transaction. The Company has also agreed to certain interim operating covenants and, for so long as the Holder directly or indirectly beneficially owns 15% or more of the outstanding Common Stock, post-closing consent rights in favor of the Holder including, among others, limitations on certain (i) incurrences of indebtedness, (ii) capital expenditures, (iii) sales, leases, licenses, disposals or encumbrances of material properties or assets, (iv) acquisitions of, or agreements to acquire, businesses or assets, and agreements with respect to joint ventures, strategic alliances or partnerships, (v) amendments to the Charter and the Company's bylaws and changes to the size of Board and (vi) declarations or payments of dividends or distributions during the period between execution and closing under the Agreement. The Company convertible into, exercisable for or exchangeable into Common Stock, so long as the Holder directly or indirectly beneficially owns 20% or more of the outstanding Common Stock. The Company has also agreed, for so long as the Holder directly or indirectly beneficially owns 10% or more of the outstanding Common Stock, to provide the Holder with certain information concerning the business of the Company, including its financial condition and results of operations.

The Company has agreed to pay a termination fee to the Holder if the Board, acting on the recommendation of the Special Committee, authorizes the Company to enter into a definitive agreement for a bona fide written acquisition proposal for at least a majority of the outstanding shares of capital stock of the Company or at least a majority of the consolidated assets of the Company, for consideration consisting exclusively of cash or publicly traded equity securities (and for which financing, to the extent required by the person submitting the written acquisition proposal, is then fully committed), that (i) would be, if consummated, more favorable to the

Company's stockholders from a financial point of view than the Transactions and (ii) is not less likely to receive required governmental and other regulatory approvals on a timely basis than the Transactions and is otherwise reasonably likely to be consummated on the terms proposed (the "Superior Proposal") and the Company enters into such definitive agreement in connection with the termination of the Agreement. In such circumstances, the Company will (i) pay to the Holder in immediately available funds a fee of \$3,500,000 within one (1) business day following the termination of the Agreement, and (ii) upon closing of such Superior Proposal transaction cause the Series E-1 Preferred and the Series F-1 Preferred to be redeemed for cash based upon the liquidation preferences or redemption prices therefor. If the closing of such Superior Proposal transaction does not occur, the Company has agreed that, in lieu of effecting the redemption of the Series E-1 Preferred and the Series F-1 Preferred, the Company will pay to the Holder in immediately available funds an additional fee of \$7,000,000 one (1) business day following the date of termination of the definitive agreement concerning such Superior Proposal or the occurrence of the "outside date", "end date" or similar event in respect thereof and the Holder will also have the option, at its election, to reinstate the Agreement and to consummate the Transaction at the lower of \$6.16, the VWAP of the Common Stock for the 30 trading days ending on the date immediately prior to the date of reinstatement of the Agreement.

The foregoing description of the Agreement is qualified in its entirety by the full text of the Agreement filed as Exhibit 10.1 hereto and incorporated into this Current Report on Form 8-K by reference.

Item 3.02 Unregistered Sales of Equity Securities.

The information set forth in Item 1.01 is incorporated by reference in this Item 3.02. The Company expects the Issued Shares to be issued in reliance on the exemption from registration under Section 4(a)(2) of the Securities Act of 1933, as amended (the "Securities Act"), for a transaction by an issuer not involving any public offering within the meaning of Section 4(a)(2) thereunder, and/or Regulation D promulgated thereunder.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year

On July 19, 2021, the Board approved an amendment and restatement of the Company's bylaws (the "Second Amended and Restated Bylaws") to (i) consolidate the First Amendment, dated February 18, 2010, the Second Amendment, dated January 17, 2014, the Third Amendment, dated February 4, 2014, the Fourth Amendment, dated August 21, 2014, the Fifth Amendment, dated April 26, 2015, the Sixth Amendment, dated December 2, 2015, and the Seventh Amendment, dated December 22, 2015, to the Amended and Restated Bylaws of the Company, dated August 20, 2009, and (ii) add an exclusive forum bylaw. Specifically, new Article XII provides that, as a general rule, unless the Company consents in writing to the selection of an alternate forum, (i) the Court of Chancery of the State of Delaware will be the sole and exclusive forum for each of the following: (a) any derivative action or proceeding brought on behalf of the Company, (b) any action asserting a claim of breach of a fiduciary duty owed by any current or former director, officer, or other employee or stockholder of the Company, (c) any action asserting a claim arising pursuant to any provision of the Delaware General Corporation Law or bylaws of the Company, (d) any action to interpret, apply, enforce or determine the validity of these bylaws of the Company, or (e) any action asserting a claim governed by the internal affairs doctrine and (ii) the federal district courts of the United States shall be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act.

The foregoing description of the Company's Second Amended and Restated Bylaws is qualified in its entirety by the full text of the Second Amended and Restated Bylaws, which is filed as Exhibit 3.1 hereto and incorporated into this Current Report on Form 8-K by reference. A copy of the Second Amended and Restated Bylaws marked to show all changes resulting from the amendment and restatement is attached as Exhibit 3.2 to this Current Report on Form 8-K.

Item 7.01 Regulation FD Disclosure.

On July 20, 2021, the Company issued a press release announcing the Transaction and the Agreement. The press release is attached hereto as Exhibit 99.1 and incorporated by reference in this Item 7.01.

On July 20, 2021, the Company posted an investor presentation to its website. A copy of the investor presentation is attached hereto as Exhibit 99.2 and incorporated by reference in this Item 7.01.

The information in this Item 7.01 and in the accompanying Exhibit 99.1 and Exhibit 99.2 to this Current Report on Form 8-K is being furnished and shall not be deemed to be "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act") or otherwise subject to the liabilities of that Section, nor shall such information be deemed to be incorporated by reference in any registration statement or other document filed under the Securities Act or the Exchange Act except as otherwise stated in such filing.

The information in this Item 7.01 and in the accompanying Exhibit 99.1 and Exhibit 99.2 to this Current Report on Form 8-K contains certain forward-looking statements within the meaning of the Private Securities Litigation Reform Act of 1995. These forward-looking statements generally are identifiable by use of the words "may," "believe," "expect," "intend," "plan to," "estimate," "project" or similar expressions, and include but are not limited to: performance improvement and attaining targeted operating rates.

Investors are cautioned that such forward-looking statements are not guarantees of future performance and involve risk and uncertainties. Though we believe that expectations reflected in such forward-looking statements are reasonable, we can give no assurance that such expectation will prove to be correct. Actual results may differ materially from the forward-looking statements as a result of various factors. These and other risk factors are discussed in the Company's filings with the Securities and Exchange Commission (SEC), including those set forth under "Risk Factors" and "Special Note Regarding Forward-Looking Statements" in our Form 10-K for the year ended December 31, 2020 and, if applicable, our Quarterly Reports on Form 10-Q and our Current Reports on Form 8-K. All forward-looking statements included in this press release are expressly qualified in their entirety by such cautionary statements. We expressly disclaim any obligation to update, amend or clarify any forward-looking statement to reflect events, new information or circumstances occurring after the date of this press release except as required by applicable law.

The information in this Item 7.01 and in the accompanying Exhibit 99.1 and Exhibit 99.2 to this Current Report on Form 8-K is not intended to and shall not constitute an offer to buy or sell or the solicitation of an offer to buy or sell any securities, or a solicitation of any vote or approval, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made, except by means of a prospectus meeting the requirements of Section 10 of the Securities Act or an exemption from the registration requirements thereof.

In connection with the proposed transaction, the Company intends to file with the SEC a proxy statement for the special meeting of the stockholders and may also file other relevant documents with the SEC regarding the proposed transaction. This communication is not a substitute for the proxy statement or any other document that the Company may file with the SEC. The definitive proxy statement (if and when available) will be mailed to the stockholders. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THE PROXY STATEMENT AND ANY OTHER RELEVANT DOCUMENTS THAT MAY BE FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, CAREFULLY AND IN THEIR ENTIRETY IF AND WHEN THEY BECOME AVAILABLE BECAUSE THEY CONTAIN OR WILL CONTAIN IMPORTANT INFORMATION ABOUT THE COMPANY AND THE PROPOSED TRANSACTION. Investors and security holders will be able to obtain free copies of the proxy statement (if and when available) and other documents containing important information about the Company and the proposed transaction, once such documents are filed with the SEC through the website maintained by the SEC at http://www.sec.gov. Copies of the documents filed with the SEC by the Company may be obtained free of charge on the Company's website at www.lsbindustries.com or by contacting Michael Foster, General Counsel and Secretary by email at mfoster@lsbindustries.com or by phone at 405-510-3596.

The Company and certain of its directors and executive officers may be deemed to be participants in the solicitation of proxies in respect of the proposed transaction. Information about the directors and executive officers of the Company, including a description of their direct or indirect interests, by security holdings or otherwise, is set forth in the Company's proxy statement for its 2021 Annual Meeting of Stockholders, which was filed with the SEC on April 19, 2021, and the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2020, which was filed with the SEC on February 25, 2021. Other information regarding the participants in the proxy solicitations and a description of their direct and indirect interests, by security holdings or otherwise, will be contained in the proxy statement and other relevant materials to be filed with the SEC regarding the proposed transaction when such materials become available. Investors should read the proxy statement carefully when it becomes available before making any voting or investment decisions. You may obtain free copies of these documents from the Company using the sources indicated above.

ITEM 9.01 Financial Statements and Exhibits

(d) Exhibits.

Exhibit No.	Description
3.1	Second Amended and Restated Bylaws of LSB Industries, Inc.
3.2	Marked Second Amended and Restated Bylaws of LSB Industries, Inc.
10.1	Securities Exchange Agreement, dated July 19, 2021, by and between LSB Industries, Inc. and LSB Funding LLC
99.1	Press Release, dated July 20, 2021

- 99.2 <u>Investor Presentation</u>
- 104 Cover Page Interactive Data File (embedded within the XBRL document)

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

Date: July 20, 2021

LSB INDUSTRIES, INC.

By: <u>/s/ Michael J. Foster</u>

Michael J. Foster Executive Vice President and General Counsel

LSB INDUSTRIES, INC. (a Delaware Corporation)

SECOND AMENDED AND RESTATED BYLAWS

ARTICLE I Offices

<u>Section 1</u>. The principal office of the Corporation shall be in Oklahoma City, County of Oklahoma, State of Oklahoma, and the Corporation may also have offices at such other places as the Board of Directors may from time to time appoint or at such other places as the business of the Corporation requires.

ARTICLE II Seal

<u>Section 1</u>. The corporate seal shall be in such form as the Board of Directors may from time to time prescribe. Said seal may be used by causing it, or a facsimile thereof, to be impressed or affixed or reproduced or otherwise.

ARTICLE III Stockholders; Business to be Conducted at Annual or Special Meeting of Stockholders; and Stockholder Access to Corporation's Proxy Statement

Section 1. <u>Place</u>. All meetings of the stockholders shall be held in Oklahoma City, Oklahoma, or at such other place as the directors may designate.

Section 2. <u>Annual Meeting</u>. An annual meeting of stockholders shall be held each calendar year on such date and at such time as shall be designated from time to time by the Board of Directors and stated in the notice of the meeting or in a duly executed waiver of notice of such meeting. At such meeting, the stockholders shall elect directors and transact such other business as may properly be brought before the meeting.

Section 3. <u>Quorum</u>. The holders of record of a majority of the stock issued and outstanding, and entitled to vote thereat, present in person, or represented by proxy, shall be requisite and shall constitute a quorum at all meetings of the stockholders for the transaction of business, except as otherwise provided by law, by the Certificate of Incorporation or by these Bylaws, but in the absence of a quorum the holders of record, present in person or represented by proxy at such meeting shall have power to adjourn the meeting from time to time, without notice other than announcement at the meeting, until the requisite amount of voting stock shall be present. At such adjourned meeting at which the requisite amount of voting stock shall be represented, any business may be transacted which might have been transacted at the meeting as originally notified.

Section 4. <u>Voting</u>; <u>Proxies</u>. Except as otherwise provided by the laws of the State of Delaware or the Certificate of Incorporation of the Corporation or these Bylaws:

(a) At every meeting of the stockholders every shareholder having the right to vote shall be entitled to one vote for each share of capital stock having voting rights held by him.

(b) Each stockholder entitled to vote at a meeting of stockholders or to express consent or dissent to corporate action in writing without a meeting may authorize another person or persons to act for him by proxy, but no such proxy shall be voted or acted upon after three years from its date, unless the proxy provides for a longer period.

(c) Each matter properly presented to any meeting shall be decided by a majority of the votes cast on the matter.

(d) Election of directors and the vote on any other matter presented to a meeting need not be by written ballots, but written ballots may be used if ordered by the chairman of the meeting or if so requested by any stockholder present or represented by proxy at the meeting entitled to vote in such election or on such matter, as the case may be.

Section 5. <u>Notice of Meeting</u>. For each meeting of stockholders written notice shall be given stating the place, date and hour, and, in the case of a special meeting, the purpose or purposes for which the meeting is called and, if the list of stockholders required by Section 6 is not to be at the place of said meeting at least 10 days prior to the meeting, the place where said list will be. Except as otherwise provided by Delaware law, the written notice of any meeting shall be given not less than 10 nor more than 60 days before the date of the meeting to each stockholder entitled to vote at such meeting. If mailed, notice shall be deemed to be given when deposited in the United States mail, postage prepaid, directed to the stockholder at his address as it appears on the records of the Corporation.

Section 6. List of Stockholders Entitled to Vote. At least 10 days before every meeting of stockholders a complete list of the stockholders entitled to vote at the meeting, arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder, shall be prepared and shall be open to the examination of any stockholder for any purpose germane to the meeting, during ordinary business hours, for a period of at least 10 days prior to the meeting, either at a place within the city where the meeting is to be held, which place shall be specified in the notice of the meeting, or, if not so specified, at the place where the meeting is to be held. Such list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present.

Section 7. <u>Special Meetings</u>. A special meeting of stockholders may be called at any time by the Chairman or by a majority of the directors then in office, and shall be called by the Chairman upon receipt of a written request to do so specifying the matter or matters, appropriate for action at such meeting, proposed to be presented at the meeting and signed by holders of record of two-thirds of the shares of stock that would be entitled to be voted on such matter or matters if the meeting was held on the day such request is received and the record date for such meeting was the close of business on the preceding day. Any such meeting shall be held at such time and at such place, within or without the State of Delaware, as shall be determined by the body or person calling such meeting and as shall be stated in the notice of such meeting."

Section 8. <u>Chairman and Secretary at Meeting</u>. At each meeting of stockholders, the Chairman of the Board of Directors or, in the absence or inability to serve by the Chairman of the Board of Directors, the Vice Chairman of the Board of Directors or, in the absence or inability to serve by both the Chairman of the Board of Directors and the Vice Chairman of the Board of Directors, the President or, in the absence or inability to serve by the Chairman of the Board of Directors, Vice Chairman of the Board of Directors, the President or, in the absence or inability to serve by the Chairman of the Board of Directors, Vice Chairman of the Board of Directors and the President, the person designated in writing by the President or, if no person is so designated by the Board of Directors shall preside as Chairman of the meeting; if no person is so designated, then the Board of Directors shall choose a Chairman by plurality vote. The Secretary or in his absence a person designated by the Chairman of the meeting shall act as Secretary of the meeting.

Section 9. <u>Adjourned Meetings</u>. A meeting of stockholders may be adjourned to another time or place as provided in Sections 3 or 4(d) of this Article III. Unless the Board of Directors fixes a new record date, stockholders of record for an adjourned meeting shall be as originally determined for the meeting from which the adjournment was taken. If the adjournment is for more than 30 days, or if after the adjournment, a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote. At the adjourned meeting any business may be transacted that might have been transacted at the meeting as originally called.

Section 10. Consent of Stockholders in Lieu of Meeting.

10.1 <u>Action by Written Consent</u>. Any action which is required to be or may be taken at any annual or special meeting of stockholders of the Corporation may be taken without a meeting, without prior notice and without a vote, if consents in writing, setting forth the action so taken, shall have been signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or to take such action at a meeting at which all shares entitled to vote thereon were present and voted; provided however, that prompt notice of the taking of the corporate action without a meeting and by less than unanimous written consent shall be given to those stockholders who have not consented in writing.

10.2 Determination of Record Date of Action by Written Consent. In order to inform the Corporation's stockholders and the investing public in advance that a record date for action by consent will occur and to comply with the procedures contained in the New York Stock Exchange (or such other exchange on which the Corporation's securities are listed for trading) policies and rules, the record date for determining stockholders entitled to express consent to corporate action in writing without a meeting is fixed by the Board of Directors of the Corporation pursuant to Section 213 of the Delaware General Corporation Law as follows: The Board of Directors shall set as the record date the 10th day after (i) any stockholder of record seeking to have the stockholders authorize or take corporate action by written consent without a

meeting shall, by written notice to the Secretary which may be given by telex or telecopy, advise the Corporation of the corporate action proposed for which consents will be sought and request from the Board of Directors a record date unless a later date is specified by such stockholder, or (ii) the Board of Directors determines that the Corporation should seek corporate action by written consent, unless a later record date is specified in the resolution of the Board of Directors containing such determination. In the event that the record date set as provided falls on a Saturday, Sunday or legal holiday, the record date shall be the first day next following such date that is not a Saturday, Sunday or legal holiday. Any record date determined pursuant to this Subsection 10.2 shall be announced by a press release prior to the opening of trading on the New York Stock Exchange (or such other exchange on which the Corporation's securities are listed for trading) on the next trading day after a request for a record date pursuant to clause (i) above is received by the Secretary or a Board of Directors' determination pursuant to clause (ii) above.

10.3 Duration and Revocation of Consents. In order that the Corporation's stockholders shall have an opportunity to receive and consider the information germane to an informed judgment as to whether to give a written consent and in accordance with the procedures contained in the New York Stock Exchange (or such other exchange on which the Corporation's securities are listed for trading) policies and rules, the stockholders of the Corporation shall be given at least 20 days from the record date to give or revoke written consents. Consents to corporate action shall be valid for a maximum of 60 days after the record date. Consents may be revoked by written notice (i) to the Corporation, (ii) to the stockholder or stockholders soliciting consents or soliciting revocations in opposition to action by consent proposed by the Corporation (the "Soliciting Stockholders"), or (iii) to a proxy solicitor or other agent designated by the Corporation of the Soliciting Stockholder(s).

10.4 <u>Retention and Duties of Inspectors of Election</u>. Within two business days after receipt of a request by a stockholder for the setting of a record date or a determination by the Board of Directors that the Corporation should seek corporate action by written consent, as the case may be, the Secretary of the Corporation shall engage nationally recognized independent inspectors of elections for the purpose of performing a ministerial review of the validity of the consents and revocations. The inspectors shall review all consents and revocations, determine whether the requisite number of valid and unrevoked consents has been obtained to authorize or take the action specified in the consents, and forthwith certify such determination for entry in the records of the Corporation kept for the purpose of recording the proceedings of meetings of stockholders. The cost of retaining inspectors of elections shall be borne by the party proposing the action by consent.

10.5 Procedures for Counting and Challenging Consents. Consents and revocations shall be delivered to the inspectors upon receipt by the Corporation, the Soliciting Stockholders or their proxy solicitors or other designated agents. As soon as consents and revocations are received, the inspectors shall review the consents and revocations and shall maintain a count of the number of valid and unrevoked consents. The inspectors shall keep such count confidential and shall not reveal the count to the Corporation, the Soliciting Stockholders or their representatives. As soon as practicable after the earlier of (i) 60 days after the record date for the consents or (ii) a request therefore by the Corporation or the Soliciting Stockholders (whichever is soliciting consents) made after expiration of the period for giving or revoking consents under

Subsection 10.3 above, notice of which request shall be given to the party opposing the solicitation of consents, which request shall state that the Corporation or Soliciting Stockholder(s) (as the case may be) in good faith believe that it or they have received the requisite number of valid and unrevoked consents to authorize or take the action specified in the consents, the inspectors shall issue a preliminary report to the Corporation and the Soliciting Stockholders stating:

- (i) The number of valid consents;
- (ii) The number of valid revocations;
- (iii) The number of valid and unrevoked consents;
- (iv) The number of invalid consents;
- (v) The number of invalid revocations;
- (vi) Whether, based on their preliminary count, the requisite number of valid and unrevoked consents has been obtained to authorize or take the action specified in the consents.

Unless the Corporation and the Soliciting Stockholder(s) shall agree to a shorter or longer period, the Corporation and the Soliciting Stockholder(s) shall have 48 hours to review the consents and revocations and to advise the inspectors and the opposing party in writing as to whether they intend to challenge the preliminary report of the inspectors. If no written notice of an intention to challenge the preliminary report is received within 48 hours after the inspector's issuance of the preliminary report, the inspectors shall issue to the Corporation and the Soliciting Stockholder(s) their final report containing the information from the inspectors' determination with respect to whether the requisite number of valid and unrevoked consents was obtained to authorize and take the action specified in the consents. If the Corporation or the Soliciting Stockholder(s) issue written notice of an intention to challenge the inspectors' preliminary report within 48 hours after the issuance of that report, a challenge session shall be scheduled by the inspectors as promptly as practicable. A transcript of the challenge session shall be recorded by a certified court reporter. Following completion of the challenge session, the inspectors shall as promptly as practicable issue their final report to the Corporation and the Soliciting Stockholder(s) containing the information included in the preliminary report, plus all changes in the vote totals as a result of the challenges and a certification of whether the requisite number of valid and unrevoked consents was obtained to authorize or take the action specified in the consents after the action specified in the consents.

10.6 <u>Notice of Results</u>. The Corporation shall give prompt notice to the stockholders of the results of any consent solicitation or the taking of the corporate action without a meeting and by less than unanimous written consent.

Section 11. <u>Fixing of Record Date</u>. In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, or to express consent to corporate action in writing without a meeting, or entitled receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix, in advance, a record date, which shall not be more than 60 nor less than 10 days before the date of such meeting, nor more than 60 days prior to any other action. If no record date is fixed, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held; the record date for determining stockholders entitled to express consent to corporate action in writing without a meeting, when no prior action by the Board of Directors is necessary, shall be the day on which the first written consent is expressed; and the record date for determining stockholders for any other purpose shall be at the close of business on the day on which the first written consent is expressed; and the record date for determining stockholders for any other purpose shall be at the close of business on the day on which the first written consent is expressed; and the record date for determining stockholders for any other purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto.

Section 12. <u>Business to be Conducted at the Annual or Special Meeting of the Stockholders; Notice of Proposals</u>. At any annual meeting of the stockholders, only such business shall be conducted as shall have been brought before the meeting (i) by or at the direction of the Board of Directors, or (ii) by any stockholder of the Corporation who is entitled to vote with respect thereto and who: (a) was a stockholder of record (and, with respect to any beneficial owner, if different, on whose behalf such business as proposed, only if such beneficial owner was the beneficial owner of shares of the Corporation) both at the time of giving notice provided for in this Section 12 and at the time of the meeting; and (b) complies with the notice procedures set forth in this Section 12.

Except for proposals properly made in accordance with Rule 14a-8 under the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder (as so amended and inclusive of such rules as regulations, the "Exchange Act") and included in the notice of the meeting given by or at the direction of the Board of Directors, the foregoing clause (ii) will be the exclusive means for a stockholder to propose business to be brought before the annual meeting of stockholders.

For business to be properly brought before the annual meeting by a stockholder, the Proposing Person (as defined below) must have given timely notice thereof in writing to the Secretary of the Corporation. The Proposing Person's notice will be timely if delivered or mailed to and received at the principal executive offices at the Corporation not less than 120 nor more than 150 days before the date on which the Corporation first mailed its proxy materials for the prior year's annual meeting of stockholders; provided however, that if the date of the annual meeting is more than 30 days before or more than 60 days after such date, notice by the stockholder timely must be so delivered, or mailed and received not later than the 90th day prior to such annual meeting, or if later, the 10th day following the date on which the public disclosure of the date of such annual meeting was such made. Any adjournment of an annual meeting or the announcement hereof will not commence a new time period for giving the notice described above.

The Proposing Person's notice to the Secretary shall set forth as to each matter such stockholder proposes to bring before the annual meeting, the following:

- (i) the name and address, as they appear on the Corporation books, of the stockholder proposing such business;
- the class or series and number of shares of the Corporation's securities that are, directly or indirectly, owned of record or beneficially owned (within the meaning of Rule 13d-3 under the Exchange Act) by the Proposing Person, except the Proposing Person will be deemed to beneficially own any shares or class or series of the Corporation's securities which the Proposing Person has a right to acquire beneficially ownership at any time in the future (collectively, the "Stockholder Information");
- (iii) as to each item of business that the stockholder proposes to bring before the annual meeting, (A) a reasonably brief description of the business desired to be brought before the annual meeting, the reasons for conducting such business at the annual meeting and any material interest in such business of each Proposing Person, (B) the text of the proposal or business (including the text of any resolutions proposed for consideration), and
- (iv) a reasonably detailed description of all agreements, arrangements and understandings, oral or in writing (x) between or among any of the Proposing Persons or (y) between or among any Proposing Person and any other record or beneficial holder of the shares of any class or series of the Corporation (including their names) in connection with the proposal of such business by such stockholder or (z) between or among any Proposing Person and any other persons or entities (including their names) acting in concert with the Proposing Person.

For purposes of this Section 12, the term "Proposing Person" shall mean (i) the stockholder providing the notice of business proposed to be brought before an annual meeting, (ii) the beneficial owner or beneficial owners, if different, on whose behalf the notice of the business proposed to be brought before the annual meeting is made, and (iii) any affiliate or associate (each within the meaning of Rule 12b-2 under the Exchange Act for purposes of these Bylaws) of such stockholder or beneficial owner; and (iv) any material interest of such stockholder with respect to such business.

Notwithstanding anything in these Bylaws to the contrary, no business (other than nominations of directors, which must be made in compliance with, and shall be exclusively governed by, Article III, Section 13 of these Bylaws) shall be brought before or conducted at the annual meeting except in accordance with the provisions of this Section 12. The officer of the Corporation or other person presiding over the annual meeting shall, if the facts so warrant, determine and declare to the meeting that business was not properly brought before the meeting in accordance with the provisions of this Section 12 and, if he should so determine, he shall so declare to the meeting and any such business so determined to be not properly brought before the meeting shall not be transacted.

A stockholder providing notice of business proposed to be brought before an annual meeting shall further update and supplement such notice, as needed, so that the information provided or required to be provided in such notice pursuant to this Section 12 shall be true and correct as of the record date for the meeting and as of the date that is 10 business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary of the Corporation at the principal executive offices of the Corporation not later than five business days after the record date for the meeting (in the case of the update and supplement required to be made as of the record date), and not later than eight business days prior to the date for the meeting (in the case of the update and supplement required to be made as of 10 business days prior to the meeting).

This Section 12 is expressly intended to apply to any business proposed to be brought before an annual meeting of stockholders; except this Section 12 shall not apply to any proposal made pursuant to Rule 14a-8 of the Exchange Act, or to the nomination of persons for election to the Corporation's Board of Directors at a meeting of stockholders at which directors are to be elected which shall be governed by Article III, Section 13 of these Bylaws. In addition to the requirements of this Section 12 with respect to any business proposed to be brought before an annual meeting, each Proposing Person shall comply with all applicable requirements of the Exchange Act with respect to any such business. Nothing in this Section 12 shall be deemed to affect the rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act.

For purposes of these Bylaws, "public disclosure" shall mean disclosure in a press release reported by a national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Sections 13, 14 or 15(d) of the Exchange Act.

The officer of the Corporation or other person presiding at the meeting shall, if the facts so warrant, determine that business was not properly brought before the meeting in accordance with the procedures set forth in this Section 12, and, if he or she should so determine, he or she shall so declare to the meeting and any such business not properly brought before the meeting shall not be transacted.

At any special meeting of the stockholders, only such business shall be conducted as shall have been brought before the meeting by or at the direction of the Board of Directors.

Section 13. Election to the Board of Directors.

13.1 Only persons who are nominated in accordance with the procedures set forth in these Bylaws shall be eligible for election as directors of the Corporation. Nominations of persons for election to the Board of Directors of the Corporation may be made at a meeting of stockholders at which directors are to be elected only:

- (i) By or at the direction of the Board of Directors; or
- 8

(ii) By any stockholder of the Corporation entitled to vote for the election of directors at the meeting who complies with the notice procedures set forth in Subsection 13.2 below.

The foregoing clause (ii) will be the exclusive means by which a stockholder may nominate a person for election to the Board of Directors.

13.2 Nominations of election as a director of the Corporation, other than those made by or at the direction of the Board of Directors, shall be made by timely notice in writing to the Secretary of the Corporation. To be timely, a stockholder's notice shall be delivered or mailed to and received at the principal executive offices of the Corporation not less than 120 nor more than 150 days prior to the anniversary date of the Corporation's immediately preceding annual meeting of stockholders; provided, however, that in the event the date of the annual meeting is more than 30 days before or more than 60 days after such date, notice by the stockholder to be timely must be so delivered, or mailed and received not later than the 90th day prior to such annual meeting, or if later, the 10th day following the date on which the public disclosure of the date of such annual meeting was so made. Any adjournment of an annual meeting or the announcement hereof will not commence a new time period for giving the timely notice described above. Such stockholder's notice shall set forth:

- (i) As to each person whom such stockholder proposes to nominate for election or reelection as a director, (x) all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934, as amended (including such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected, and (y) a representation that such a person shall also provide any other information reasonably requested by the Corporation within 10 business days after such request); and
- (ii) As to the stockholder giving the notice (x) the name and address, as they appear on the Corporation's books, of such stockholder, and (y) the class and number of shares of the Corporation's voting capital stock that are beneficially owned by such stockholder.

At the request of the Board of Directors, any person nominated by the Board of Directors for election as a director shall furnish to the Secretary of the Corporation that information required to be set forth in a stockholder's notice of nomination which pertains to the nominee.

No person shall be eligible for election as a director of the Corporation unless nominated in accordance with the provisions of this Section 13, and, if the stockholder desires or requests access to the Corporation's Proxy Statement with respect to the election of a director, Article III, Section 14 of these Bylaws.

The officer of the Corporation or other person presiding at the meeting shall, if the facts so warrant, determine that a nomination was not made in accordance with such provisions and, if he or she should so determine, he or she shall so declare to the meeting and the defective nomination shall be disregarded.

Section 14. Stockholder Access to Corporation's Proxy Statement.

(a) Whenever the Corporation solicits proxies with respect to an election of directors at an annual meeting (an "Election"), it shall include in its proxy statement and on its proxy card, in addition to individuals nominated by the Board of Directors, up to the Permitted Number of individuals nominated in compliance with these Bylaws by one or more Eligible Stockholders. Any Eligible Stockholder seeking to have its nominee included in the Corporation's proxy statement and on the Corporation's proxy card shall comply with all provisions of these Bylaws otherwise applicable to shareholder nominations and furnish to the Secretary of the Corporation, no later than the last day on which stockholder nominations for consideration in the Election may be made under Article III, Section 13.2 \Box of these Bylaws (the "Advance Notice Date"),

- (i) the information set forth in Sections 13.2 of these Bylaws,
- (ii) the written undertakings described in subsections (d) and (e) below, and
- (iii) any accompanying statement from the Eligible Stockholder to be included in the Corporation's proxy statement, which statement in order to be so included shall not exceed 500 words and must fully comply with Section 14 of the Exchange Act and the rules and regulations promulgated thereunder, including without limitation Rule 14a-9.
- (b) For purposes of this Section:
 - (i) The "Permitted Number" means one seat on the Corporation's Board of Directors to be filled in the Election; provided, however, that the Permitted Number shall be reduced, but not below zero, by the sum of (i) number of such director candidates for which the Corporation shall have received from Eligible Stockholders by the Advance Notice Date one or more valid stockholder notices nominating director candidates, and (ii) the number of directors in office and serving in the class of directors to be considered at the Election for whom access to the Corporation's proxy materials was provided pursuant to this Section ("Access Director"), other than any who has served as a director continuously for at least six years. In no event will the number of nominees nominated by an Eligible Stockholder for which access to the Corporation's proxy materials may be provided in the Election plus the number of Access Directors serving on the Board at the time of the Election exceed 25% of the total number of directors that shall constitute the whole board.
 - (ii) An "Eligible Stockholder" means a stockholder of the Corporation that, together with its Affiliates, has continuously held Beneficial Ownership and Economic Interest of not less than the Required Interest for at least one year preceding the Advance Notice Date, and that complies with all applicable provisions of these Bylaws.
 - 10

- (iii) "Beneficial Ownership" means the power to vote or direct the voting of, or to dispose or direct the disposition of, the securities in question.
- (iv) An "Economic Interest" in a security means having or sharing the opportunity, directly or indirectly, to profit or share in any profit (or loss) derived from a transaction in the security, including through options, swaps or other derivative securities or synthetic arrangements.
- (v) "Independent" with respect to a nominee for director pursuant to this Section 14 shall mean (a) that the nominee would be considered an independent director in accordance with the listing standards of the principal U.S. securities market in which the common stock of the Corporation trades or, if no such listing standards are applicable at the time, in accordance with the standards used by the Board of Directors or a duly authorized committee thereof in determining and disclosing the independence of the Corporation's directors in accordance with the rules of the SEC and (b) the nominee is not an employee or officer of, or consultant to, the Eligible Stockholder or any of its Affiliates and has no other material association, by agreement, understanding or familial or other relationship, with the Eligible Stockholder or any of its Affiliates.
- (vi) A "Disqualified Repeat Nominee" in respect of an election shall mean an individual as to whom access to the Corporation's proxy materials for the immediately preceding election was provided and who (i) withdrew from or became ineligible or unavailable for election at the meeting, or (ii) received at such meeting votes in favor of his or her election representing less than 50% of the total votes cast for or withheld from his or her election.
- (vii) The "Required Interest" means 5% of the voting power of the outstanding voting securities of the Corporation entitled to vote in the Election, based upon the number of outstanding voting securities of the Corporation most recently disclosed prior to the Advance Notice Date by the Corporation in a filing with the Securities and Exchange Commission.
- (viii) "Affiliate" of a specified person means a person that, directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with, the specified person, and, with respect to any investment company (as defined in the Investment Company Act of 1940, whether or not exempt from registration thereunder), shall also include all other investment companies managed by the same investment adviser or any of its Affiliates.

(c) Subject to the following sentence and any undertaking previously provided by an Eligible Stockholder pursuant to Section 14(d) below, each Eligible Stockholder, together with its Affiliates, may nominate one, and not more than one, individual under this Section for inclusion in the Corporation's proxy statement and on its proxy card. If the Corporation shall receive more than the Permitted Number of proposed nominations from Eligible Stockholders in compliance with these Bylaws, then the nominee nominated by the Eligible Stockholder possessing the largest Beneficial Ownership of voting securities of the Corporation as of the Advance Notice Date shall be the only nominee for purposes of this Section 14.

(d) Any Eligible Stockholder nominating an individual for director in accordance with this Section shall execute and deliver to the Corporation no later than the Advance Notice Date an undertaking, in a form to be provided by the Secretary of the Corporation, that it will, and will cause its Affiliates to,

- (i) not sell or otherwise dispose of its Beneficial Ownership and Economic Interest of voting securities of the Corporation so as to reduce the Beneficial Ownership and Economic Interest held by such Eligible Stockholder, together with its Affiliates, below the Required Interest on or prior to the date of the Election (and representing that they have no present intention of reducing, within one year following the Election, their aggregate Beneficial Ownership and Economic Interest below the greater of (x) the Required Interest and (y) 75% of their aggregate Beneficial and Economic Interest as of the Advance Notice Date),
- (ii) comply with the provisions of the Corporation's Certificate of Incorporation and Bylaws and all laws and regulations relating to the accompanying statement submitted by the Eligible Stockholder and any solicitation or communications with stockholders of the Corporation in connection with such nomination,
- (iii) indemnify the Corporation and its agents and representatives in respect of any and all liabilities that may arise out of the accompanying statement submitted by the Eligible Stockholder or any solicitation or communications with stockholders of the Corporation by such Eligible Stockholder, its Affiliates or their respective agents or representatives in connection with such nomination, including as a result of any violation of law or regulation by such Eligible Stockholder, its Affiliates or their respective agents or representatives in connection therewith,
- (iv) not use any proxy card other than the Corporation's proxy card in soliciting stockholders in connection with the matters to be voted on at the meeting at which the Election is held,
- (v) file all solicitation materials used by it or on its behalf with the Securities and Exchange Commission under cover of Schedule 14A promulgated under the Exchange Act, and

(vi) for a period of one year from the date of the Election, not (x) nominate any individual to be a director of the Corporation or conduct any solicitation with respect to an election for directors of the Corporation other than with respect to the Election and in accordance with this Section, or (y) acquire or propose to acquire Beneficial Ownership of or an Economic Interest in any voting securities of the Corporation such that such Eligible Stockholder, together with its Affiliates, would have aggregate Beneficial Ownership of, and/or an Economic Interest in, more than the greater of (I) 10% of the voting power of the outstanding voting securities of the Corporation or (II) an additional 5% of the voting power of the Corporation's outstanding voting securities in excess of the aggregate Beneficial Ownership and Economic Interest held by such Eligible Stockholder, together with its Affiliates, as of the Advance Notice Date (the "Aggregate Beneficial Ownership"); provided that the Aggregate Beneficial Ownership shall not equal or exceed the amount that would equal or exceed the beneficial ownership threshold necessary to trigger the Corporation preferred share rights plan as may be in effect during such time.

(e) Any Eligible Stockholder nominating an individual for director in accordance with this Section shall also deliver to the Corporation no later than the Advance Notice Date a signed undertaking of its nominee agreeing that he or she will tender his or her resignation from the Board of Directors if

- (i) any of the information provided to the Corporation by the Eligible Stockholder or the nominee pursuant to this Bylaw is determined to be inaccurate in any material respect, or
- (ii) the Eligible Stockholder or any of its Affiliates shall breach their obligations under the undertakings described in subsection(d) above in any material respect.

(f) The Nominating and Governance Committee shall consider a nomination pursuant to this Section 14, and shall determine if the Access Nominee is Independent and may, in its discretion, make a recommendation to the Board of Directors as to whether the Access Nominee should be nominated by the Board of Directors for election at the Annual Meeting of Stockholders.

If the Board of Directors nominates an Access Nominee as part of the Board's slate of nominees, the Notice of Access will be deemed withdrawn and the former Access Nominee shall be presented to the stockholders in the same manner as any other nominee of the Board of Directors. If the Board of Directors does not so nominate the Access Nominee, access to the Corporation's proxy materials shall be provided in accordance with the terms and subject to the conditions of this Section.

The Board of Directors or a committee thereof may adopt such rules or guidelines for applying the provisions of this Section as it determines are appropriate. These may include timing and other such adjustments as may be appropriate in the event an Access Nominee for whom Notice of Access has been provided becomes unavailable or unwilling to serve or becomes ineligible.

(g) This Section shall provide the exclusive method for stockholders to include nominees for director in the Corporation's proxy statement and on the Corporation's proxy card.

ARTICLE IV Directors

Section 1. Number, Term, Qualifications and Vacancies. The property, business and affairs of the Corporation shall be managed by or under the direction of its Board of Directors.

The number of directors that shall constitute the whole Board of Directors may be fixed from time to time pursuant to a resolution adopted by a vote of two-thirds of the entire Board of Directors and may consist of no fewer than three nor more than fourteen members. The directors shall be divided into three classes. Each class shall consist, as nearly as possible, of one-third of the whole number of the Board of Directors. At each annual election of the successors to the class of directors whose terms have expired in that year shall be elected to hold office for a term of three years. Each director elected shall hold office until his successor is elected and qualified or until his earlier resignation or removal. Directors and officers need not be stockholders.

Vacancies and newly created directorships resulting from any increase in the authorized number of directors shall be filled only by a majority of the directors then in office, although less than a quorum, or by the sole remaining director. Each director chosen to fill a vacancy or newly created directorship shall hold office until the next election of the class for which such directors shall have been chosen and until his successor is duly elected and qualified or until his earlier resignation or removal.

Section 2. Offices and Books. The directors may have one or more offices, and keep the books of the Corporation at the offices of the Corporation in Oklahoma City, Oklahoma, or at such other places as they may from time to time determine.

Section 3. Resignation. Any director of the Corporation may resign at any time by giving written notice of such resignation to the Board of Directors, the Chairman of the Board of Directors, the President or the Secretary of the Corporation. Any such resignation shall take effect at the time specified therein or, if no time be specified, upon receipt thereof by the Board of Directors, or one of the above-named officers; and, unless specified therein, the acceptance of such resignation shall not be necessary to make it effective. When one or more directors shall resign from the Board of Directors, such vacancy shall be filled only by a majority of the directors then in office, although less than a quorum, or by the sole remaining director. Each director so chosen shall hold office until the next election of the class for which such director shall have been chosen and until his successor is duly elected and qualified or until his earlier resignation or removal.

Section 4. <u>Removal</u>. Any one or more directors may be removed only for cause by the vote or written consent of the holders of a majority of the issued and outstanding shares of stock of the Corporation entitled to vote for the election of all directors. For purposes of this Article IV, Section 4, cause for removal shall be deemed to exist only if the director whose removal is proposed has been convicted of a felony by a court of competent jurisdiction or has been adjudged by a court of competent jurisdiction to be liable for intentional misconduct or knowing violation of law in the performance of such director's duty to the Corporation and, in each case, such adjudication is no longer subject to direct appeal.

Section 5. <u>Regular and Annual Meetings</u>; <u>Notice</u>. Regular meetings of the Board of Directors shall be held at such time and at such place, within or without the State of Delaware, as the Board of Directors may from time to time prescribe. No notice need be given of any regular meeting and a notice, if given, need not specify the purposes thereof. A meeting of the Board of Directors may be held without notice immediately after an annual meeting of stockholders at the same place as that at which such annual meeting of stockholders was held.

Section 6. <u>Special Meetings</u>; <u>Notice</u>. A special meeting of the Board of Directors may be called at any time by the Chairman or a majority of the directors then in office. Any such meeting shall be held at such time and at such place, within or without the State of Delaware, as shall be determined by the body or person calling such meeting. Notice of such meeting stating the time and place thereof shall be given (a) by deposit of the notice in the United States mail, first class, postage prepaid, at least three days before the day fixed for the meeting addressed to each director at his address as it appears on the Corporation's records or at such other address as the director may have furnished the Corporation for that purpose, or (b) by delivery of the notice similarly addressed for dispatch by telegraph, cable or radio or by delivery of the notice by telephone or in person, in each case at least two days before the time fixed for the meeting.

Section 7. <u>Presiding Officer and Secretary at Meetings</u>. Each meeting of the Board of Directors shall be presided over by the Chairman of the Board of Directors or in his absence by the President or if neither is present by such member of the Board of Directors as shall be chosen by the meeting. The Secretary, or in his absence an Assistant Secretary, shall act as secretary of the meeting, or if no such officer is present, a secretary of the meeting shall be designated by the person presiding over the meeting.

Section 8. <u>Quorum</u>. A majority of the whole Board of Directors shall constitute a quorum for the transaction of business, but in the absence of a quorum, a majority of those present (or if only one be present, then that one) may adjourn the meeting, without notice other than announcement at the meeting, until such time as a quorum is present. Except as otherwise required by the Certificate of Incorporation or these Bylaws, the vote of the majority of the directors present at a meeting at which a quorum is present shall be the act of the Board of Directors.

Section 9. <u>Meeting by Telephone</u>. Members of the Board of Directors or of any committee thereof may participate in meetings of the Board of Directors or of such committee by means of conference telephone or similar communications equipment by means of which all persons participating in the meeting can hear each other, and such participation shall constitute presence in person at such meeting.

Section 10. <u>Action Without Meeting</u>. Unless otherwise restricted by the Certificate of Incorporation, any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting if all members of the Board of Directors or of such committee, as the case may be, consent thereto in writing and the writing or writings are filed with the minutes of proceedings of the Board of Directors or of such committee.

Section 11. Executive and Other Committees. The Board of Directors may, by resolution passed by a majority of the whole Board of Directors. designate an Executive Committee and one or more other committees, each such committee to consist of two or more directors as the Board of Directors may from time to time determine. Any such committee, to the extent provided in such resolution or resolutions, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers that may require it; but no such committee shall have the power or authority in reference to amending the Certificate of Incorporation (except that a committee may, to the extent authorized in the resolution or resolutions providing for the issuance of shares of stock adopted by the Board of Directors, fix the designations and any of the preferences or rights of such shares relating to dividends, redemption, dissolution, any distribution of assets of the Corporation or the conversion into, or the exchange of such shares for, shares of any other class or classes or any other series of the same or any other class or classes of stock of the Corporation or fix the number of shares of any series of stock or authorize the increase or decrease of the shares of any series), adopting an agreement of merger or consolidation, recommending to the stockholders the sale, lease, or exchange of all or substantially all of the Corporation's property and assets, recommending to the stockholders a dissolution of the Corporation or a revocation of a dissolution, or amending the Bylaws; and unless the resolution shall expressly so provide, no such committee shall have the power or authority to declare a dividend or to authorize the issuance of stock. In the absence or disqualification of a member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not he or they constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disgualified member. Each such committee other than the Executive Committee shall have such name as may be determined from time to time by the Board of Directors. Any committee of directors may be discharged or discontinued at any time, with or without cause, by a majority vote of the Board of Directors at any meeting at which there is a quorum present, likewise, any member of any committee of directors may be removed from committee membership, with or without cause, by a majority vote of the Board of Directors at any meeting at which there is a quorum present.

Section 12. <u>Compensation</u>. Each director shall be entitled to reimbursement of his reasonable expenses incurred in attending meetings or otherwise in connection with his attention to the affairs of the Corporation. Each director who is not a salaried officer of the Corporation or of a subsidiary of the Corporation shall, as such director and as a member of any committee, be entitled to receive such amounts as may be fixed from time to time by the Board of Directors, in the form either of fees for attendance at meetings of the Board and of committees thereof, or of payment at the rate of a fixed sum per month, or both.

Section 13. <u>Additional Powers</u>. In addition to the powers and authorities by these Bylaws expressly conferred upon it, the Board of Directors may exercise all such powers of the Corporation and do all such lawful acts and things as are not by statute or by the Certificate of Incorporation, as from time to time amended, or by these Bylaws, as from time to time amended, directed or required to be exercised or done by the stockholders.

ARTICLE V Officers

Section 1. <u>Designation</u>. The Corporation shall have such officers with such titles and duties as set forth in these Bylaws or in any one or more resolutions of the Board of Directors adopted on or after the effective date of these Bylaws which are not inconsistent with these Bylaws and as may be necessary to enable the Corporation to sign instruments and stock certificates as required by law.

Section 2. <u>Election; Qualification</u>. The officers of the Corporation shall be a Chief Executive Officer, a President, one or more Vice Presidents, a Secretary and a Treasurer, each of whom shall be elected by the Board of Directors. The Board of Directors may also elect a Chairman of the Board of Directors, a Vice Chairman of the Board of Directors, a Controller, one or more Assistant Secretaries, one or more Assistant Treasurers, one or more Assistant Treasurers, one or more Assistant Controllers, and such other officers as it may from time to time determine. The Chairman of the Board of Directors and Vice Chairman of the Board, if any, shall be elected from among the directors. Two or more offices may be held by the same person.

Section 3. <u>Term of Office</u>. Each officer shall hold office from the time of his election and qualification to the time at which his successor is elected and qualified, unless sooner he shall die or resign or shall be removed pursuant to Article V, Section 5.

Section 4. <u>Resignation</u>. Any officer of the Corporation may resign at any time by giving written notice of such resignation to the Board of Directors, the President or the Secretary of the Corporation. Any such resignation shall take effect at the time specified therein or, if no time be specified, upon receipt thereof by the Board of Directors or one of the above-named officers; and, unless specified therein, the acceptance of such resignation shall not be necessary to make it effective.

Section 5. <u>Removal</u>. Any officer may be removed at any time, with or without cause, by the vote of a majority of the whole Board of Directors.

Section 6. Vacancies. Any vacancy however caused in any office of the Corporation may be filled by the Board of Directors.

Section 7. Compensation. The compensation of each officer shall be such as the Board of Directors may from time to time determine.

Section 8. <u>Chairman of the Board of Directors and Vice Chairman of the Board of Directors</u>. The Chairman of the Board of Directors and, in his absence or inability to serve, the Vice Chairman of the Board of Directors, if such offices be occupied, shall serve as Chairman of the meetings of the Board of Directors and shall further advise and consult with the Chief Executive Officer and the President concerning the business and affairs of the Corporation and shall also have such powers and duties as the Bylaws or the Board of Directors may from time to time prescribe.

Section 9. <u>Chief Executive Officer</u>. Chief Executive Officer of the Corporation shall have general charge of the business and affairs of the Corporation and shall perform all such other duties as are incident to the chief executive officer, subject, however, to the right of the Board of Directors to confer specified powers on the Chief Executive Officer of the Corporation.

Section 10. <u>President</u>. In the absence of the Chief Executive Officer or his inability to same, the President shall serve as the Chief Executive Officer of the Corporation and shall have general charge of the business and affairs of the Corporation and shall perform all such other duties as are incident to the Chief Executive Officer, subject however to the right of the Chief Executive Officer or the Board of Directors to confer specified duties and/or powers on the President of the Corporation from time to time.

Section 11. <u>Vice President</u>. Each Vice President shall have such powers and duties as generally pertain to the office of Vice President and as the Board of Directors or the President may from time to time prescribe. During the absence of the President or his inability to act, the Vice President, or if there shall be more than one Vice President, then that one designated by the Board of Directors, shall exercise the powers and shall perform the duties of the President, subject to the direction of the Board of Directors.

Section 12. <u>Secretary</u>. The Secretary shall keep the minutes of all meetings of stockholders and of the Board of Directors. He shall be custodian of the corporate seal and shall affix it or cause it to be affixed to such instruments as he deems necessary or appropriate and attest the same and shall exercise the powers and shall perform the duties incident to the office of Secretary, and those that may otherwise from time to time be assigned to him subject to the direction of the Board of Directors.

Section 13. <u>Treasurer</u>. The Treasurer shall be the chief accounting officer of the Corporation (unless the Board of Directors appoints or has appointed another person to serve in the position of chief accounting officer) and shall have care of all funds and securities of the Corporation and shall exercise the powers and shall perform the duties incident to the office of Treasurer, subject to the direction of the Board of Directors.

Section 14. <u>Other Officers</u>. Each other officer of the Corporation shall exercise the powers and shall perform the duties incident to his office, subject to the direction of the Board of Directors.

ARTICLE VI Capital Stock

Section 1. <u>Stock Certificates</u>. The interest of each holder of stock of the Corporation shall be (a) evidenced by a certificate or certificates in such form as the Board of Directors may from time to time prescribe or (b) represented by uncertificated shares as issued by the Corporation. The issuance of shares in uncertificated form shall not affect shares already represented by a certificate until the certificate is surrendered to the Corporation. In the case of certificated shares, each certificate shall be signed by or, in the name of the Corporation by the Chairman of the Board of Directors, or the President or a Vice President, and by the Treasurer or an Assistant Treasurer, or the Secretary or an Assistant Secretary of the Corporation. If such certificate is countersigned (a) by a transfer agent other than the Corporation or its employee, or (b) by a registrar other than the Corporation or its employee, any other signature on the certificate may be facsimile. If any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if he were such officer, transfer agent or registrar at the date of issue.

Section 2. <u>Transfer of Stock</u>. Shares of stock shall be transferable on the books of the Corporation pursuant to applicable law and such rules and regulations as the Board of Directors shall from time to time prescribe on or after the effective date of these Bylaws.

Section 3. <u>Holders of Record</u>. Prior to due presentment for registration or transfer or receipt of proper transfer instructions, the Corporation may treat the holder of record of a share of its stock as the complete owner thereof exclusively entitled to vote, to receive notifications and otherwise entitled to all the rights and powers of a complete owner thereof, notwithstanding notice to the contrary.

Section 4. Lost, Stolen, Destroyed, or Mutilated Certificates. The Corporation may issue a new certificate of stock or uncertificated shares to replace a certificate alleged to have been lost, stolen, destroyed or mutilated upon terms and conditions as the Board of Directors may from time to time prescribe, and the Board of Directors may, in its discretion, require the owner of the lost or destroyed certificate or his legal representative, to give the Corporation a bond, in such sum as it may direct, not exceeding double the value of the stock, to indemnify the Corporation against any claim that may be made against it on account of the alleged loss of any such certificate.

Section 5. <u>Transfer Agent and Registrar</u>. The Board of Directors may appoint one or more Transfer Agents and Registrars for the Common Stock and Preferred Stock of the Corporation. The Transfer Agent shall be in charge of the issue, transfer, and cancellation of shares of stock and shall maintain stock transfer books, which shall include a record of the stockholders, giving the names and addresses of all stockholders, and the number and class of shares held by each; prepare voting lists for meetings of stockholders; produce and keep open these lists at the meetings; and perform such other duties as may be delegated by the Board of Directors. Stockholders may give notice of changes of their addresses to the Transfer Agent. The Registrar shall be in charge of preventing the over-issue of shares, shall register all certificated or uncertificated shares of stock, and perform such other duties as may be delegated by the Board of Directors.

ARTICLE VII Checks

Section 1. All checks or demands for money and notes of the Corporation shall be signed by such officer or officers or such other person or persons as the Board of Directors may from time to time designate.

ARTICLE VIII Fiscal Year

Section 1. The fiscal year shall begin the first day of January in each year.

ARTICLE IX Dividends

Section 1. <u>Declaration</u>. Dividends upon the capital stock of the Corporation, subject to the provisions of the Certificate of Incorporation, may be declared by the Board of Directors at any regular or special meeting, pursuant to law. Dividends may be paid in cash, in property, or in shares of the capital stock of the Corporation.

Section 2. <u>Reserve Fund</u>. The Board of Directors may set aside out of any funds of the Corporation available for dividends a reserve or reserves for any proper purposes and in such sum or sums as the directors from time to time, in their absolute discretion, believe to be proper, and the Board of Directors may abolish any such reserve.

ARTICLE X Notice

Section 1. <u>Waiver of Notice</u>. Whenever notice is required by the Certificate of Incorporation, the Bylaws, or as otherwise provided by law, a written waiver thereof, signed by the person entitled to notice, shall be deemed equivalent to notice, whether before or after the time required for such notice. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting except when the person attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the stockholders, directors, or members of a committee of directors need be specified in any written waiver of notice.

Section 2. <u>Mailing of Notice</u>. Whenever under the provisions of these Bylaws notice is required to be given to any director, officer or shareholder and such notice is not waived as provided in Section 1 of this Article X, it shall not be construed to mean personal notice, but such notice may be given in writing, by mail, by depositing the same in the post office or letter box, in post- paid sealed wrapper, addressed to such shareholder, officer or director at such address as appears on the books of the Corporation, or, in default of other address, to such director, officer or shareholder at the General Post Office in Oklahoma City, Oklahoma, and such notice shall be deemed to be given at the time when the same shall be thus mailed.

<u>ARTICLE XI</u> <u>Amendment of Bylaws</u>

Section 1. <u>Amendment</u>. These Bylaws may be made, amended, altered, added to, revised or repealed only by a vote of a majority of the directors then in office or by a vote of the holders of two-thirds of the issued and outstanding shares of stock of the Corporation entitled to vote for the election of directors; provided, however, that Article IV, Section 1 of these Bylaws and this Article XI, Section 1, may be amended, altered, added to, revised or repealed only by a vote of two-thirds of the entire Board of Directors or by a vote of two-thirds of the issued and outstanding shares of stock of the Corporation entitled to vote for the election of directors.

<u>ARTICLE XII</u> Exclusive Jurisdiction for Certain Claims

Section 1. Exclusive Forum. Unless the Board of Directors or one of its committees otherwise approves the selection of an alternate forum, the Court of Chancery of the State of Delaware (or, if, and only if, the Court of Chancery of the State of Delaware dismisses a Covered Claim (as defined below) for lack of subject matter jurisdiction, any other state or federal court in the State of Delaware that does have subject matter jurisdiction) shall, to the fullest extent permitted by applicable law, be the sole and exclusive forum for any (i) derivative claim brought in the right of the Corporation, (ii) claim asserting a breach of a fiduciary duty to the Corporation or the Corporation arising pursuant to any provision of the General Corporation Law of the State of Delaware, this Second Amended and Restated Bylaws, (iv) claim to interpret, apply, enforce or determine the validity of this Second Amended and Restated Bylaws, (iv) claim asserting one or more of the claims specified in clauses (i) through (v) herein above (each a "<u>Covered Claim</u>"); provided, however, that the provisions of this <u>Article XII(1)</u> will not apply to claims brought to enforce any liability or duty created by the Securities Exchange Act of 1934, as amended.

Section 2. <u>Personal Jurisdiction</u>. If any person or entity (a "<u>Claiming Party</u>") files an action asserting a Covered Claim in a court other than one determined in accordance with <u>Section 1</u> above (each a "<u>Foreign Action</u>") without the prior approval of the Board of Directors or one of its committees, such Claiming Party shall be deemed to have consented to (i) the personal jurisdiction of the court determined in accordance with <u>Section 1</u> in connection with any such action brought in any such court to enforce <u>Section 1</u> (an "<u>Enforcement Action</u>") and (ii) having service of process made upon such Claiming Party in any such Enforcement Action by service upon such Claiming Party's counsel in the Foreign Action as agent for such Claiming Party.

Section 3. <u>Notice and Consent</u>. Any person or entity purchasing or otherwise acquiring any interest in the shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Article XII and waived any argument relating to the inconvenience of the forums referenced above in connection with any Covered Claim.

Section 4. <u>Federal Forum</u>. Unless the Corporation consents in writing to the selection of an alternative forum, the federal district courts of the United States shall be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended. Any person or entity purchasing or otherwise acquiring any interest in any security of the Corporation shall be deemed to have notice of and consented to this provision.

These Amended and Restated Bylaws of LSB Industries, Inc. amend, restate and set forth the entire Bylaws of the Corporation and have been approved by the Board of Directors of the Corporation as of the 19th day of July, 2021.

LSB INDUSTRIES, INC.

/s/ Mark T. Behrman Mark T. Behrman Chief Executive Officer and President

/s/ Michael J. Foster Michael J. Foster, Secretary

LSB INDUSTRIES, INC. (a Delaware Corporation)

SECOND AMENDED AND RESTATED BYLAWS

ARTICLE I Offices

<u>Section 1</u>. The principal office of the Corporation shall be in Oklahoma City, County of Oklahoma, State of Oklahoma, and the Corporation may also have offices at such other places as the Board of Directors may from time to time appoint or at such other places as the business of the Corporation requires.

<u>ARTICLE II</u> <u>Seal</u>

<u>Section 1</u>. The corporate seal shall be in such form as the Board of Directors may from time to time prescribe. Said seal may be used by causing it, or a facsimile thereof, to be impressed or affixed or reproduced or otherwise.

ARTICLE III

Stockholders; Business to be Conducted at Annual or Special Meeting of Stockholders; and Stockholder Access to Corporation's Proxy Statement

Section 1. <u>Place</u>. All meetings of the stockholders shall be held in Oklahoma City, Oklahoma, or at such other place as the directors may designate.

Section 2. <u>Annual Meeting</u>. <u>Annual meetings</u> <u>An annual meeting</u> of stockholders to shall be held each calendar year on such date and at such time as shall be designated from time to time by the Board of Directors and stated in the notice of the meeting or in a duly executed waiver of notice of such meeting. At such meeting, the stockholders shall elect directors and transact such other business as may properly be presented to brought before the meeting shall be held on the last Tuesday in June of each year if not a legal holiday, and if a legal holiday, then on the next secular day following, at 10:00 a.m., or if the annual meeting is not held on the above designated date, then the directors shall cause the annual meeting to be held as soon thereafter as is convenient.

Section 3. Quorum. The holders of record of a majority of the stock issued and outstanding, and entitled to vote thereat, present in person, or represented by proxy, shall be requisite and shall constitute a quorum at all meetings of the stockholders for the transaction of business, except as otherwise provided by law, by the Certificate of Incorporation or by these Bylaws, but in the absence of a quorum the holders of record, present in person or represented by proxy at such meeting shall have power to adjourn the meeting from time to time, without notice other than announcement at the meeting, until the requisite amount of voting stock shall be present. At such adjourned meeting at which the requisite amount of voting stock shall be represented, any business may be transacted which might have been transacted at the meeting as originally notified.

Section 4. <u>Voting</u>; <u>Proxies</u>. Except as otherwise provided by the laws of the State of Delaware or the Certificate of Incorporation of the Corporation or these Bylaws:

(a) At every meeting of the stockholders every shareholder having the right to vote shall be entitled to one vote for each share of capital stock having voting rights held by him.

(b) Each stockholder entitled to vote at a meeting of stockholders or to express consent or dissent to corporate action in writing without a meeting may authorize another person or persons to act for him by proxy, but no such proxy shall be voted or acted upon after three years from its date, unless the proxy provides for a longer period.

(c) Each matter properly presented to any meeting shall be decided by a majority of the votes cast on the matter.

(d) Election of directors and the vote on any other matter presented to a meeting need not be by written ballots, but written ballots may be used if ordered by the chairman of the meeting or if so requested by any stockholder present or represented by proxy at the meeting entitled to vote in such election or on such matter, as the case may be.

Section 5. <u>Notice of Meeting</u>. For each meeting of stockholders written notice shall be given stating the place, date and hour, and, in the case of a special meeting, the purpose or purposes for which the meeting is called and, if the list of stockholders required by Section 6 is not to be at the place of said meeting at least 10 days prior to the meeting, the place where said list will be. Except as otherwise provided by Delaware law, the written notice of any meeting shall be given not less than 10 nor more than 60 days before the date of the meeting to each stockholder entitled to vote at such meeting. If mailed, notice shall be deemed to be given when deposited in the United States mail, postage prepaid, directed to the stockholder at his address as it appears on the records of the Corporation.

Section 6. List of Stockholders Entitled to Vote. At least 10 days before every meeting of stockholders a complete list of the stockholders entitled to vote at the meeting, arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder, shall be prepared and shall be open to the examination of any stockholder for any purpose germane to the meeting, during ordinary business hours, for a period of at least 10 days prior to the meeting, either at a place within the city where the meeting is to be held, which place shall be specified in the notice of the meeting, or, if not so specified, at the place where the meeting is to be held. Such list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present.

Section 7. <u>Special Meetings</u>. A special meeting of stockholders may be called at any time by the Chairman or by a majority of the directors then in office, and shall be called by the Chairman upon receipt of a written request to do so specifying the matter or matters, appropriate for action at such meeting, proposed to be presented at the meeting and signed by holders of record of two-thirds of the shares of stock that would be entitled to be voted on such matter or matters if the meeting was held on the day such request is received and the record date for such meeting was the close of business on the preceding day. Any such meeting shall be held at such time and at such place, within or without the State of Delaware, as shall be determined by the body or person calling such meeting and as shall be stated in the notice of such meeting."

Section 8. <u>Chairman and Secretary at Meeting</u>. At each meeting of stockholders, the Chairman of the Board of Directors or, in the absence or inability to serve by the Chairman of the Board of Directors, the Vice Chairman of the Board of Directors or, in the absence or inability to serve by both the Chairman of the Board of Directors and the Vice Chairman of the Board of Directors, the President or, in the absence or inability to serve by the Chairman of the Board of Directors, Vice Chairman of the Board of Directors and the President or, if the Board of Directors, Vice Chairman of the Board of Directors shall preside as Chairman of the meeting; if no person is so designated, then the Board of Directors shall choose a Chairman by plurality vote. The Secretary or in his absence a person designated by the Chairman of the meeting shall act as Secretary of the meeting.

Section 9. <u>Adjourned Meetings</u>. A meeting of stockholders may be adjourned to another time or place as provided in Sections 3 or 4(d) of this Article III. Unless the Board of Directors fixes a new record date, stockholders of record for an adjourned meeting shall be as originally determined for the meeting from which the adjournment was taken. If the adjournment is for more than 30 days, or if after the adjournment, a new record date is fixed for the adjourned meeting, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote. At the adjourned meeting any business may be transacted that might have been transacted at the meeting as originally called.

Section 10. Consent of Stockholders in Lieu of Meeting.

10.1 Action by Written Consent. Any action which is required to be or may be taken at any annual or special meeting of stockholders of the Corporation may be taken without a meeting, without prior notice and without a vote, if consents in writing, setting forth the action so taken, shall have been signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or to take such action at a meeting at which all shares entitled to vote thereon were present and voted; provided however, that prompt notice of the taking of the corporate action without a meeting and by less than unanimous written consent shall be given to those stockholders who have not consented in writing.

10.2 Determination of Record Date of Action by Written Consent. In order to inform the Corporation's stockholders and the investing public in advance that a record date for action by consent will occur and to comply with the procedures contained in the New York Stock Exchange (or such other exchange on which the Corporation's securities are listed for trading) policies and rules, the record date for determining stockholders entitled to express consent to corporate action in writing without a meeting is fixed by the Board of Directors of the Corporation pursuant to Section 213 of the Delaware General Corporation Law as follows: The Board of Directors shall set as the record date the 10th day after (i) any stockholder of record seeking to have the stockholders authorize or take corporate action by written consent without a meeting shall, by written notice to the Secretary which may be given by telex or telecopy, advise the Corporation of

the corporate action proposed for which consents will be sought and request from the Board of Directors a record date unless a later date is specified by such stockholder, or (ii) the Board of Directors determines that the Corporation should seek corporate action by written consent, unless a later record date is specified in the resolution of the Board of Directors containing such determination. In the event that the record date set as provided falls on a Saturday, Sunday or legal holiday, the record date shall be the first day next following such date that is not a Saturday, Sunday or legal holiday. Any record date determined pursuant to this Subsection 10.2 shall be announced by a press release prior to the opening of trading on the New York Stock Exchange (or such other exchange on which the Corporation's securities are listed for trading) on the next trading day after a request for a record date pursuant to clause (i) above is received by the Secretary or a Board of Directors' determination pursuant to clause (ii) above.

10.3 Duration and Revocation of Consents. In order that the Corporation's stockholders shall have an opportunity to receive and consider the information germane to an informed judgment as to whether to give a written consent and in accordance with the procedures contained in the New York Stock Exchange (or such other exchange on which the Corporation's securities are listed for trading) policies and rules, the stockholders of the Corporation shall be given at least 20 days from the record date to give or revoke written consents. Consents to corporate action shall be valid for a maximum of 60 days after the record date. Consents may be revoked by written notice (i) to the Corporation, (ii) to the stockholder or stockholders soliciting consents or soliciting revocations in opposition to action by consent proposed by the Corporation (the "Soliciting Stockholders"), or (iii) to a proxy solicitor or other agent designated by the Corporation of the Soliciting Stockholder(s).

10.4 <u>Retention and Duties of Inspectors of Election</u>. Within two business days after receipt of a request by a stockholder for the setting of a record date or a determination by the Board of Directors that the Corporation should seek corporate action by written consent, as the case may be, the Secretary of the Corporation shall engage nationally recognized independent inspectors of elections for the purpose of performing a ministerial review of the validity of the consents and revocations. The inspectors shall review all consents and revocations, determine whether the requisite number of valid and unrevoked consents has been obtained to authorize or take the action specified in the consents, and forthwith certify such determination for entry in the records of the Corporation kept for the purpose of recording the proceedings of meetings of stockholders. The cost of retaining inspectors of elections shall be borne by the party proposing the action by consent.

10.5 Procedures for Counting and Challenging Consents. Consents and revocations shall be delivered to the inspectors upon receipt by the Corporation, the Soliciting Stockholders or their proxy solicitors or other designated agents. As soon as consents and revocations are received, the inspectors shall review the consents and revocations and shall maintain a count of the number of valid and unrevoked consents. The inspectors shall keep such count confidential and shall not reveal the count to the Corporation, the Soliciting Stockholders or their representatives. As soon as practicable after the earlier of (i) 60 days after the record date for the consents or (ii) a request therefore by the Corporation or the Soliciting Stockholders (whichever is soliciting consents) made after expiration of the period for giving or revoking consents under Subsection 10.3 above, notice of which request shall be given to the party opposing the solicitation of consents, which request shall state that the Corporation or Soliciting Stockholder(s) (as the case may be) in good faith believe that it or they have received the requisite number of valid and unrevoked consents to authorize or take the action specified in the consents, the inspectors shall issue a preliminary report to the Corporation and the Soliciting Stockholders stating:

- (i) The number of valid consents;
- (ii) The number of valid revocations;
- (iii) The number of valid and unrevoked consents;
- (iv) The number of invalid consents;
- (v) The number of invalid revocations;
- (vi) Whether, based on their preliminary count, the requisite number of valid and unrevoked consents has been obtained to authorize or take the action specified in the consents.

Unless the Corporation and the Soliciting Stockholder(s) shall agree to a shorter or longer period, the Corporation and the Soliciting Stockholder(s) shall have 48 hours to review the consents and revocations and to advise the inspectors and the opposing party in writing as to whether they intend to challenge the preliminary report of the inspectors. If no written notice of an intention to challenge the preliminary report is received within 48 hours after the inspector's issuance of the preliminary report, the inspectors shall issue to the Corporation and the Soliciting Stockholder(s) their final report containing the information from the inspectors' determination with respect to whether the requisite number of valid and unrevoked consents was obtained to authorize and take the action specified in the consents. If the Corporation or the Soliciting Stockholder(s) issue written notice of an intention to challenge the inspectors' preliminary report within 48 hours after the issuance of that report, a challenge session shall be scheduled by the inspectors as promptly as practicable. A transcript of the challenge session shall be recorded by a certified court reporter. Following completion of the challenge session, the inspectors shall as promptly as practicable issue their final report to the Corporation and the Soliciting Stockholder(s) containing the information included in the preliminary report, plus all changes in the vote totals as a result of the challenges and a certification of whether the requisite number of valid and unrevoked consents was obtained to authorize or take the action specified in the consents after the issue to the Corporation and the Soliciting Stockholder(s) containing the information included in the preliminary report, plus all changes in the vote totals as a result of the challenges and a certification of whether the requisite number of valid and unrevoked consents was obtained to authorize or take the action specified in the consents.

10.6 <u>Notice of Results</u>. The Corporation shall give prompt notice to the stockholders of the results of any consent solicitation or the taking of the corporate action without a meeting and by less than unanimous written consent.

Section 11. <u>Fixing of Record Date</u>. In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, or to express consent to corporate action in writing without a meeting, or entitled receive payment of any dividend or other distribution or allotment of any rights, or entitled to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix, in advance, a record date, which shall not be more

than 60 nor less than 10 days before the date of such meeting, nor more than 60 days prior to any other action. If no record date is fixed, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held; the record date for determining stockholders entitled to express consent to corporate action in writing without a meeting, when no prior action by the Board of Directors is necessary, shall be the day on which the first written consent is expressed; and the record date for determining stockholders for any other purpose shall be at the close of business on the day on which the Board of Directors adopts the resolution relating thereto.

Section 12. <u>Business to be Conducted at the Annual or Special Meeting of the Stockholders; Notice of Proposals</u>. At any annual meeting of the stockholders, only such business shall be conducted as shall have been brought before the meeting (i) by or at the direction of the Board of Directors, or (ii) by any stockholder of the Corporation who is entitled to vote with respect thereto and who: (a) was a stockholder of record (and, with respect to any beneficial owner, if different, on whose behalf such business as proposed, only if such beneficial owner was the beneficial owner of shares of the Corporation) both at the time of giving notice provided for in this Section 12 and at the time of the meeting; and (b) complies with the notice procedures set forth in this Section 12.

Except for proposals properly made in accordance with Rule 14a-8 under the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder (as so amended and inclusive of such rules as regulations, the "Exchange Act") and included in the notice of the meeting given by or at the direction of the Board of Directors, the foregoing clause (ii) will be the exclusive means for a stockholder to propose business to be brought before the annual meeting of stockholders.

For business to be properly brought before the annual meeting by a stockholder, the Proposing Person (as defined below) must have given timely notice thereof in writing to the Secretary of the Corporation. The Proposing Person's notice will be timely if delivered or mailed to and received at the principal executive offices at the Corporation not less than 120 nor more than 150 days before the date on which the Corporation first mailed its proxy materials for the prior year's annual meeting of stockholders; provided however, that if the date of the annual meeting is more than 30 days before or more than 60 days after such date, notice by the stockholder timely must be so delivered, or mailed and received not later than the 90th day prior to such annual meeting, or if later, the 10th day following the date on which the public disclosure of the date of such annual meeting was such made. Any adjournment of an annual meeting or the announcement hereof will not commence a new time period for giving the notice described above.

The Proposing Person's notice to the Secretary shall set forth as to each matter such stockholder proposes to bring before the annual meeting, the following:

(i) the name and address, as they appear on the Corporation books, of the stockholder proposing such business;

- (ii) the class or series and number of shares of the Corporation's securities that are, directly or indirectly, owned of record or beneficially owned (within the meaning of Rule 13d-3 under the Exchange Act) by the Proposing Person, except the Proposing Person will be deemed to beneficially own any shares or class or series of the Corporation's securities which the Proposing Person has a right to acquire beneficially ownership at any time in the future (collectively, the "Stockholder Information");
- (iii) as to each item of business that the stockholder proposes to bring before the annual meeting, (A) a reasonably brief description of the business desired to be brought before the annual meeting, the reasons for conducting such business at the annual meeting and any material interest in such business of each Proposing Person, (B) the text of the proposal or business (including the text of any resolutions proposed for consideration), and
- (iv) a reasonably detailed description of all agreements, arrangements and understandings, oral or in writing (x) between or among any of the Proposing Persons or (y) between or among any Proposing Person and any other record or beneficial holder of the shares of any class or series of the Corporation (including their names) in connection with the proposal of such business by such stockholder or (z) between or among any Proposing Person and any other persons or entities (including their names) acting in concert with the Proposing Person.

For purposes of this Section 12, the term "Proposing Person" shall mean (i) the stockholder providing the notice of business proposed to be brought before an annual meeting, (ii) the beneficial owner or beneficial owners, if different, on whose behalf the notice of the business proposed to be brought before the annual meeting is made, and (iii) any affiliate or associate (each within the meaning of Rule 12b-2 under the Exchange Act for purposes of these Bylaws) of such stockholder or beneficial owner; and (iv) any material interest of such stockholder with respect to such business.

Notwithstanding anything in these Bylaws to the contrary, no business (other than nominations of directors, which must be made in compliance with, and shall be exclusively governed by, Article III, Section 13 of these Bylaws) shall be brought before or conducted at the annual meeting except in accordance with the provisions of this Section 12. The officer of the Corporation or other person presiding over the annual meeting shall, if the facts so warrant, determine and declare to the meeting that business was not properly brought before the meeting in accordance with the provisions of this Section 12 and, if he should so determine, he shall so declare to the meeting and any such business so determined to be not properly brought before the meeting shall not be transacted.

A stockholder providing notice of business proposed to be brought before an annual meeting shall further update and supplement such notice, as needed, so that the information provided or required to be provided in such notice pursuant to this Section 12 shall be true and correct as of the record date for the meeting and as of the date that is 10 business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be

delivered to, or mailed and received by, the Secretary of the Corporation at the principal executive offices of the Corporation not later than five business days after the record date for the meeting (in the case of the update and supplement required to be made as of the record date), and not later than eight business days prior to the date for the meeting (in the case of the update and supplement required to be made as of 10 business days prior to the meeting or any adjournment or postponement thereof).

This Section 12 is expressly intended to apply to any business proposed to be brought before an annual meeting of stockholders; except this Section 12 shall not apply to any proposal made pursuant to Rule 14a-8 of the Exchange Act, or to the nomination of persons for election to the Corporation's Board of Directors at a meeting of stockholders at which directors are to be elected which shall be governed by Article III, Section 13 of these Bylaws. In addition to the requirements of this Section 12 with respect to any business proposed to be brought before an annual meeting, each Proposing Person shall comply with all applicable requirements of the Exchange Act with respect to any such business. Nothing in this Section 12 shall be deemed to affect the rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act.

For purposes of these Bylaws, "public disclosure" shall mean disclosure in a press release reported by a national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Sections 13, 14 or 15(d) of the Exchange Act.

The officer of the Corporation or other person presiding at the meeting shall, if the facts so warrant, determine that business was not properly brought before the meeting in accordance with the procedures set forth in this Section 12, and, if he or she should so determine, he or she shall so declare to the meeting and any such business not properly brought before the meeting shall not be transacted.

At any special meeting of the stockholders, only such business shall be conducted as shall have been brought before the meeting by or at the direction of the Board of Directors.

Section 13. Election to the Board of Directors.

13.1 Only persons who are nominated in accordance with the procedures set forth in these Bylaws shall be eligible for election as directors of the Corporation. Nominations of persons for election to the Board of Directors of the Corporation may be made at a meeting of stockholders at which directors are to be elected only:

- (i) By or at the direction of the Board of Directors; or
- (ii) By any stockholder of the Corporation entitled to vote for the election of directors at the meeting who complies with the notice procedures set forth in Subsection 13.2 below.

The foregoing clause (ii) will be the exclusive means by which a stockholder may nominate a person for election to the Board of Directors.

13.2 Nominations of election as a director of the Corporation, other than those made by or at the direction of the Board of Directors, shall be made by timely notice in writing to the Secretary of the Corporation. To be timely, a stockholder's notice shall be delivered or mailed to and received at the principal executive offices of the Corporation not less than 120 nor more than 150 days prior to the <u>anniversary</u> date of the <u>Corporation's immediately</u> <u>preceding annual</u> meeting <u>of stockholders</u>; provided, however, that in the event the date of the annual meeting is more than 30 days before or more than 60 days after such date, notice by the stockholder to be timely must be so delivered, or mailed and received not later than the 90th day prior to such annual meeting, or if later, the 10th day following the date on which the public disclosure of the date of such annual meeting was so made. Any adjournment of an annual meeting or the announcement hereof will not commence a new time period for giving the timely notice described above. Such stockholder's notice shall set forth:

- (i) As to each person whom such stockholder proposes to nominate for election or reelection as a director, (x) all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934, as amended (including such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected, and (y) a representation that such a person shall also provide any other information reasonably requested by the Corporation within 10 business days after such request); and
- (ii) As to the stockholder giving the notice (x) the name and address, as they appear on the Corporation's books, of such stockholder, and (y) the class and number of shares of the Corporation's voting capital stock that are beneficially owned by such stockholder.

At the request of the Board of Directors, any person nominated by the Board of Directors for election as a director shall furnish to the Secretary of the Corporation that information required to be set forth in a stockholder's notice of nomination which pertains to the nominee.

No person shall be eligible for election as a director of the Corporation unless nominated in accordance with the provisions of this Section 13, and, if the stockholder desires or requests access to the Corporation's Proxy Statement with respect to the election of a director, Article III, Section 14 of these Bylaws.

The officer of the Corporation or other person presiding at the meeting shall, if the facts so warrant, determine that a nomination was not made in accordance with such provisions and, if he or she should so determine, he or she shall so declare to the meeting and the defective nomination shall be disregarded.

Section 14. Stockholder Access to Corporation's Proxy Statement.

(a) Whenever the Corporation solicits proxies with respect to an election of directors at an annual meeting (an "Election"), it shall include in its proxy statement and on its proxy card, in addition to individuals nominated by the Board of Directors, up to the Permitted Number of individuals nominated in compliance with these Bylaws by one or more Eligible Stockholders. Any Eligible Stockholder seeking to have its nominee included in the Corporation's proxy statement and on the Corporation's proxy card shall comply with all provisions of these Bylaws otherwise applicable to shareholder nominations and furnish to the Secretary of the Corporation, no later than the last day on which stockholder nominations for consideration in the Election may be made under Article III, Section 13.2 \Box of these Bylaws (the "Advance Notice Date"),

- (i) the information set forth in Sections 3.2 of these Bylaws,
- (ii) the written undertakings described in subsections (d) and (e) below, and
- (iii) any accompanying statement from the Eligible Stockholder to be included in the Corporation's proxy statement, which statement in order to be so included shall not exceed 500 words and must fully comply with Section 14 of the Exchange Act and the rules and regulations promulgated thereunder, including without limitation Rule 14a-9.
- (b) For purposes of this Section:
 - (i) The "Permitted Number" means one seat on the Corporation's Board of Directors to be filled in the Election; *provided*, *however*, that the Permitted Number shall be reduced, but not below zero, by the sum of (i) number of such director candidates for which the Corporation shall have received from Eligible Stockholders by the Advance Notice Date one or more valid stockholder notices nominating director candidates, and (ii) the number of directors in office and serving in the class of directors to be considered at the Election for whom access to the Corporation's proxy materials was provided pursuant to this Section ("Access Director"), other than any who has served as a director continuously for at least six years. In no event will the number of nominees nominated by an Eligible Stockholder for which access to the Corporation's proxy materials may be provided in the Election plus the number of Access Directors serving on the Board at the time of the Election exceed 25% of the total number of directors that shall constitute the whole board.
 - (ii) An "Eligible Stockholder" means a stockholder of the Corporation that, together with its Affiliates, has continuously held Beneficial Ownership and Economic Interest of not less than the Required Interest for at least one year preceding the Advance Notice Date, and that complies with all applicable provisions of these Bylaws.
 - (iii) "Beneficial Ownership" means the power to vote or direct the voting of, or to dispose or direct the disposition of, the securities in question.
 - (iv) An "Economic Interest" in a security means having or sharing the opportunity, directly or indirectly, to profit or share in any profit (or loss) derived from a transaction in the security, including through options, swaps or other derivative securities or synthetic arrangements.

- (v) "Independent" with respect to a nominee for director pursuant to this Section 14 shall mean (a) that the nominee would be considered an independent director in accordance with the listing standards of the principal U.S. securities market in which the common stock of the Corporation trades or, if no such listing standards are applicable at the time, in accordance with the standards used by the Board of Directors or a duly authorized committee thereof in determining and disclosing the independence of the Corporation's directors in accordance with the rules of the SEC and (b) the nominee is not an employee or officer of, or consultant to, the Eligible Stockholder or any of its Affiliates and has no other material association, by agreement, understanding or familial or other relationship, with the Eligible Stockholder or any of its Affiliates.
- (vi) A "Disqualified Repeat Nominee" in respect of an election shall mean an individual as to whom access to the Corporation's proxy materials for the immediately preceding election was provided and who (i) withdrew from or became ineligible or unavailable for election at the meeting, or (ii) received at such meeting votes in favor of his or her election representing less than 50% of the total votes cast for or withheld from his or her election.
- (vii) The "Required Interest" means 5% of the voting power of the outstanding voting securities of the Corporation entitled to vote in the Election, based upon the number of outstanding voting securities of the Corporation most recently disclosed prior to the Advance Notice Date by the Corporation in a filing with the Securities and Exchange Commission.
- (viii) "Affiliate" of a specified person means a person that, directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with, the specified person, and, with respect to any investment company (as defined in the Investment Company Act of 1940, whether or not exempt from registration thereunder), shall also include all other investment companies managed by the same investment adviser or any of its Affiliates.

(c) Subject to the following sentence and any undertaking previously provided by an Eligible Stockholder pursuant to Section 14(d) below, each Eligible Stockholder, together with its Affiliates, may nominate one, and not more than one, individual under this Section for inclusion in the Corporation's proxy statement and on its proxy card. If the Corporation shall receive more than the Permitted Number of proposed nominations from Eligible Stockholders in compliance with these Bylaws, then the nominee nominated by the Eligible Stockholder possessing the largest Beneficial Ownership of voting securities of the Corporation as of the Advance Notice Date shall be the only nominee for purposes of this Section 14.

(d) Any Eligible Stockholder nominating an individual for director in accordance with this Section shall execute and deliver to the Corporation no later than the Advance Notice Date an undertaking, in a form to be provided by the Secretary of the Corporation, that it will, and will cause its Affiliates to,

- (i) not sell or otherwise dispose of its Beneficial Ownership and Economic Interest of voting securities of the Corporation so as to reduce the Beneficial Ownership and Economic Interest held by such Eligible Stockholder, together with its Affiliates, below the Required Interest on or prior to the date of the Election (and representing that they have no present intention of reducing, within one year following the Election, their aggregate Beneficial Ownership and Economic Interest below the greater of (x) the Required Interest and (y) 75% of their aggregate Beneficial and Economic Interest as of the Advance Notice Date),
- (ii) comply with the provisions of the Corporation's Certificate of Incorporation and Bylaws and all laws and regulations relating to the accompanying statement submitted by the Eligible Stockholder and any solicitation or communications with stockholders of the Corporation in connection with such nomination,
- (iii) indemnify the Corporation and its agents and representatives in respect of any and all liabilities that may arise out of the accompanying statement submitted by the Eligible Stockholder or any solicitation or communications with stockholders of the Corporation by such Eligible Stockholder, its Affiliates or their respective agents or representatives in connection with such nomination, including as a result of any violation of law or regulation by such Eligible Stockholder, its Affiliates or their respective agents or representatives in connection therewith,
- (iv) not use any proxy card other than the Corporation's proxy card in soliciting stockholders in connection with the matters to be voted on at the meeting at which the Election is held,
- (v) file all solicitation materials used by it or on its behalf with the Securities and Exchange Commission under cover of Schedule 14A promulgated under the Exchange Act, and
- (vi) for a period of one year from the date of the Election, not (x) nominate any individual to be a director of the Corporation or conduct any solicitation with respect to an election for directors of the Corporation other than with respect to the Election and in accordance with this Section, or (y) acquire or propose to acquire Beneficial Ownership of or an Economic Interest in any voting securities of the Corporation such that such Eligible Stockholder, together with its Affiliates, would have aggregate Beneficial Ownership of, and/or an Economic Interest in, more than the greater of (I) 10% of the voting power of the outstanding voting securities of the Corporation or (II)

an additional 5% of the voting power of the Corporation's outstanding voting securities in excess of the aggregate Beneficial Ownership and Economic Interest held by such Eligible Stockholder, together with its Affiliates, as of the Advance Notice Date (the "Aggregate Beneficial Ownership"); provided that the Aggregate Beneficial Ownership shall not equal or exceed the amount that would equal or exceed the beneficial ownership threshold necessary to trigger the Corporation preferred share rights plan as may be in effect during such time.

(e) Any Eligible Stockholder nominating an individual for director in accordance with this Section shall also deliver to the Corporation no later than the Advance Notice Date a signed undertaking of its nominee agreeing that he or she will tender his or her resignation from the Board of Directors if

- (i) any of the information provided to the Corporation by the Eligible Stockholder or the nominee pursuant to this Bylaw is determined to be inaccurate in any material respect, or
- (ii) the Eligible Stockholder or any of its Affiliates shall breach their obligations under the undertakings described in subsection(d) above in any material respect.

(f) The Nominating and Governance Committee shall consider a nomination pursuant to this Section 14, and shall determine if the Access Nominee is Independent and may, in its discretion, make a recommendation to the Board of Directors as to whether the Access Nominee should be nominated by the Board of Directors for election at the Annual Meeting of Stockholders.

If the Board of Directors nominates an Access Nominee as part of the Board's slate of nominees, the Notice of Access will be deemed withdrawn and the former Access Nominee shall be presented to the stockholders in the same manner as any other nominee of the Board of Directors. If the Board of Directors does not so nominate the Access Nominee, access to the Corporation's proxy materials shall be provided in accordance with the terms and subject to the conditions of this Section.

The Board of Directors or a committee thereof may adopt such rules or guidelines for applying the provisions of this Section as it determines are appropriate. These may include timing and other such adjustments as may be appropriate in the event an Access Nominee for whom Notice of Access has been provided becomes unavailable or unwilling to serve or becomes ineligible.

(g) This Section shall provide the exclusive method for stockholders to include nominees for director in the Corporation's proxy statement and on the Corporation's proxy card.

ARTICLE IV Directors

Section 1. <u>Number, Term, Qualifications and Vacancies</u>. The property, business and affairs of the Corporation shall be managed by or under the direction of its Board of Directors.

The number of directors that shall constitute the whole Board of Directors may be fixed from time to time pursuant to a resolution adopted by a vote of two-thirds of the entire Board of Directors and may consist of no fewer than three nor more than thirteenfourteen members. The directors shall be divided into three classes. Each class shall consist, as nearly as possible, of one-third of the whole number of the Board of Directors. At each annual election of the successors to the class of directors whose terms have expired in that year shall be elected to hold office for a term of three years. Each director elected shall hold office until his successor is elected and qualified or until his earlier resignation or removal. Directors and officers need not be stockholders.

Vacancies and newly created directorships resulting from any increase in the authorized number of directors shall be filled only by a majority of the directors then in office, although less than a quorum, or by the sole remaining director. Each director chosen to fill a vacancy or newly created directorship shall hold office until the next election of the class for which such directors shall have been chosen and until his successor is duly elected and qualified or until his earlier resignation or removal.

Section 2. <u>Offices and Books</u>. The directors may have one or more offices, and keep the books of the Corporation at the offices of the Corporation in Oklahoma City, Oklahoma, or at such other places as they may from time to time determine.

Section 3. <u>Resignation</u>. Any director of the Corporation may resign at any time by giving written notice of such resignation to the Board of Directors, the Chairman of the Board of Directors, the President or the Secretary of the Corporation. Any such resignation shall take effect at the time specified therein or, if no time be specified, upon receipt thereof by the Board of Directors, or one of the above-named officers; and, unless specified therein, the acceptance of such resignation shall not be necessary to make it effective. When one or more directors shall resign from the Board of Directors, such vacancy shall be filled only by a majority of the directors then in office, although less than a quorum, or by the sole remaining director. Each director so chosen shall hold office until the next election of the class for which such director shall have been chosen and until his successor is duly elected and qualified or until his earlier resignation or removal.

Section 4. <u>Removal</u>. Any one or more directors may be removed only for cause by the vote or written consent of the holders of a majority of the issued and outstanding shares of stock of the Corporation entitled to vote for the election of all directors. For purposes of this Article IV, Section 4, cause for removal shall be deemed to exist only if the director whose removal is proposed has been convicted of a felony by a court of competent jurisdiction or has been adjudged by a court of competent jurisdiction to be liable for intentional misconduct or knowing violation of law in the performance of such director's duty to the Corporation and, in each case, such adjudication is no longer subject to direct appeal.

Section 5. <u>Regular and Annual Meetings; Notice</u>. Regular meetings of the Board of Directors shall be held at such time and at such place, within or without the State of Delaware, as the Board of Directors may from time to time prescribe. No notice need be given of any regular meeting and a notice, if given, need not specify the purposes thereof. A meeting of the Board of Directors may be held without notice immediately after an annual meeting of stockholders at the same place as that at which such annual meeting of stockholders was held.

Section 6. <u>Special Meetings; Notice</u>. A special meeting of the Board of Directors may be called at any time by the Chairman or a majority of the directors then in office. Any such meeting shall be held at such time and at such place, within or without the State of Delaware, as shall be determined by the body or person calling such meeting. Notice of such meeting stating the time and place thereof shall be given (a) by deposit of the notice in the United States mail, first class, postage prepaid, at least three days before the day fixed for the meeting addressed to each director at his address as it appears on the Corporation's records or at such other address as the director may have furnished the Corporation for that purpose, or (b) by delivery of the notice similarly addressed for dispatch by telegraph, cable or radio or by delivery of the notice by telephone or in person, in each case at least two days before the time fixed for the meeting.

Section 7. <u>Presiding Officer and Secretary at Meetings</u>. Each meeting of the Board of Directors shall be presided over by the Chairman of the Board of Directors or in his absence by the President or if neither is present by such member of the Board of Directors as shall be chosen by the meeting. The Secretary, or in his absence an Assistant Secretary, shall act as secretary of the meeting, or if no such officer is present, a secretary of the meeting shall be designated by the person presiding over the meeting.

Section 8. <u>Quorum</u>. A majority of the whole Board of Directors shall constitute a quorum for the transaction of business, but in the absence of a quorum, a majority of those present (or if only one be present, then that one) may adjourn the meeting, without notice other than announcement at the meeting, until such time as a quorum is present. Except as otherwise required by the Certificate of Incorporation or these Bylaws, the vote of the majority of the directors present at a meeting at which a quorum is present shall be the act of the Board of Directors.

Section 9. <u>Meeting by Telephone</u>. Members of the Board of Directors or of any committee thereof may participate in meetings of the Board of Directors or of such committee by means of conference telephone or similar communications equipment by means of which all persons participating in the meeting can hear each other, and such participation shall constitute presence in person at such meeting.

Section 10. <u>Action Without Meeting</u>. Unless otherwise restricted by the Certificate of Incorporation, any action required or permitted to be taken at any meeting of the Board of Directors or of any committee thereof may be taken without a meeting if all members of the Board of Directors or of such committee, as the case may be, consent thereto in writing and the writing or writings are filed with the minutes of proceedings of the Board of Directors or of such committee.

Section 11. <u>Executive and Other Committees</u>. The Board of Directors may, by resolution passed by a majority of the whole Board of Directors, designate an Executive Committee and one or more other committees, each such committee to consist of two or more directors as the Board of Directors may from time to time determine. Any such committee, to the extent provided in such resolution or resolutions, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the

seal of the Corporation to be affixed to all papers that may require it; but no such committee shall have the power or authority in reference to amending the Certificate of Incorporation (except that a committee may, to the extent authorized in the resolution or resolutions providing for the issuance of shares of stock adopted by the Board of Directors, fix the designations and any of the preferences or rights of such shares relating to dividends, redemption, dissolution, any distribution of assets of the Corporation or the conversion into, or the exchange of such shares for, shares of any other class or classes or any other series of the same or any other class or classes of stock of the Corporation or fix the number of shares of any series of stock or authorize the increase or decrease of the shares of any series), adopting an agreement of merger or consolidation, recommending to the stockholders the sale, lease, or exchange of all or substantially all of the Corporation's property and assets, recommending to the stockholders a dissolution of the Corporation or a revocation of a dissolution, or amending the Bylaws; and unless the resolution shall expressly so provide, no such committee shall have the power or authority to declare a dividend or to authorize the issuance of stock. In the absence or disqualified of a member of a committee, the member of member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member. Each such committee of directors may be determined from time to time by the Board of Directors at any meeting at which there is a quorum present, likewise, any meeting at which there is a quorum present.

Section 12. <u>Compensation</u>. Each director shall be entitled to reimbursement of his reasonable expenses incurred in attending meetings or otherwise in connection with his attention to the affairs of the Corporation. Each director who is not a salaried officer of the Corporation or of a subsidiary of the Corporation shall, as such director and as a member of any committee, be entitled to receive such amounts as may be fixed from time to time by the Board of Directors, in the form either of fees for attendance at meetings of the Board and of committees thereof, or of payment at the rate of a fixed sum per month, or both.

Section 13. <u>Additional Powers</u>. In addition to the powers and authorities by these Bylaws expressly conferred upon it, the Board of Directors may exercise all such powers of the Corporation and do all such lawful acts and things as are not by statute or by the Certificate of Incorporation, as from time to time amended, or by these Bylaws, as from time to time amended, directed or required to be exercised or done by the stockholders.

ARTICLE V Officers

Section 1. <u>Designation</u>. The Corporation shall have such officers with such titles and duties as set forth in these Bylaws or in any one or more resolutions of the Board of Directors adopted on or after the effective date of these Bylaws which are not inconsistent with these Bylaws and as may be necessary to enable the Corporation to sign instruments and stock certificates as required by law.

Section 2. <u>Election; Qualification</u>. The officers of the Corporation shall be a Chief Executive Officer, a President, one or more Vice Presidents, a Secretary and a Treasurer, each of whom shall be elected by the Board of Directors. The Board of Directors may also elect a Chairman of the Board of Directors, a Vice Chairman of the Board of Directors, a Controller, one or more Assistant Secretaries, one or more Assistant Treasurers, one or more Assistant Treasurers, one or more Assistant Controllers, and such other officers as it may from time to time determine. The Chairman of the Board of Directors and Vice Chairman of the Board, if any, shall be elected from among the directors. Two or more offices may be held by the same person.

Section 3. <u>Term of Office</u>. Each officer shall hold office from the time of his election and qualification to the time at which his successor is elected and qualified, unless sooner he shall die or resign or shall be removed pursuant to Article V, Section 5.

Section 4. <u>Resignation</u>. Any officer of the Corporation may resign at any time by giving written notice of such resignation to the Board of Directors, the President or the Secretary of the Corporation. Any such resignation shall take effect at the time specified therein or, if no time be specified, upon receipt thereof by the Board of Directors or one of the above-named officers; and, unless specified therein, the acceptance of such resignation shall not be necessary to make it effective.

Section 5. <u>Removal</u>. Any officer may be removed at any time, with or without cause, by the vote of a majority of the whole Board of Directors.

Section 6. Vacancies. Any vacancy however caused in any office of the Corporation may be filled by the Board of Directors.

Section 7. Compensation. The compensation of each officer shall be such as the Board of Directors may from time to time determine.

Section 8. <u>Chairman of the Board of Directors and Vice Chairman of the Board of Directors</u>. The Chairman of the Board of Directors and, in his absence or inability to serve, the Vice Chairman of the Board of Directors, if such offices be occupied, shall serve as Chairman of the meetings of the Board of Directors and shall further advise and consult with the Chief Executive Officer and the President concerning the business and affairs of the Corporation and shall also have such powers and duties as the Bylaws or the Board of Directors may from time to time prescribe.

Section 9. <u>Chief Executive Officer</u>. Chief Executive Officer of the Corporation shall have general charge of the business and affairs of the Corporation and shall perform all such other duties as are incident to the chief executive officer, subject, however, to the right of the Board of Directors to confer specified powers on the Chief Executive Officer of the Corporation.

Section 10. <u>President</u>. In the absence of the Chief Executive Officer or his inability to same, the President shall serve as the Chief Executive Officer of the Corporation and shall have general charge of the business and affairs of the Corporation and shall perform all such other duties as are incident to the Chief Executive Officer, subject however to the right of the Chief Executive Officer or the Board of Directors to confer specified duties and/or powers on the President of the Corporation from time to time.

Section 11. <u>Vice President</u>. Each Vice President shall have such powers and duties as generally pertain to the office of Vice President and as the Board of Directors or the President may from time to time prescribe. During the absence of the President or his inability to act, the Vice President, or if there shall be more than one Vice President, then that one designated by the Board of Directors, shall exercise the powers and shall perform the duties of the President, subject to the direction of the Board of Directors.

Section 12. <u>Secretary</u>. The Secretary shall keep the minutes of all meetings of stockholders and of the Board of Directors. He shall be custodian of the corporate seal and shall affix it or cause it to be affixed to such instruments as he deems necessary or appropriate and attest the same and shall exercise the powers and shall perform the duties incident to the office of Secretary, and those that may otherwise from time to time be assigned to him subject to the direction of the Board of Directors.

Section 13. <u>Treasurer</u>. The Treasurer shall be the chief accounting officer of the Corporation <u>(unless the Board of Directors appoints or has appointed another person to serve in the position of chief accounting officer</u>) and shall have care of all funds and securities of the Corporation and shall exercise the powers and shall perform the duties incident to the office of Treasurer, subject to the direction of the Board of Directors.

Section 14. <u>Other Officers</u>. Each other officer of the Corporation shall exercise the powers and shall perform the duties incident to his office, subject to the direction of the Board of Directors.

ARTICLE VI Capital Stock

Section 1. <u>Stock Certificates</u>. The interest of each holder of stock of the Corporation shall be (a) evidenced by a certificate or certificates in such form as the Board of Directors may from time to time prescribe or (b) represented by uncertificated shares as issued by the Corporation. The issuance of shares in uncertificated form shall not affect shares already represented by a certificate until the certificate is surrendered to the Corporation. In the case of certificated shares, each certificate shall be signed by or, in the name of the Corporation by the Chairman of the Board of Directors, or the President or a Vice President, and by the Treasurer or an Assistant Treasurer, or the Secretary or an Assistant Secretary of the Corporation. If such certificate is countersigned (a) by a transfer agent other than the Corporation or its employee, or (b) by a registrar other than the Corporation or its employee, any other signature on the certificate may be facsimile. If any officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if he were such officer, transfer agent or registrar at the date of issue.

Section 2. <u>Transfer of Stock</u>. Shares of stock shall be transferable on the books of the Corporation pursuant to applicable law and such rules and regulations as the Board of Directors shall from time to time prescribe on or after the effective date of these Bylaws.

Section 3. <u>Holders of Record</u>. Prior to due presentment for registration or transfer or receipt of proper transfer instructions, the Corporation may treat the holder of record of a share of its stock as the complete owner thereof exclusively entitled to vote, to receive notifications and otherwise entitled to all the rights and powers of a complete owner thereof, notwithstanding notice to the contrary.

Section 4. Lost, Stolen, Destroyed, or Mutilated Certificates. The Corporation may issue a new certificate of stock or uncertificated shares to replace a certificate alleged to have been lost, stolen, destroyed or mutilated upon terms and conditions as the Board of Directors may from time to time prescribe, and the Board of Directors may, in its discretion, require the owner of the lost or destroyed certificate or his legal representative, to give the Corporation a bond, in such sum as it may direct, not exceeding double the value of the stock, to indemnify the Corporation against any claim that may be made against it on account of the alleged loss of any such certificate.

Section 5. <u>Transfer Agent and Registrar</u>. The Board of Directors may appoint one or more Transfer Agents and Registrars for the Common Stock and Preferred Stock of the Corporation. The Transfer Agent shall be in charge of the issue, transfer, and cancellation of shares of stock and shall maintain stock transfer books, which shall include a record of the stockholders, giving the names and addresses of all stockholders, and the number and class of shares held by each; prepare voting lists for meetings of stockholders; produce and keep open these lists at the meetings; and perform such other duties as may be delegated by the Board of Directors. Stockholders may give notice of changes of their addresses to the Transfer Agent. The Registrar shall be in charge of preventing the over-issue of shares, shall register all certificated or uncertificated shares of stock, and perform such other duties as may be delegated by the Board of Directors.

ARTICLE VII Checks

Section 1. All checks or demands for money and notes of the Corporation shall be signed by such officer or officers or such other person or persons as the Board of Directors may from time to time designate.

ARTICLE VIII Fiscal Year

Section 1. The fiscal year shall begin the first day of January in each year.

ARTICLE IX Dividends

Section 1. <u>Declaration</u>. Dividends upon the capital stock of the Corporation, subject to the provisions of the Certificate of Incorporation, may be declared by the Board of Directors at any regular or special meeting, pursuant to law. Dividends may be paid in cash, in property, or in shares of the capital stock of the Corporation.

Section 2. <u>Reserve Fund</u>. The Board of Directors may set aside out of any funds of the Corporation available for dividends a reserve or reserves for any proper purposes and in such sum or sums as the directors from time to time, in their absolute discretion, believe to be proper, and the Board of Directors may abolish any such reserve.

ARTICLE X Notice

Section 1. <u>Waiver of Notice</u>. Whenever notice is required by the Certificate of Incorporation, the Bylaws, or as otherwise provided by law, a written waiver thereof, signed by the person entitled to notice, shall be deemed equivalent to notice, whether before or after the time required for such notice. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting except when the person attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the stockholders, directors, or members of a committee of directors need be specified in any written waiver of notice.

Section 2. <u>Mailing of Notice</u>. Whenever under the provisions of these Bylaws notice is required to be given to any director, officer or shareholder and such notice is not waived as provided in Section 1 of this Article X, it shall not be construed to mean personal notice, but such notice may be given in writing, by mail, by depositing the same in the post office or letter box, in post- paid sealed wrapper, addressed to such shareholder, officer or director at such address as appears on the books of the Corporation, or, in default of other address, to such director, officer or shareholder at the General Post Office in Oklahoma City, Oklahoma, and such notice shall be deemed to be given at the time when the same shall be thus mailed.

<u>ARTICLE XI</u> <u>Amendment of Bylaws</u>

Section 1. <u>Amendment</u>. These Bylaws may be made, amended, altered, added to, revised or repealed only by a vote of a majority of the directors then in office or by a vote of the holders of two-thirds of the issued and outstanding shares of stock of the Corporation entitled to vote for the election of directors; provided, however, that Article IV, Section 1 of these Bylaws and this Article XI, Section 1, may be amended, altered, added to, revised or repealed only by a vote of two-thirds of the entire Board of Directors or by a vote of two-thirds of the issued and outstanding shares of stock of the Corporation entitled to vote for the election of Directors.

<u>ARTICLE XII</u> Exclusive Jurisdiction for Certain Claims

Section 1. Exclusive Forum. Unless the Board of Directors or one of its committees otherwise approves the selection of an alternate forum, the Court of Chancery of the State of Delaware (or, if, and only if, the Court of Chancery of the State of Delaware dismisses a Covered Claim (as defined below) for lack of subject matter jurisdiction, any other state or federal court in the State of Delaware that does have subject matter jurisdiction) shall, to the fullest extent permitted by applicable law, be the sole and exclusive forum for any (i) derivative claim brought in the right of the Corporation, (ii) claim asserting a breach of a fiduciary duty to the Corporation or the Corporation arising pursuant to any provision of the General Corporation Law of the State of Delaware, this Second Amended and Restated Bylaws, (iv) claim to interpret, apply, enforce or determine the validity of this Second Amended and Restated Bylaws, (iv) claim against the Corporation asserting one or more of the claims specified in clauses (i) through (v) herein above (each a "Covered Claim"); provided, however, that the provisions of this Article XII(1) will not apply to claims brought to enforce any liability or duty created by the Securities Exchange Act of 1934, as amended.

Section 2. Personal Jurisdiction. If any person or entity (a "Claiming Party") files an action asserting a Covered Claim in a court other than one determined in accordance with Section 1 above (each a "Foreign Action") without the prior approval of the Board of Directors or one of its committees, such Claiming Party shall be deemed to have consented to (i) the personal jurisdiction of the court determined in accordance with Section 1 in connection with any such action brought in any such court to enforce Section 1 (an "Enforcement Action") and (ii) having service of process made upon such Claiming Party in any such Enforcement Action by service upon such Claiming Party's counsel in the Foreign Action as agent for such Claiming Party.

Section 3. Notice and Consent. Any person or entity purchasing or otherwise acquiring any interest in the shares of capital stock of the Corporation shall be deemed to have notice of and consented to the provisions of this Article XII and waived any argument relating to the inconvenience of the forums referenced above in connection with any Covered Claim. Section 4. Federal Forum. Unless the Corporation consents in writing to the selection of an alternative forum, the federal district courts of the United States shall be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended. Any person or entity purchasing or otherwise acquiring any interest in any security of the Corporation shall be deemed to have notice of and consented to this provision.

These Amended and Restated Bylaws of LSB Industries, Inc. amend, restate and set forth the entire Bylaws of the Corporation and have been approved by the Board of Directors of the Corporation as of the day of August, 2009_____20___.

LSB INDUSTRIES, INC.

<u>Mark T. Behrman</u> <u>Chief Executive Officer and President</u> Jack E. Golsen Chairman of the Board

David M. ShearMichael J. Foster, Secretary

LSB INDUSTRIES, INC.

SECURITIES EXCHANGE AGREEMENT

This Securities Exchange Agreement (this "*Agreement*") is made as of July 19, 2021, by and between LSB Industries, Inc., a Delaware corporation (the "*Company*"), and LSB Funding LLC, a Delaware limited liability company (the "*Holder*"). Capitalized terms not otherwise defined herein have the meanings given to such terms in <u>Section 1.4</u> below.

WHEREAS, the Company, the Holder and, solely for purposes of Section 7.12 thereunder, Security Benefit Corporation, a Kansas corporation, entered into that certain Securities Purchase Agreement, dated as of December 4, 2015 (the "*Purchase Agreement*"), pursuant to which, in relevant part, the Holder purchased 210,000 shares of the Company's Series E Cumulative Redeemable Class C Preferred Stock (the "*Series E Preferred*") and one (1) share of the Company's Series F Redeemable Class C Preferred Stock (the "*Series F Preferred*").

WHEREAS, on September 19, 2016, the Company redeemed 70,232 shares of Series E Preferred, with the result that 139,768 shares of Series E Preferred remained outstanding and held by the Holder.

WHEREAS, pursuant to that certain Securities Exchange Agreement dated October 18, 2018, the Holder and the Company exchanged the Holder's (i) Series E Preferred for newly issued shares of Series E-1 Cumulative Redeemable Class C Preferred Stock (the "*Series E-1 Preferred*") and (ii) Series F Preferred for a newly issued share of Series F-1 Redeemable Class C Preferred Stock (the "*Series F-1 Preferred*" and together with the Series E-1 Preferred, the "*Existing Preferred*"), in each case on a one-for-one basis.

WHEREAS, in connection with the exploration of certain potential strategic transactions, including strategic transactions that potentially could be with entities with which members of the Board of Directors of the Company may be affiliated or associated, on September 14, 2020 the Board of Directors of the Company formed a special committee of the Board of Directors of the Company comprised of independent and disinterested directors (the "*Special Committee*") to, among other things, (a) conduct an independent review of certain potential strategic transactions and make an independent determination with respect thereto; (b) evaluate strategic alternatives that may otherwise be available to the Company; and (c) negotiate or direct negotiations; and (d) take all other actions related to such potential strategic transactions and any alternatives as the special committee may deem necessary or appropriate in order for the special committee to discharge its duties.

WHEREAS, in connection with the foregoing exploration of potential strategic transactions, the Company determined that it would be beneficial for the Holder and the Company to exchange the Holder's Existing Preferred for common stock, par value \$0.10, of the Company ("*Common Stock*").

WHEREAS, although the earlier explored strategic transactions which were under consideration were not consummated, the Company determined that it would be beneficial to proceed with the Exchange (as defined below) independent of such strategic transactions, and the Company and the Holder desire to proceed with the Exchange, subject to the terms and conditions set forth herein.

WHEREAS, following the Special Committee's review of alternative transactions and its determination that the Exchange and the other transactions contemplated by this Agreement (the "*Transactions*") maximize value to the Company's Stockholders, the Special Committee has reviewed and approved the terms of the Transactions and has recommended that the Company's Stockholders vote in favor of the Company Stockholder Approval (the "*Special Committee Recommendation*").

WHEREAS, the Board of Directors of the Company, in consideration of, among other things, the Special Committee Recommendation, also recommended that the Company's Stockholders vote in favor of the Company Stockholder Approval (the "*Board Recommendation*").

WHEREAS, the Board of Directors of the Company has determined for purposes of the Section 382 Rights Agreement, dated as of July 6, 2020 between the Company and Computershare Trust Company, N.A., as Rights Agent (the "*Rights Agreement*"), that the Transaction is an "Exempt Transaction" (as defined in the Rights Agreement), since the Transaction is not expected to have any adverse impact upon the Company's utilization of its net operating losses.

WHEREAS, the Holder has conditioned consummation of the Transactions contemplated by this Agreement on the review and approval of the Transactions by the Special Committee, and on the Company obtaining the Company Stockholder Approval.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

1. EXCHANGE OF EXISTING PREFERRED FOR COMMON STOCK.

1.1 Exchange of Existing Preferred. On the terms and subject to the conditions of this Agreement, the Holder hereby agrees that, effective at and contingent upon the Closing, (a) the 139,768 shares of Series E-1 Preferred held by the Holder as of immediately prior to the Closing shall be automatically exchanged for a number of shares, rounded down to the nearest whole share, of Common Stock equal to the aggregate Liquidation Preference (as defined in the Series E-1 Certificate of Designations) of such shares of Series E-1 Preferred as of the Closing Date, divided by the 30-Day VWAP and (b) the one (1) share of Series F-1 Preferred held by the Holder as of immediately prior to the Closing shall be automatically exchanged for a number of shares, rounded down to the nearest whole share, of Common Stock equal to the Redemption Price (as defined in the Series F-1 Certificate of Designations), divided by the 30-Day VWAP; provided, however, that the aggregate number of shares of Common Stock issuable pursuant to clauses (a) and (b) above shall be reduced by the number of shares of Common Stock that the Holder will receive in respect of the payment of the Dividend on the Common Stock held by the Holder on the Record Date (such net number of shares of Common Stock to be issued pursuant to this Section 1.1, the "Issued Shares" and such transactions, the "Exchange"). In accordance with the foregoing, and solely for purposes of illustration, if the Closing were to occur as of the date of this Agreement, the Holder would receive, collectively, 47,761,401 shares of Common Stock pursuant to the Exchange. All Existing Preferred surrendered for exchange shall no longer be deemed to be outstanding and all rights with respect to such Existing Preferred shall immediately cease and terminate at the Closing, except for the right of the Holder to receive the Issued Shares in exchange for the Existing Preferred (notwithstanding the failure of the Holder to surrender the certificates representing shares of Existing Preferred at or prior to the Closing or otherwise). From and after the date of this Agreement until the Closing or earlier termination of this Agreement in accordance with its terms, Holder shall not either directly or indirectly purchase or acquire, or cause any of its Affiliates to purchase or acquire, any stock or equity securities of the Company; provided that, this Section 1.1 shall not prohibit any such acquisition that is (a) the result of a dividend, distribution, issuance, sub-division (including split-up, recapitalization or otherwise) reclassification, combination or exchange of shares, separation, reorganization, merger, consolidation, liquidation, or any other transaction, circumstance or event approved by the Board of Directors of the Company or (b) made with the prior written consent of the Company.

1.2 Closing; Delivery.

1.2.1 Subject to the terms and conditions hereof, the consummation of the Exchange (the "*Closing*") shall take place (i) remotely via the electronic exchange of the applicable documents and signatures at 9:00 a.m. (New York City time) on the date that is three (3) Business Days after the satisfaction or waiver by the appropriate party (to the extent permitted by applicable Law) of the conditions precedent set forth in <u>Section 1.3</u> (except for those conditions that by their nature are to be satisfied by the delivery of documents or taking of actions at the Closing, but subject to the satisfaction of such conditions at the Closing or, if permissible, waiver in writing by the party hereto entitled to the benefit of those conditions at the Closing) or (ii) such other date, time and place as may be agreed in writing by the parties (such day of Closing, the "*Closing Date*").

1.2.2 At the Closing, the Holder shall deliver its certificates representing the 139,768 shares of Series E-1 Preferred and one (1) share of Series F-1 Preferred (or, as applicable, Lost Security Documentation in respect thereof) and, in exchange therefor, at the Closing, the Issued Shares shall be issued to the Holder and held in book-entry form by Computershare Trust Company, N.A., in the name of the Holder.

1.2.3 At the Closing, the Company shall deliver or cause to be delivered (unless waived by the Holder) to the Holder, a certificate executed on the Company's behalf by the Secretary of the Company, certifying as to and attaching resolutions of the Board of Directors of the Company and the Special Committee authorizing the execution and delivery by the Company of the Transaction Documents and the consummation of the transactions contemplated thereby, including the issuance of the Issued Shares.

1.3 Conditions to Closing.

1.3.1 <u>Company's Conditions</u>. The obligation of the Company to consummate the Transactions at the Closing shall be subject to the satisfaction at the Closing of each of the following conditions (any or all of which may be waived by the Company in writing, in whole or in part, to the extent permitted by applicable Law):

(a) (i) the representations and warranties of the Holder set forth in <u>Section 3.1</u>, <u>Section 3.2</u>, <u>Section 3.3</u>, <u>Section 3.4(a)</u> and <u>Section 3.5</u> (collectively, the "*Holder Fundamental Representations*") shall be true and correct in all respects at and as of the Closing Date as if made at and as of such time (except to the extent expressly made as of an earlier date, in which case as of such earlier date), and (ii) all other representations and warranties set forth in <u>Section 3</u> shall be true and correct in all respects at and as of the Closing Date as if made at and as of such time (except to the extent expressly made as of such earlier date), except, in the case of this clause (ii), where the failure of such representations and warranties to be true and correct in all respects (without giving effect to any qualifications as to "materiality", "Holder Material Adverse Effect" or other similar qualifications as to materiality) does not have, and would not reasonably be expected to have, individually or in the aggregate, a Holder Material Adverse Effect;

(b) the Holder shall have complied in all material respects with all of its obligations under this Agreement to be performed on or prior to the Closing;

(c) the Holder shall have delivered, or caused to be delivered, to the Company at or prior to the Closing the following closing

deliveries:

(i) consent to the termination of the Contracts set forth on <u>Schedule 1.3.1(c)(i)</u> (the "*Related Party Contracts*"), duly executed by an appropriate representative of the Holder;

(ii) a certificate, dated as of the Closing Date, executed on the Holder's behalf by a manager or authorized signatory of the Holder, certifying as to the matters set forth in <u>Section 1.3.1(a)</u> and <u>Section 1.3.1(b)</u>;

(iii) a cross-receipt, dated as of the Closing Date, duly executed by the Holder, certifying that it has received from the Company the certificates evidencing the Issued Shares; and

(iv) an amendment to Section 3 of the Existing Standstill Agreement duly executed by the Purchaser Parties and the Golsen Holders (each, as defined therein) in the form attached hereto as <u>Exhibit A</u> (the "*Standstill Amendment*").

1.3.2 <u>Holder's Conditions</u>. The obligation of the Holder to consummate the Transactions at the Closing shall be subject to the satisfaction at the Closing of each of the following conditions (any or all of which may be waived by the Holder in writing, in whole or in part, to the extent permitted by applicable Law):

(a) (i) the representations and warranties of the Company set forth in Section 2.1, Section 2.2, Section 2.3(i), Section 2.4, and Section 2.26 (collectively, the "Company Fundamental Representations"), Section 2.5 and Section 2.8, shall be true and correct in all respects at and as of the Closing Date as if made at and as of such time (except to the extent expressly made as of an earlier date, in which case as of such earlier date) and (ii) all other representations and warranties set forth in Section 2 shall be true and correct in all respects at and as of the Closing Date as if made at and as of an earlier date, in which case as of such earlier date), except, in the case of this clause (ii), where the failure of such representations and warranties to be true and correct in all respects (without giving effect to any qualifications as to "materiality", "Material Adverse Effect" or other similar qualifications as to materiality) does not have, and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect;

(b) the Company shall have complied in all material respects with its obligations under this Agreement to be performed on or prior to the Closing;

to the Closing

(c) the Company shall have delivered, or caused to be delivered, to the Holder at or prior to the Closing the following closing deliveries:

(i) evidence of issuance as of the Closing Date by book entry (which may be in the form of a screen shot of the transfer agents records or other customary record) of the Issued Shares, bearing restrictive notations meeting the applicable requirements of this Agreement;

(ii) a certificate, dated as of the Closing Date, executed on the Company's behalf by the Chief Executive Officer or the Chief Financial Officer of the Company, or other officer of the Company reasonably satisfactory to the Holder, certifying as to the matters set forth in Section 1.3.2(a) and Section 1.3.2(b);

(iii) a cross-receipt, dated as of the Closing Date, duly executed by the Company certifying that it has received from the Holder of the Company the certificates evidencing the Existing Preferred;

Company;

(iv) consent to the termination of the Related Party Contracts, duly executed by the appropriate representative of the

(v) the Standstill Amendment duly executed by the Company and the Golsen Holders (each, as defined in the Existing

Standstill Agreement);

(d) since the date hereof, there shall not have been any event, occurrence or development of a state of circumstances that (individually or in the aggregate) has had or would reasonably be expected to have a Material Adverse Effect;

(e) the Company shall have declared the Dividend in accordance with <u>Section 4.5;</u>

(f) the Company shall have filed with the Delaware Secretary of State, an amendment to the Charter reflecting (i) the Charter Amendment and (ii) the Series E-1 Charter Amendment, prior to the Closing, which shall continue to be in full force and effect as of the Closing;

(g) the Issued Shares shall have been approved for listing on the NYSE, subject to official notice of issuance;

(h) the Company shall have prepared any and all documentation necessary for compliance with the provisions set forth in the Company's existing senior notes resulting from the Transactions, which documentation shall be reasonably satisfactory to the Holder and ready for distribution to the Company's existing noteholders immediately following the Closing; and

(i) the opinion described in <u>Section 2.32</u> shall have been delivered to the Special Committee immediately prior to the execution and delivery of this Agreement and shall not have been withdrawn or otherwise modified prior to the Closing.

1.3.3 <u>Mutual Conditions</u>. The respective obligations of each party to consummate the Transactions at the Closing shall be subject to the satisfaction on or prior to the Closing of each of the following conditions:

(a) no temporary restraining order, preliminary or permanent injunction or other judgment or Order issued by any Governmental Authority of competent jurisdiction or other Law, shall be pending, threatened or in effect which, in each case, enjoins, prohibits, restrains or renders (or seeks to enjoin, prohibit, restrain or render) illegal the consummation of the Transactions contemplated hereby or would cause such Transactions to be rescinded;

(b) any applicable waiting period under the HSR Act relating to the Transactions will have expired or been terminated;

(c) receipt of the waivers, permits, consents, approvals or other authorizations which are necessary for the consummation of the Transactions set forth on <u>Schedule 1.3.3(c)</u>, including the receipt from the required lenders under the Credit Agreement of a consent or waiver of any Change of Control under the Credit Agreement resulting from the Transactions, in form and substance satisfactory to each party to this Agreement; and

(d) the Company Stockholder Approval has been obtained.

1.4 Defined Terms Used in this Agreement. In addition to the terms defined elsewhere in this Agreement, the following terms used in this Agreement shall be construed to have the meanings set forth or referenced below.

"**30-Day VWAP**" means \$6.16, which represents the volume weighted average price of the Common Stock for the 30 trading days immediately prior to July 19, 2021 on the NYSE as reported by Bloomberg Financial Markets or, if Bloomberg Financial Markets is not then reporting such prices, by a comparable reporting service of national reputation mutually selected by the Company and Holder.

"Acquisition Proposal" means, other than the Transactions, any offer or proposal of any Third Party or "group" (within the meaning of Section 13(d)(3) of the Exchange Act) relating to (i) any acquisition or purchase, direct or indirect, of assets equal to 20% or more of the consolidated assets of the Company or to which 20% or more of the consolidated revenues or earnings of the Company are attributable or 20% or more of any class of equity or voting securities of the Company, (ii) any tender offer or exchange offer that, if consummated, would result in such Third Party or "group" beneficially owning 20% or more of any class of equity or voting securities of the Company, or (iii) a merger, consolidation, statutory share exchange, business combination, sale of all or substantially all of the assets, liquidation, dissolution or other similar extraordinary transaction involving the Company or any of its Subsidiaries whose assets, individually or in the aggregate, constitute 20% or more of the consolidated assets of the Company or to which 20% or more of the consolidated revenues or earnings of the Company are attributable.

"Adverse Recommendation Change" means the occurrence of any of the following: (i) the Board of Directors of the Company or the Special Committee has failed to make, has withdrawn or has changed, qualified or modified in a manner adverse to the Holder, the Board Recommendation or the Special Committee Recommendation (as applicable) (or publicly recommended an Acquisition Proposal) or publicly proposed to do any of the foregoing, (ii) the Board of Directors of the Company or the Special Committee has approved or adopted any Acquisition Proposal, has recommended the approval or adoption of, or publicly proposed to recommend, approve or adopt, any Acquisition Proposal, (iii) the Board of Directors of the Company or the Special Committee Recommendation or the Board Recommendation in the Company Proxy Statement, (iv) if any Acquisition Proposal has been made public, fails to reaffirm the Special Committee Recommendation or the Board Recommendation (as applicable) upon written request by the Holder within five (5) Business Days following the date that the Holder makes such request, or (v) if a tender offer or exchange offer for shares of capital stock of the Company that constitutes an Acquisition Proposal is commenced, fails to recommend against acceptance of such tender offer or exchange offer by the stockholders of the Company; provided that a customary "stop, look and listen" communication by the Board of Directors of the Company pursuant to Rule 14d-9 of the Exchange Act shall not be prohibited within ten (10) Business Days after commencement of such tender offer or exchange).

"*Affiliate*" means with respect to any Person, any other Person that directly or indirectly through one or more intermediaries controls, is controlled by or is under common control with, the Person in question; <u>provided</u>, that, for purposes of <u>4.12</u>, <u>4.13.1</u>, <u>4.13.2</u> and <u>4.14.1</u>, commonly controlled portfolio companies of any Person shall not be considered "Affiliates". As used herein, the term "control" means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through ownership of voting securities, by contract or otherwise. For the avoidance of doubt, for purposes of this Agreement, the Company, on the one hand, and the Holder, on the other, shall not be considered Affiliates of each other.

"Agreement" has the meaning set forth in the Preamble.

"Amended Registration Rights Agreement" has the meaning set forth in Section 7.12.

"Anti-Corruption Laws" means all Laws of any jurisdiction applicable to the Company or its Subsidiaries from time to time concerning or relating to bribery or corruption (including U.S. Foreign Corrupt Practices Act (the "FCPA") and the U.K. Bribery Act of 2010).

"Antitrust Laws" means the Sherman Act, 15 U.S.C. §§ 1-7, as amended; the Clayton Act, 15 U.S.C. §§ 12-27, 29 U.S.C. §§ 52-53, as amended; the HSR Act; the Federal Trade Commission Act, 15 U.S.C. § 41-58, as amended; and all other federal, state and foreign statutes, rules, regulations, Orders, decrees, administrative and judicial doctrines, and other Laws that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or competition.

"Board Recommendation" has the meaning set forth in the Recitals.

"*Business Day*" means any day other than a Saturday, Sunday, any federal legal holiday or day on which banking institutions in the state of New York are authorized or required by Law or other governmental action to close.

"Cap" means \$35,000,000.

"Capital Budget" means the annual capital budget of the Company and its Subsidiaries approved by the Board of Directors for any given year.

"*Charter*" means the Company's Amended and Restated Certificate of Incorporation, dated January 21, 1977, as amended on August 27, 1987, including the Series E-1 Certificate of Designations and the Series F-1 Certificate of Designations, and any other certificates of designations of preferred stock of the Company, in each case as in effect, as amended from time to time.

"*Claims*" has the meaning set forth in <u>Section 6.1</u>.

"Common Stock" has the meaning set forth in the Recitals.

"Company" has the meaning set forth in the Preamble.

"Company Disclosure Schedules" has the meaning set forth in Section 2.

"*Company Documents*" means all Contracts to which the Company or any of its Subsidiaries is a party or by which the Company or any of its Subsidiaries is bound or to which any of the property or assets of the Company or any of its Subsidiaries is subject.

"Company Financial Statements" has the meaning set forth in Section 2.7.

"Company Fundamental Representations" has the meaning set forth in Section 1.3.2(a).

"*Company Proxy Statement*" means the proxy statement of the Company to be filed with the SEC in connection with obtaining the Company Stockholder Approval and any amendments or supplements thereto.

"Company Related Parties" has the meaning set forth in Section 6.2.

"Company SEC Documents" has the meaning set forth in Section 2.7.

"Company Stock Option Plan" means the 2016 Long Term Incentive Plan adopted by the Company, effective as of April 16, 2016.

"Company Stockholder Approval" means (i) the approval of the stockholders of the Company other than Holder and any Affiliates of the Holder ("Disinterested Stockholders"), holding greater than 50% of the voting power of the outstanding shares of capital stock of the Company entitled to vote thereon held by the Disinterested Stockholders of (x) the Transactions and (y) this Agreement; (ii) the approval and adoption of holders of greater than 50% of the voting power of the outstanding shares of capital stock of the Company entitled to vote thereon of an amendment to the Charter to increase the total number of authorized shares of Common Stock to 150,000,000 shares of Common Stock (the "Charter Amendment"); (iii) the approval of the stockholders of the Company, by a majority of the votes cast, of the Transactions and this Agreement pursuant to Section 312.03 of the New York Stock Exchange Listed Company Manual; and (iv) the approval of holders of greater than 50% of the company entitled to vote thereon and greater than 50% of the outstanding shares of series E-1 Preferred entitled to vote thereon of an amendment to the Charter to receive any dividend or other payments thereunder in connection with the Dividend and to remove the right of holders of Series E-1 Preferred to receive any dividend or other payments thereunder in connection with the Dividend and a waiver of any dividends otherwise payable in respect of such Series E-1 Preferred, in form reasonably acceptable to the Holder and the Company (the "Series E-1 Charter Amendment").

"Company Stockholder Meeting" has the meaning set forth in Section 4.3.1.

"Confidentiality Agreement" means the Confidentiality Agreement by and between the Company and the Holder, dated July 16, 2020.

"*Contracts*" means any contract, indenture, mortgage, deed of trust, loan or credit agreement, bond, note, debenture, evidence of Indebtedness, swap agreement, lease, license or other instrument or agreement.

"*COVID-19*" means the novel coronavirus, SARS-CoV-2 or COVID-19 (and all related strains and sequences), including any intensification, resurgence or any evolutions or mutations thereof, and/or related or associated epidemics, pandemics, disease outbreaks or public health emergencies.

"*COVID-19 Measures*" means any applicable quarantine, "shelter in place," "stay at home," workforce reduction, social distancing, shut down, closure, sequester or any other applicable Law, Order, directive, guideline or requirement by an applicable Governmental Authority in connection with or in response to the COVID-19 pandemic, including the Coronavirus Aid, Relief, and Economic Security Act (CARES).

"Credit Agreement" means that certain Third Amended and Restated Loan and Security Agreement, dated as of January 17, 2017, by and among Wells Fargo Capital Finance, LLC, the lenders party thereto, the Company, certain subsidiaries of the Company as borrowers thereto and certain subsidiaries of the company designated as guarantors thereto, as amended by that First Amendment, dated as of April 16, 2018, that Second Amendment, dated as of February 26, 2019 and that Third Amendment, dated as of April 20, 2020 (as may be amended, restated or otherwise modified from time to time).

"Deductible" has the meaning set forth in Section 6.3.1.

"DGCL" means the Delaware General Corporation Law.

"Dividend" has the meaning set forth in Section 4.5.

"Environmental Laws" has the meaning set forth in Section 2.18.

"*ERISA*" means the Employee Retirement Income Security Act of 1974, as amended, and the regulations and published interpretations thereunder.

"Exchange" shall have the meaning set forth in Section 1.1.

"Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, and the rules and regulations of the Commission promulgated thereunder.

"Excluded Claims" has the meaning set forth in Section 6.4.

"Existing Preferred" has the meaning set forth in the Recitals.

"*Existing Standstill Agreement*" means the Board Representation and Standstill Agreement by and among the Company, Holder, and the other parties thereto, dated December 4, 2015, as amended.

"Financial Advisors" has the meaning set forth in Section 2.26.

"Fundamental Representations" means the Holder Fundamental Representations and the Company Fundamental Representations.

"GAAP" means United States generally accepted accounting principles.

"Governmental Authority" means, with respect to a particular Person, (i) any country, state, county, city and political subdivision, (ii) any selfregulatory organization or other non-governmental regulatory authority or quasi-governmental authority, or (iii) any court or other tribunal, in each case that exercises valid jurisdiction over any such Person or such Person's Property, and any court, agency, department, commission, board, bureau or instrumentality of any of them and any monetary authority which in each case exercises valid jurisdiction over any such Person or such Person's Property.

"Governmental Licenses" has the meaning set forth in Section 2.28.

"Hazardous Materials" has the meaning set forth in Section 2.18.

"HMT" has the meaning set forth in Section 2.25.

"*Holder*" has the meaning set forth in the Preamble.

"Holder Fundamental Representations" has the meaning set forth in Section 1.3.1(a).

"Holder Material Adverse Effect" means any event that would prevent or materially delay Holder from consummating the Transactions or performing its obligations hereunder.

"Holder Related Parties" has the meaning set forth in Section 6.1.

"Houlihan Lokey" has the meaning set forth in Section 2.26.

"HSR Act" means the Hart Scott Rodino Antitrust Improvements Act of 1976, as amended.

"Indebtedness" means, with respect to any Person, without duplication, as of the date of determination (i) all obligations of such Person for borrowed money, including accrued and unpaid interest, and any prepayment fees or penalties, (ii) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (iii) all obligations of such Person issued or assumed as the deferred purchase price of property or services (including any potential future earn-out, purchase price adjustment, release of "holdback" or similar payment, but excluding obligations of such person incurred in the ordinary course of business consistent with past practice), (iv) all lease obligations of such Person required to be capitalized on the books and records of such Person and any purchase money indebtedness and similar financing arrangements, (v) all Indebtedness of others secured by a Lien on property or assets owned or acquired by such Person, whether or not the Indebtedness secured thereby have been assumed, (vi) all obligations of such Person under interest rate, currency or commodity derivatives or hedging transactions or similar arrangements (valued at the termination value thereof), (vii) all letters of credit or performance bonds issued for the account of such Person, to the extent drawn upon, (viii) all guarantees and keepwell arrangements of such Person of any Indebtedness of any other Person other than a wholly owned Subsidiary of such Person and (ix) all obligations of such Person in respect of any preferred stock (valued at the liquidation preference thereof plus any accrued and unpaid dividends).

"Indemnified Party" has the meaning set forth in Section 6.4.2.

"Indemnifying Party" has the meaning set forth in Section 6.4.2.

"IP Rights" has the meaning set forth in Section 2.27.

"Knowledge of the Company" means the actual knowledge of Mark Behrman, Cheryl Maguire and Michael Foster.

"*Law*" means any federal, state, local or foreign Order, settlement, award, statute, law (including common law), rule, regulation or other requirement or rule of law of any Governmental Authority.

"*Legal Proceeding*" means any private or governmental action, inquiry, claim, charge, complaint, demand, proceeding, suit, hearing, litigation, arbitration, mediation, audit or investigation, in each case whether civil, criminal, administrative, judicial or investigative, or any appeal therefrom.

"*Lien*" means, with respect to any property or asset, any mortgage, deed of trust, pledge, hypothecation, security, interest, encumbrance, interference, option, right of first refusal, preemptive right, community property interest, claim, lien or restriction of any kind (including any restriction on the voting of any security, any restriction on the transfer of any security or other asset, or any restriction on the possession, exercise or transfer of any other attribute of ownership of any asset).

"Losses" has the meaning set forth in <u>Section 6.1</u>.

"*Lost Security Documentation*" means documentation reasonably satisfactory to the Company with regard to a lost or stolen certificate evidencing ownership of Existing Preferred including, if required by the Company, an affidavit of lost security in favor of the Company with respect to such lost or stolen security.

"Material Adverse Effect" means (i) a materially adverse effect on the business, assets, liabilities, results of operations and condition (financial or otherwise) of the Company and its Subsidiaries, taken as a whole, or (ii) any event that would prevent or materially delay the Company from consummating the Transactions or performing its obligations hereunder, other than, solely in the case of clause (i), to the extent caused by, arising out of or attributable to any of the following: (a) changes solely attributable to the announcement or pendency of this Agreement, or any litigation by any Person with respect to this Agreement or the transactions contemplated hereby, (b) changes or proposed changes in applicable Law or accounting standards or interpretations thereof applicable to the Company and its Subsidiaries, (c) changes generally affecting the economy, credit or financial or capital markets, in the United States or elsewhere in the world, including changes in interest or exchange rates; (d) national or international political conditions, including acts of war (whether or not declared), armed hostilities, sabotage, military actions or the escalation thereof (whether or not declared), and terrorism; (e) changes generally applicable in the industries in which the Company and its Subsidiaries operate, (f) changes resulting from compliance by the Company and its Subsidiaries with the express terms of this Agreement, (g) any changes as a result of any action expressly approved by the Holder in writing, (h) any failure of the Company to meet internal or analysts' estimates, projections, forecasts, guidance, milestones or budgets or internal or analysts' financial or operating predictions of revenues, earnings, cash flows or cash positions, or other financial or business metrics (it being understood that any cause of any such failure may be taken into consideration when determining whether a Material Adverse Effect has occurred or would be reasonably likely to occur), (i) changes in the market price, or trading volume, of the Common Stock (it being understood that any cause of any such change may be taken into consideration when determining whether a Material Adverse Effect has occurred or would be reasonably likely to occur), or (j) any embargo, riot, earthquake, hurricane, tsunami, tornado, flood, mudslide, wild fire, other weather-related or meteorological event, pandemic (including the COVID-19 pandemic and any COVID-19 Measures), epidemic, disease outbreak or other natural disaster or act of god; provided, that, in the case of clauses (b), (c), (d), (e) and (j), that such conditions or changes do not have a materially disproportionate impact on the Company and its Subsidiaries, taken as a whole, relative to other participants in the industries in which the Company and its Subsidiaries operate.

"Money Laundering Laws" has the meaning set forth in Section 2.24.

"*New Common Stock*" has the meaning set forth in <u>Section 4.13.1</u>.

"*NYSE*" means the New York Stock Exchange.

"OFAC" means the Office of Foreign Assets Control.

"Options or Convertible Securities" has the meaning set forth in Section 4.13.

"Order" means, with respect to any Person, any injunction, order, writ, decree, consent decree, judgment, ruling, verdict, stipulation, determination or award entered, issued, made or rendered by any Governmental Authority of competent jurisdiction affecting such Person or any of its Properties.

"Outside Date" has the meaning set forth in Section 5.2.1.

"Per Claim Threshold" has the meaning set forth in Section 6.3.1.

"Permitted Lien" means, collectively, (i) suppliers', mechanics', carriers', workmen's, repairmen's, materialmen's, warehousemen's, construction and other similar Liens arising or incurred by operation of law or otherwise incurred in the ordinary course of business securing obligations that are not due and payable or which are being contested in good faith by appropriate proceedings and for which adequate accruals or reserves have been established in accordance with GAAP, (ii) Liens for taxes, utilities and other governmental charges that are not due and payable or which are being contested in good faith by appropriate proceedings and for which adequate accruals or reserves have been established in accordance with GAAP, (iii) requirements and restrictions of zoning, building and other applicable Laws and municipal by-laws, and development, site plan, subdivision or other agreements with municipalities that do not materially detract from the value of such property or materially interfere with the business of the Company and its Subsidiaries as currently conducted, (iv) licenses or other grants of rights in intellectual property made in the ordinary course of business, consistent with past practices, (v) statutory Liens of landlords for amounts not due and payable or which are being contested in good faith by appropriate proceedings and for which adequate accruals or reserves have been established in accordance with GAAP. (vi) deposits made in the ordinary course of business to secure payments of worker's compensation, unemployment insurance or other types of social security benefits or the performance of bids, tenders, sales, contracts (other than for the repayment of borrowed money), public or statutory obligations, and surety, stay, appeal, customs or performance bonds, or similar obligations arising in each case in the ordinary course of business consistent with past practices, (vii) Liens in favor of customs and revenue authorities arising as a matter of law and in the ordinary course of business, consistent with past practices, to secure payment of customs duties in connection with the importation of goods, (viii) Liens resulting from securities laws, (ix) licenses granted to third parties in the ordinary course of business consistent with past practices by the Company or its Subsidiaries, (x) such other Liens or imperfections of title that are not material in amount and do not materially detract from the value of or materially impair the existing use of the property affected by such Lien or imperfection of title, (xi) Liens that do not materially detract from the value of such property or interfere in any material respect with the use, operation or occupancy by the Company or any of its Subsidiaries of such property and (xii) Liens arising under any Indebtedness incurred in accordance with Section 4.1.1.

"*Person*" means any individual, corporation, company, voluntary association, partnership, joint venture, trust, limited liability company, unincorporated organization, government or any agency, instrumentality or political subdivision thereof or any other form of entity.

"Plan" has the meaning set forth in Section 2.21.

"Preemptive Right Notice" has the meaning set forth in Section 4.13.2.

"*Property*" means any interest in any kind of property or asset, whether real, personal or mixed, or tangible or intangible (including intellectual property rights).

"**Pro Rata Share**" means a fraction (expressed as a percentage), the numerator of which equals the number of shares of Common Stock held by the Holder or any Affiliate thereof (including, for the avoidance of doubt, the Issued Shares, and assuming the conversion in full to Common Stock of any preferred stock of the Company held by the Holder or any Affiliate thereof) and the denominator of which equals the total number of issued and outstanding shares of Common Stock outstanding at such time (calculated on a fully diluted basis treating any options or convertible securities as exercised or converted).

"Purchase Agreement" has the meaning set forth in the Recitals.

"*Record Date*" has the meaning set forth in <u>Section 4.5</u>.

"*Registration Rights Agreement*" means that certain Registration Rights Agreement, by and between the Company and the Holder, dated as of December 4, 2015, as amended from time to time.

"Related Party Contracts" has the meaning set forth in Section 1.3.1(c)(i).

"Representative" means, with respect to a specified Person, the investors, officers, directors, managers, employees, agents, advisors, counsel, accountants, investment bankers and other representatives of such Person.

"Required Approval" has the meaning set forth in Section 2.5.

"Rights Agreement" has the meaning set forth in the Recitals.

"*Sanctioned Person*" means, at any time, (i) any Person listed in any Sanctions-related list of designated Persons maintained by the U.S. Government, the UNSC, the European Union, HMT or other relevant sanctions authority, (ii) any Person operating, organized or resident in a Sanctioned Country or (iii) any Person controlled by any such Person.

"Sanctions" has the meaning set forth in Section 2.25.

"SEC" means the Securities and Exchange Commission.

"Section 382" has the meaning set forth in Section 4.6.

"*Securities Act*" means the United States Securities Act of 1933, as amended, and the rules and regulations promulgated by the Securities and Exchange Commission thereunder.

"Series E Preferred" has the meaning set forth in the Recitals.

"Series E-1 Certificate of Designations" means the Certificate of Designations of Series E-1 Cumulative Redeemable Class C Preferred Stock of the Company, dated as of October 18, 2018.

"Series E-1 Preferred" has the meaning set forth in the Recitals.

"Series F Preferred" has the meaning set forth in the Recitals.

"Series F-1 Certificate of Designations" means the Certificate of Designations of Series F-1 Cumulative Redeemable Class C Preferred Stock of the Company, dated as of October 18, 2018.

"*Series F-1 Preferred*" has the meaning set forth in the Recitals.

"Special Committee" has the meaning set forth in the Recitals.

"Special Committee Recommendation" has the meaning set forth in the Recitals.

"Stockholder Litigation" has the meaning set forth in Section 4.2.2.

"*Subsidiary*" means, as to any Person, any corporation or other entity of which: (i) such Person or a Subsidiary of such Person is a general partner or managing member; (ii) at least a majority of the outstanding equity interests having by the terms thereof ordinary voting power to elect a majority of the board of directors or similar governing body of such corporation or other entity (irrespective of whether or not at the time any equity interest of any other class or classes of such corporation or other entity shall have or might have voting power by reason of the happening of any contingency) is at the time directly or indirectly owned or controlled by such Person or one or more of its Subsidiaries; or (iii) any corporation or other entity as to which such Person consolidates for accounting purposes.

"Superior Proposal" means a bona fide written Acquisition Proposal for at least a majority of the outstanding shares of capital stock of the Company or at least a majority of the consolidated assets of the Company and its Subsidiaries, in each case, for consideration consisting exclusively of cash and/or publicly traded equity securities (and for which financing, to the extent required by the Person submitting the written Acquisition Proposal, is then fully committed) that did not result from a breach of this Agreement, and that the Special Committee determines in good faith, after consultation with its financial advisor and outside legal counsel, and taking into account all relevant terms and conditions of such Acquisition Proposal, (i) would be, if consummated, more favorable to the Company's stockholders from a financial point of view than the Transactions and (ii) is not less likely to receive required governmental or other regulatory approvals on a timely basis than the Transactions and is otherwise reasonably likely to be consummated on the terms proposed.

"Termination Fee" has the meaning set forth in Section 5.4.1.

"Third Party" means any Person other than the Holder.

"Third Party Claim" has the meaning set forth in Section 6.4.2.

"*Transaction Documents*" means this Agreement (including the Schedules and Exhibits attached hereto), the Amended Registration Rights Agreement and the other documents, certificates and agreements delivered in connection with the Transactions.

"Transactions" means the Exchange and the other transactions contemplated by this Agreement.

"UNSC" has the meaning set forth in Section 2.25.

"Willful and Material Breach" means, with respect to any representation, warranty, agreement or covenant in this Agreement, a deliberate action or omission intentionally undertaken by the breaching party (i) where the breaching party has actual knowledge that such action or omission is or would result in a breach of such representation, warranty, agreement or covenant and (ii) such action or omission constitutes a material breach of this Agreement.

2. <u>REPRESENTATIONS AND WARRANTIES OF THE COMPANY</u>. Except (a) as disclosed in the Company SEC Documents (excluding in each case any disclosures set forth in the risk factors or "forward-looking statements" section of such Company SEC Documents or under any similar precautionary sections, and any other disclosures included therein to the extent they are predictive, cautionary or forward-looking in nature) or (b) the disclosure schedule delivered to the Holder prior to or concurrently with the execution of this Agreement (the "*Company Disclosure Schedule*"), it being agreed that disclosure of any item in any section or subsection of the Company Disclosure Schedule shall also be deemed disclosure with respect to any other section or subsection of the relevance of such item is reasonably apparent on its face, the Company hereby represents and warrants to the Holder (unless otherwise set forth herein, as of the date hereof) as set forth below:

2.1 <u>Formation and Qualification of the Company</u>. The Company has been duly incorporated and is validly existing and in good standing under the laws of the State of Delaware, and has the power and authority to own, lease and operate its properties and to conduct its business in all material respects in which it is engaged and to enter into and perform its obligations under the Transaction Documents to which it is a party. The Company is duly qualified to transact business and is in good standing in the state of its principal place of business and in each other jurisdiction in which such qualification is required, whether by reason of the ownership or leasing of property or the conduct of business, except (solely in the case of jurisdictions other than its principal place of business) where the failure so to qualify or to be in good standing would not, individually or in the aggregate, result in a Material Adverse Effect.

2.2 <u>Authority; Enforceability</u>. Except for the Company Stockholder Approval and the filing of the Charter Amendment and the Series E-1 Amendment with the Delaware Secretary of State, all corporate action on the part of the Company, its officers, directors, and stockholders necessary for the authorization, execution, delivery, and performance of all obligations of the Company under the Transaction Documents and the authorization, issuance, reservation for issuance, and delivery of the Issued Shares have been taken or shall be taken at or prior to the Closing; and this Agreement constitutes, and the other Transaction Documents when executed and delivered, shall constitute, valid and legally binding obligations of the Company, enforceable against it in accordance with their respective terms, except as may be limited by (a) applicable bankruptcy, insolvency, reorganization, or other Laws of general application relating to or affecting the enforcement of creditors' rights generally and (b) the effect of rules of law governing the availability of equitable remedies.

2.3 <u>Absences of Defaults; No Conflicts</u>. Neither the Company nor any of its Subsidiaries is (a) in violation of its organizational documents, (b) in violation of any Law applicable to the Company or any of its Subsidiaries or any of their respective assets, properties or operations or (c) in breach or default (or with or without the giving of notice or the passage of time or both, would be in breach or default) in the performance or observation of any obligation, agreement, covenant or condition contained in any Company Document, except in the case of clauses (b) or (c) for such violations, breaches or defaults that would not, individually or in the aggregate, result in a Material Adverse Effect. The execution, delivery and performance of this Agreement by the Company and the consummation by the Company of the Transactions (including the issuance of the Issued Shares) do not and will not (i) result in a violation of the Charter or Bylaws of the Company (as amended and restated), (ii) conflict with, or

constitute a default (or an event which with notice or lapse of time or both would become a default) under, or give to others any rights of termination, amendment, acceleration or cancellation of, any agreement, indenture or instrument to which the Company or any of its Subsidiaries is a party, or (iii) assuming the accuracy of the Holder's representations and warranties set forth in <u>Section 3</u> hereof, result in a violation of any Law or Order (including, without limitation, U.S. federal and state securities Laws) applicable to the Company, or by which any Property or asset of the Company is bound or affected, except, in the case of the foregoing clauses (ii) and (iii), for such conflicts, defaults, violations, terminations, amendments, accelerations, and cancellations as would not, individually or in the aggregate, have a Material Adverse Effect.

2.4 Capitalization and Valid Issuance of the Issued Shares; Ownership of Subsidiaries.

(a) As of the date hereof (prior to the consummation of the Exchange), the issued and outstanding capital stock of the Company consists of 30,305,076 shares of Common Stock, 20,000 shares of Series B cumulative, convertible preferred stock, \$100 par value, 1,000,000 shares of Series D cumulative Class C preferred stock no par value, 139,768 shares of Series E-1 Preferred and one (1) share of Series F-1 Preferred. All outstanding shares of the Company's capital stock have been duly authorized and validly issued in accordance with the Charter and are fully paid and nonassessable. Schedule 2.4(a) sets forth the authorized capital stock of the Company, solely for illustration purposes, as if the Closing took place on the date of this Agreement.

(b) Except (i) as provided for in the Charter, (ii) for 3,015,634 shares of Common Stock available for issuance under the Company Stock Option Plan, (iii) for options to acquire 49,475 shares of Common Stock issued and outstanding under the Company Stock Option Plan, (iv) as provided in the Purchase Agreement and (v) as provided in the Existing Standstill Agreement, there are no (1) preemptive rights or other rights to subscribe for or to purchase, nor any restriction upon the voting or transfer of, any equity securities of the Company or (2) outstanding options, warrants or other rights to purchase any securities of the Company.

(c) Subject to the adoption and filing of the Charter Amendment with the Secretary of State of the State of Delaware, the Issued Shares have been duly and validly authorized and when issued and delivered against the Existing Preferred pursuant to this Agreement, the Issued Shares will be duly and validly issued, fully paid and nonassessable.

(d) All of the issued and outstanding shares of capital stock of each Subsidiary of the Company that is a corporation, all of the issued and outstanding partnership interests of each Subsidiary of the Company that is a limited or general partnership and all of the issued and outstanding limited liability company interests, membership interests or other similar interests of each Subsidiary of the Company that is a limited liability company have been duly authorized and validly issued, are fully paid and (except in the case of general partnership interests) non-assessable and are owned by the Company, directly or through Subsidiaries, free and clear of any Lien (other than Permitted Liens) and restrictions on transfer set forth in the organizational documents of the relevant Subsidiary or under federal and state securities Laws. None of the issued and outstanding shares of capital stock of any such Subsidiary that is a corporation, none of the issued and outstanding partnership interests or other similar interests of any such Subsidiary that is a limited or general partnership, and none of the issued and outstanding limited liability company interests, membership interests or other similar rights of any such Subsidiary that is a limited liability company was issued in violation of any preemptive rights, rights of first refusal or other similar rights of any security holder of such Subsidiary or any other Person. The only Subsidiaries of the Company are the Subsidiaries listed on Schedule 2.4(d) hereto, and such schedule accurately sets forth whether each such Subsidiary is a corporation, limited or general partnership or limited liability company and the jurisdiction of organization of each such Subsidiary.

2.5 <u>Approvals</u>. Except for (i) the Company Stockholder Approval, (ii) Charter Amendment, (iii) Series E-1 Charter Amendment, (iv) as may be required under the Exchange Act, state securities Laws or "Blue Sky" Laws or the rules of the NYSE and (v) those waivers, permits, consents approvals and authorizations set forth on <u>Schedule 1.3.3(c)</u>, no permit, consent, approval, exemption, authorization, Order, registration, filing or qualification of or with any Governmental Authority or any other Person (each, a "*Required Approval*") is necessary or required in connection with the execution, delivery or performance by, or enforcement against, the Company of this Agreement or the Company's issuance of the Issued Shares, except for such Required Approvals (x) that have already been made or obtained or will be made or obtained contemporaneously with the Closing or (y) that, if not obtained, would not be material to the Company or any of its Subsidiaries or its ability to consummate the Transactions.

2.6 <u>Other Sales</u>. The Company has not sold or issued any securities that would be integrated with the issuance of the Issued Shares contemplated by this Agreement pursuant to the Securities Act.

2.7 <u>Company SEC Documents; Company Financial Statements</u>. The historical financial statements (including the related notes and supporting schedules) (the "*Company Financial Statements*") included in the Company's reports, required to be filed by it under the Exchange Act since January 1, 2020 (all such documents filed prior to the date hereof, collectively the "*Company SEC Documents*"), at the time filed (except to the extent corrected by a subsequent Company SEC Document) (a) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make statements therein, in light of the circumstances under which they were made, not misleading, (b) complied as to form with the applicable requirements of the Exchange Act and the Securities Act, (c) present fairly in all material respects the financial condition, results of operations and cash flows of the entities purported to be shown thereby at the dates and for the periods indicated and (d) have been prepared in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated in the notes thereto or, in the case of unaudited statements, as permitted by Form 10-Q mandated by the SEC). None of the Company SEC Documents, including any financial statements of the Company contained therein, contains any untrue statement of material fact or omits to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

2.8 <u>No Material Adverse Change</u>. Since December 31, 2020 (a) no event has occurred that has had a Material Adverse Effect; (b) neither the Company nor any of its Subsidiaries has incurred any liability or obligation, direct or contingent, or entered into any transaction or agreement that, individually or in the aggregate, is material with respect to the Company and its Subsidiaries taken as a whole, and neither the Company nor any of its Subsidiaries has sustained any loss or interference with its business or operations from fire, explosion, flood, earthquake or other natural disaster or calamity, whether or not covered by insurance, or from any labor dispute or disturbance or court or governmental action, order or decree which would reasonably be expected, individually or in the aggregate, to result in a Material Adverse Effect; and (c) there has been no dividend or distribution of any kind declared, paid or made by the Company on any class of its capital stock.

2.9 <u>No Registration Required</u>. Assuming the accuracy of the representations and warranties of the Holder contained in <u>Section 3</u>, the issuance of the Issued Shares pursuant to this Agreement is exempt from the registration requirements of the Securities Act. Neither the Company nor any Person acting on its behalf, directly or indirectly, has made or will make any offers or sales of any security, or has solicited or will solicit offers to buy, or otherwise has negotiated or will negotiate in respect of, any security, under circumstances that would require registration of the Issued Shares under the Securities Act, except as provided herein.

2.10 <u>Litigation</u>. There is no Legal Proceeding before or brought by any court or governmental agency or body, domestic or foreign, now pending, or, to the Knowledge of the Company or its Subsidiaries, threatened in writing or otherwise overtly threatened, against or affecting the Company or any of its Subsidiaries that would reasonably be expected, individually or in the aggregate, to result in a Material Adverse Effect or to materially and adversely affect the consummation of the Transactions or the performance by the Company of its obligations under this Agreement.

2.11 <u>Auditors</u>. Ernst & Young LLP are independent public accountants of the Company as required by the Securities Act and the Exchange Act and the rules and regulations thereunder/pursuant to the rules applicable to such accountants under the American Institute of Certified Public Accountants.

2.12 <u>Statistical and Market-Related Data</u>. Any statistical, demographic, market-related and similar data included in the Company SEC Documents are based on or derived from sources that the Company believes to be reliable and accurate and accurately reflect the materials upon which such data is based or from which it was derived in all material respects.

2.13 <u>No Related Party Transactions</u>. No relationship, direct or indirect, that would be required to be described in a registration statement of the Company pursuant to Item 404 of Regulation S-K, exists between the Company or any of its Subsidiaries, on the one hand, and the directors, officers, shareholders, customers or suppliers of the Company or any of its Subsidiaries, on the other hand, except for relationships or transactions solely between the Company and one of its Subsidiaries and or between or among one or more Subsidiaries of the Company or as set forth in the Company SEC Documents.

2.14 <u>Insurance</u>. The Company and its Subsidiaries are insured by insurers of recognized financial responsibility against such losses and risks and in such amounts as are prudent and customary in the businesses in which they are engaged; all policies of insurance and any fidelity or surety bonds insuring the Company or any of its Subsidiaries or their respective businesses, assets, employees, officers and directors are in full force and effect; the Company and its Subsidiaries under any such policy or instrument as to which any insurance company is denying liability or defending under a reservation of rights clause. Since the earlier of (i) January 1, 2020, and (ii) the effective date of the Company's most recently issued insurance policy, none of the Company nor any of its Subsidiaries has been refused any insurance coverage sought or applied for; and none of the Company, nor any of its Subsidiaries has been refused any insurance coverage as and when such coverage expires or to obtain similar coverage from similar insurers at a cost that would not, individually or in the aggregate, result in a Material Adverse Effect.

2.15 <u>Accounting and Disclosure Controls</u>. The Company and its Subsidiaries maintain a system of internal accounting controls sufficient to provide reasonable assurance that (a) transactions are executed in accordance with management's general or specific authorizations; (b) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain asset accountability; (c) access to assets is permitted only in accordance with management's general or specific authorization; and (d) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to

any differences. Since January 1, 2020, there has been (i) no material weakness or significant deficiency (as defined in Rule 1-02 of Regulation S-X of the SEC) in the Company's internal control over financial reporting (whether or not remediated), and (ii) no fraud, whether or not material, involving management or other employees who have a role in the Company's internal control over financial reporting has occurred and, since January 1, 2020, there has been no change in the Company's internal control over financial reporting that has materially affected, or is reasonably likely to materially affect, the Company's internal control over financial reporting. The Company and its Subsidiaries have established, maintained and periodically evaluate the effectiveness of "disclosure controls and procedures" (as defined in Rules 13a-15 and 15d-15 under the Exchange Act). Such disclosure controls and procedures are designed to ensure that information required to be disclosed by the Company in the reports that it will be required to file or submit under the Exchange Act is recorded, processed, summarized and reported, within the time periods specified in the SEC's rules and forms, and is accumulated and communicated to the Company's management, including its principal executive officer or officers and principal financial officer or officers, as appropriate, to allow timely decisions regarding disclosure. The interactive data in eXtensible Business Reporting Language included or incorporated by reference in the Company SEC Documents fairly presents the information called for in all material respects and is prepared in accordance with the SEC's rules and guidelines applicable thereto. The Company's independent public accountants and the audit committee of the Board of Directors of the Company have been advised of all material weaknesses, if any, and significant deficiencies (as defined in Rule 1-02 of Regulation S-X of the SEC), if any, in the Company's internal control over financial reporting and of all fraud, if any, whether or not material, involving management or other employees who have a role in the Company's internal controls, in each case that occurred or existed, or was first detected, at any time during the three most recent fiscal years covered by the Company's most recent audited financial statements included in the Company SEC Documents.

2.16 <u>Sarbanes-Oxley Compliance</u>. There is and has been no failure on the part of the Company or any of the Company's directors or officers, in their capacities as such, to comply in any material respect with any provision of the Sarbanes-Oxley Act with which any of them is required to comply, including Section 402 related to loans.

2.17 Tax Returns. The Company and its Subsidiaries have filed all U.S. federal and all other material foreign, state, local and franchise tax returns that are required to be filed or have obtained extensions thereof, and have paid all taxes (including, without limitation, any estimated taxes) required to be paid and any other assessment, fine or penalty, to the extent that any of the foregoing is due and payable, except (a) for any such tax, assessment, fine or penalty that is currently being contested in good faith by appropriate actions and for which adequate reserves have been provided in accordance with GAAP, and (b) for such taxes, assessments, fines or penalties the nonpayment of which would not, individually or in the aggregate, result in a Material Adverse Effect.

2.18 Environmental Laws. Except as would not, individually or in the aggregate, result in a Material Adverse Effect, (a) neither the Company nor any of its Subsidiaries is in violation of any federal, state, local or foreign statute, law, rule, regulation, ordinance, code, policy or rule of common law or any judicial or administrative interpretation thereof, including any judicial or administrative order, consent, decree or judgment, relating to pollution or protection of human health and safety, the environment (including, without limitation, indoor or outdoor air, surface water, groundwater, land surface or subsurface strata) or wildlife, or to the release or threatened release of chemicals, pollutants, contaminants, wastes, toxic substances, hazardous substances, flammable, corrosive or radioactive materials, petroleum or petroleum products (collectively, "Hazardous Materials") or to the manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Materials (collectively, "Environmental Laws"), (b) the Company and its Subsidiaries have all permits, licenses,

authorizations and approvals required under any applicable Environmental Laws and are each in compliance with their requirements, (c) neither the Company nor any of its Subsidiaries is subject to any order, judgment or consent decree, and there are no pending or threatened administrative, regulatory or judicial actions, suits, demands, demand letters, claims, Liens, notices of noncompliance or violation, investigation or proceedings against the Company or any of its Subsidiaries, in each case relating to any Environmental Law, and (d) there are no events or circumstances that would reasonably be expected to form the basis of an order to install or retrofit equipment, pay penalties or for clean-up or remediation of containment, or a Legal Proceeding by any private party or Governmental Agency, against or affecting the Company or any of its Subsidiaries relating to Hazardous Materials or any Environmental Laws.

2.19 <u>Title to Property</u>. The Company and each of its Subsidiaries have good and marketable title to all real property owned in fee simple by any of them (if any) and have good and valid title to all other items of owned real and personal property that are material to the respective businesses of the Company and each of its Subsidiaries, in each case free and clear of all Liens, except for (a) Permitted Liens, (b) as are described in Schedule 2.19 or (c) where the failure to have such title would not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect. All material real property, buildings and other improvements, and all equipment and other property, held under lease or sublease by the Company or any of its Subsidiaries is held by them under valid, subsisting and enforceable leases or subleases, as the case may be (subject to applicable bankruptcy, insolvency, reorganization, or other Laws of general application relating to or affecting the enforcement of creditors' rights generally and to general principles of equity), with, solely in the case of leases or subleases relating to real property and buildings or other improvements, such exceptions as are not material and do not interfere with the use made or proposed to be made of such real property and buildings or other improvements by the Company and its Subsidiaries, and all such leases and subleases are in full force and effect in all material respects. Neither the Company nor any of its Subsidiaries has any notice of any claim of any sort that has been asserted by anyone adverse to the rights of the Company or any of its Subsidiaries under any of the leased or subleased premises or the continued use of the leased or subleased equipment or other property, except for such claims which, if successfully asserted against the Company, or any of its Subsidiaries, would not, individually or in the aggregate, result in a Material Adverse Effect.

2.20 <u>Absence of Labor Disputes</u>. No labor dispute with the employees of the Company or any of its Subsidiaries exists or, to the Knowledge of the Company or any of its Subsidiaries, is imminent, and neither the Company nor any of its Subsidiaries is aware of any existing or imminent labor disturbance by the employees of any of the principal suppliers, manufacturers, customers or contractors of the Company or any of its Subsidiaries, in each case, which has or would reasonably be expected to have a Material Adverse Effect.

2.21 <u>ERISA Compliance</u>. None of the following events has occurred or exists: (a) a failure to fulfill the obligations, if any, under the minimum funding standards of Section 302 of ERISA with respect to a Plan determined without regard to any waiver of such obligations or extension of any amortization period; (b) an audit or investigation by the Internal Revenue Service, the U.S. Department of Labor, the Pension Benefit Guaranty Corporation or any other federal, state or foreign governmental or regulatory agency with respect to the employment or compensation of employees by the Company or any of its Subsidiaries that would reasonably be expected, individually or in the aggregate, to result in a Material Adverse Effect; or (c) any breach of employees by the Company or any of its Subsidiaries that would reasonably be expected, individually be expected, individually or in the aggregate, to result in a Material Adverse Effect. None of the following events has occurred in the past two years or is

reasonably likely to occur: (i) a material increase in the aggregate amount of contributions required to be made to all Plans in the current fiscal year of the Company and its Subsidiaries compared to the amount of such contributions made in the Company's most recently completed fiscal year; (ii) a material increase in the "accumulated post-retirement benefit obligations" (within the meaning of Statement of Financial Accounting Standards 106) of the Company and its Subsidiaries compared to the amount of such obligations in the Company's most recently completed fiscal year; (iii) any event or condition giving rise to a liability under Title IV of ERISA that would reasonably be expected, individually or in the aggregate, to result in a Material Adverse Effect; or (iv) the filing of a claim by one or more employees or former employees of the Company or any of its Subsidiaries related to its or their employment that would reasonably be expected, individually or in the aggregate, to result in a Material Adverse Effect. For purposes of this paragraph and the definition of ERISA, the term "*Plan*" means a plan (within the meaning of Section 3(3) of ERISA) with respect to which the Company or any of its Subsidiaries may have any liability.

2.22 <u>Absence of Manipulation</u>. Neither the Company nor any of its Subsidiaries has taken or will take, directly or indirectly, any action designed to or that would constitute or that would reasonably be expected to cause or result in the stabilization or manipulation of the price of any security or to facilitate the sale or resale of its securities, except for the making of public disclosures in compliance with applicable securities laws in the ordinary course of business consistent with past practice.

2.23 <u>Anti-Corruption</u>. Neither the Company nor any of its Subsidiaries, nor to the Knowledge of the Company, any director, officer, agent, employee, Affiliate or other Person acting on behalf of the Company or any of its Subsidiaries is aware of or has taken any action, directly or indirectly, that has resulted in a material violation by any such Person of any Anti-Corruption Laws, including, without limitation, any offer, payment, promise to pay or authorization of the payment of any money or other property, gift, promise to give or authorization of the giving of anything of value to any "foreign official" (as such term is defined in the FCPA) or any foreign political party or official thereof or any candidate for foreign political office, in contravention of the FCPA and the Company and its Subsidiaries. To the Knowledge of the Company, it and its Subsidiaries have conducted their respective businesses in compliance with Anti-Corruption Laws and have instituted and maintained policies and procedures designed to ensure, and which are reasonably expected to ensure, continued compliance therewith in all material respects.

2.24 <u>Money Laundering Laws</u>. The operations of the Company and its Subsidiaries are and have been conducted at all times in material compliance with applicable financial recordkeeping and reporting requirements of the Currency and Foreign Transactions Reporting Act of 1970, as amended, the money laundering statutes of all applicable jurisdictions, the rules and regulations thereunder and any related or similar rules, regulations or guidelines, issued, administered or enforced by any Governmental Agency (collectively, "*Money Laundering Laws*") and no material Legal Proceeding by or before any court or Governmental Agency or any arbitrator involving the Company or any of its Subsidiaries with respect to the Money Laundering Laws is pending or, to the Knowledge of the Company or any of its Subsidiaries, threatened.

2.25 No Conflicts with Sanctions Laws. Neither the Company nor any of its Subsidiaries nor, to the Knowledge of the Company or any of its Subsidiaries, any director, officer, agent, employee, Affiliate or other Person acting on behalf of the Company or any of its Subsidiaries is currently or has been the target of any sanctions administered or enforced by the U.S. Government (including, without limitation, OFAC), the United Nations Security Council ("UNSC"), the European Union, Her Majesty's Treasury ("HMT"), or other relevant sanctions authority (collectively, "Sanctions"), nor is the Company or any of its Subsidiaries a Sanctioned Person. The Company and its Subsidiaries have conducted their businesses in material compliance with all applicable Sanctions and have instituted and maintain policies and procedures designed to ensure, and which are reasonably expected to ensure, continued compliance therewith in all material respects.

2.26 <u>Certain Fees</u>. Except for the fees and expenses payable to the Jefferies Group and Houlihan Lokey Capital, Inc. ("*Houlihan Lokey*" and, together with the Jefferies Group, the "*Financial Advisors*") and certain expenses of Perella Weinberg Partners, which will be paid by the Company, no broker, finder or investment or commercial banker, or other Person engaged by or acting on behalf of the Company or its Subsidiaries in a similar capacity, is or will be entitled to any fees, commissions or expenses in connection with the issuance of the Issued Shares or the consummation of the Transactions.

2.27 Intellectual Property; Licenses, Etc. The Company and its Subsidiaries own license, possess or otherwise have the right to use, all of the trademarks, service marks, trade names, domain names, copyrights, patents, patent rights, technology, software, know-how database rights, design rights and other intellectual property rights (collectively, the "*IP Rights*") that are reasonably necessary for the operation of their respective businesses as currently conducted, and, to the Knowledge of the Company, without, to the Knowledge of the Company, violation of the rights of any Person, except to the extent such failure to own, license, possess or otherwise have the right to use, or such violations that, have not had, or would not reasonably be expected to have, a Material Adverse Effect. To the Knowledge of the Company, no such IP Rights owned by the Company or any Subsidiary infringe upon any rights held by any Person except for such infringements, individually or in the aggregate, which could not reasonably be expected to have a Material Adverse Effect. No Legal Proceeding regarding any such IP Rights owned by the Company or any Subsidiary is pending or, to the Knowledge of the Company, threatened against the Company or any Subsidiary which would reasonably be expected to have a Material Adverse Effect.

2.28 <u>Possession of Licenses and Permits</u>. The Company and its Subsidiaries possess such permits, licenses, approvals, consents and other authorizations (collectively, "*Governmental Licenses*") issued by the appropriate federal, state, local or foreign regulatory agencies or bodies necessary to conduct the business now operated by them, except where the failure of which would not reasonably be expected to have a Material Adverse Effect. The Company and its Subsidiaries are in compliance with the terms and conditions of all such Governmental Licenses, except where the failure so to comply would not reasonably be expected to have a Material Adverse Effect. All of the Governmental Licenses are valid and in full force and effect, except where the failure of which would not reasonably be expected to have a Material Adverse Effect. Neither the Company nor any of its Subsidiaries has received any notice of proceedings relating to the revocation or modification of any such Governmental Licenses, except where such proceeding would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

2.29 <u>Investment Company</u>. Neither the Company nor any of its Subsidiaries is, and upon the issuance of the Issued Shares as herein contemplated will be, an "investment company" or an entity "controlled" by an "investment company" as such terms are defined in the Investment Company Act of 1940, as amended.

2.30 <u>No Restrictions on Dividends</u>. No Subsidiary of the Company is party to or otherwise bound by any instrument or agreement that limits or prohibits or could limit or prohibit, directly or indirectly, any Subsidiary of the Company from paying any dividends or making any other distributions on its capital stock, limited or general partnership interests, limited liability company interests, or other equity interests, as the cast may be, or from repaying any loans or advances from, or (except for instruments or agreements that by their express terms prohibit the transfer or assignment thereof or of any rights thereunder) transferring any of its properties or assets to, the Company or any other Subsidiary.

2.31 <u>Takeover Statutes</u>. The approval of the Board of Directors of the Company of this Agreement and the Transactions on or prior to the date hereof represent all the actions necessary to render inapplicable to this Agreement and the Transactions the provisions of Section 203 of the DGCL to the extent, if any, such section would otherwise be applicable to this Agreement and the Transactions, and no other state takeover statute applies to this Agreement and the Transactions.

2.32 <u>Fairness Opinion</u>. The Special Committee has received an opinion from Houlihan Lokey to the effect that, as of the date of such opinion, and based upon and subject to the assumptions, procedures, matters, qualifications and limitations set forth therein, the consideration to be paid to the Holder by the Company in the Exchange, taking into account the Dividend, is fair from a financial point of view to the holders of the Common Stock other than the Holder and its Affiliates.

2.33 <u>Proxy Statement</u>. The Company Proxy Statement, at the time the Company Proxy Statement becomes effective or at the date of mailing and at the date of the Company Stockholder Meeting, will not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading. The Company Proxy Statement will comply in all material respects with the provisions of the Securities Act and the Exchange Act. Notwithstanding the foregoing, the Company makes no representation or warranty with respect to any information provided in writing by or on behalf of the Holder expressly for inclusion in the Company Proxy Statement.

2.34 No Other Representations or Warranties. THE COMPANY ACKNOWLEDGES THAT, EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN <u>SECTION 3</u>, NEITHER THE HOLDER NOR ANY OTHER PERSON HAS MADE, AND THE COMPANY HAS NOT RELIED ON, ANY EXPRESS OR IMPLIED REPRESENTATION OR WARRANTY BY OR ON BEHALF OF THE HOLDER OR ANY THIRD PARTY OR ANY INFORMATION PROVIDED BY THE HOLDER OR ANY OTHER PERSON. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY MADE IN THIS <u>SECTION 2</u>, NO PERSON MAKES ANY REPRESENTATION OR WARRANTY WITH RESPECT TO THE COMPANY OR ANY OF ITS AFFILIATES OR THEIR RESPECTIVE BUSINESSES, OPERATIONS, ASSETS, LIABILITIES, CONDITION (FINANCIAL OR OTHERWISE) OR PROSPECTS, NOTWITHSTANDING THE DELIVERY OR DISCLOSURE TO THE HOLDER OR ANY OF ITS AFFILIATES OR REPRESENTATIVES OF ANY DOCUMENTATION, FORECASTS, PROJECTIONS OR OTHER INFORMATION WITH RESPECT TO ANY ONE OR MORE OF THE FOREGOING. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY MADE IN THIS <u>SECTION 2</u>, ALL OTHER REPRESENTATIONS AND WARRANTIES, WHETHER EXPRESS OR IMPLIED, ARE EXPRESSLY DISCLAIMED BY THE COMPANY. NONE OF THE FOREGOING SHALL LIMIT ANY CLAIM FOR FRAUD.

3. <u>REPRESENTATIONS AND WARRANTIES OF THE HOLDER</u>. The Holder represents and warrants to the Company as follows:

3.1 <u>Existence</u>. The Holder has been duly organized and is validly existing and in good standing under the laws of its state of formation, except where such failure would not reasonably be expected, individually or in the aggregate, to result in a Holder Material Adverse Effect.

3.2 <u>**Ownership**</u>. The Holder is the sole legal and beneficial owner of the Existing Preferred, free and clear of any and all Liens. The Holder has sole right, title and interest in the Existing Preferred and there have been no assignments or other transfers by the Holder of any interest whatsoever in the Existing Preferred. Except for the Existing Preferred and 4,069,324 shares of Common Stock, neither the Holder nor any of its Affiliates own any equity securities of the Company.

3.3 <u>Authorization, Enforceability</u>. All corporate action on the part of the Holder, its officers, directors, managers and members necessary for the authorization, execution, delivery, and performance of all obligations of the Holder under the Transaction Documents has been taken, and this Agreement constitutes, and the other Transaction Documents to which it is a party when executed and delivered, shall constitute, a valid and legally binding obligation of the Holder, enforceable in accordance with their respective terms, except as may be limited by (a) applicable bankruptcy, insolvency, reorganization, or other Laws of general application relating to or affecting the enforcement of creditors' rights generally and (b) the effect of rules of law governing the availability of equitable remedies.</u>

3.4 <u>No Breach</u>. The execution, delivery and performance of the Transaction Documents by the Holder and the consummation by the Holder of the transactions contemplated hereby and thereby does not and will not whether with or without the giving of notice or passage of time or both (a) conflict with or result in any violation of the provisions of the organizational documents of the Holder, or (b) violate any Law applicable to the Holder or the Property or assets of the Holder, except for the expiration of the waiting period under the HSR Act or, in the case of clauses (a) and (b), for such conflicts, breaches, violations or defaults as would not reasonably be expected, individually or in the aggregate, to result in a Holder Material Adverse Effect.

3.5 <u>Certain Fees</u>. No broker, finder or investment or commercial banker, or other Person engaged by or acting on behalf of the Holder or any of its Affiliates in a similar capacity is or will be entitled to any fees or commissions in connection with the acquisitions of the Issued Shares or the consummation of the Transactions.

3.6 Unregistered Securities.

3.6.1 <u>Accredited Investor Status; Sophisticated Purchaser</u>. The Holder is an "accredited investor" within the meaning of Rule 501 under the Securities Act and is able to bear the risk of its investment in the Issued Shares. The Holder has such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risks of the acquisition of the Issued Shares.

3.6.2 **Information**. The Holder and its Representatives have been afforded the opportunity to ask questions of, and obtain information from, the Company regarding the terms and conditions of the Transactions and the business, properties, prospects and financial condition of the Company. The Holder understands that its acquisition of the Issued Shares involves a high degree of risk. The Holder has received all information, and has sought such accounting, legal and tax advice, as it has considered necessary or appropriate to make an informed investment decision with respect to its acquisition of the Securities.

3.6.3 **Legends**. The Holder understands that, until such time as the Issued Shares have been registered pursuant to the provisions of the Securities Act, or the Issued Shares are eligible for resale pursuant to Rule 144 promulgated under the Securities Act without any restriction as to the number of securities as of a particular date that can then be immediately sold or otherwise, or as provided in this <u>Section 3.6.3</u>, the Issued Shares will bear the following restrictive legend (in addition to any legend required under applicable state securities laws):

(a) THE SECURITIES REPRESENTED BY THIS CERTIFICATE HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OR UNDER ANY STATE SECURITIES OR BLUE SKY LAWS. THE SECURITIES ARE SUBJECT TO RESTRICTIONS ON TRANSFERABILITY AND RESALE AND MAY NOT BE OFFERED, SOLD, PLEDGED, HYPOTHECATED OR OTHERWISE TRANSFERRED EXCEPT AS PERMITTED UNDER THE SECURITIES ACT AND APPLICABLE STATE SECURITIES OR BLUE SKY LAWS, PURSUANT TO REGISTRATION OR EXEMPTION THEREFROM. THE COMPANY MAY REQUIRE AN OPINION OF COUNSEL IN FORM AND SUBSTANCE REASONABLY SATISFACTORY TO THE COMPANY TO THE EFFECT THAT ANY PROPOSED TRANSFER IS IN COMPLIANCE WITH THE SECURITIES ACT AND ANY APPLICABLE STATE SECURITIES OR BLUE SKY LAWS.

Holder.

(b) Any other legends set forth in or required by the Charter and any other legends required by state securities laws applicable to the

The legend set forth in this <u>Section 3.6.3</u> shall be removed by the Company from any certificate evidencing the legended Issued Shares, upon delivery to the Company of an opinion by counsel, reasonably satisfactory to the Company, that a registration statement under the Securities Act was at that time in effect with respect to the legended security and such Shares have been sold thereunder or that such security can be freely transferred in a public sale without such a registration statement being in effect and that such transfer shall not jeopardize the exemption or exemptions from registration pursuant to which the Company issued the applicable Issued Shares.

(c) **Purchase Representation**. The Holder is purchasing the Issued Shares for its own account and not with a view to distribution in violation of any securities laws. The Holder has been advised and understands that none of the Issued Shares have been registered under the Securities Act or under the "blue sky" laws of any jurisdiction and may be resold only if registered pursuant to the provisions of the Securities Act (or if eligible, pursuant to the provisions of Rule 144 promulgated under the Securities Act or pursuant to another available exemption from the registration requirements of the Securities Act). The Holder has been advised and understands that the Company, in issuing the Issued Shares, is relying upon, among other things, the representations and warranties of the Holder contained in this <u>Section 3</u> in concluding that such issuance is a "private offering" and is exempt from the registration provisions of the Securities Act.

(d) **Rule 144**. The Holder has been advised of and is aware of the provisions of Rule 144 promulgated under the Securities Act.

(e) <u>Reliance by the Company</u>. The Holder understands that the Issued Shares are being offered and sold in reliance on a transactional exemption from the registration requirements of federal and state securities Laws and that the Company is relying upon the truth and accuracy of the representations, warranties, agreements, acknowledgments and understandings of the Holder set forth herein in order to determine the applicability of such exemptions and the suitability of the Holder to acquire the Issued Shares.

3.7 Tax and Legal Matters. The Holder hereby acknowledges that the Company has not provided the Holder with any tax advice with respect to the Exchange, the issuance of the Issued Shares to the Holder pursuant to the Exchange (or the tax consequences thereto) or the other transactions

described herein, and no such advice has been implied. The Holder has sought such tax and legal advice as it has deemed necessary in connection with the execution of this Agreement.

3.8 <u>No Conflicts with Sanctions Laws</u>. The Holder is not currently and has not, and, to the knowledge of the Holder, no director, officer, agent, employee, Affiliate or other Person acting on behalf of the Holder is currently or has been, the target of any Sanctions, nor is the Holder a Sanctioned Person. The execution, delivery and performance of the Transaction Documents by the Holder and the consummation by the Holder of the transactions contemplated hereby and thereby does not and will not violate any applicable Sanctions.

3.9 <u>HSR; Antitrust</u>. To the knowledge of the Holder, as of the date of this Agreement, there are no substantive overlaps between the offerings of the Company and the Holder of such a scope that would be reasonably likely to result in the issuance of a Second Request by the U.S. Department of Justice's Antitrust Division or the Federal Trade Commission.

3.10 No Other Representations or Warranties. THE HOLDER ACKNOWLEDGES THAT, EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN <u>SECTION 2</u>, NEITHER THE COMPANY NOR ANY OTHER PERSON HAS MADE, AND THE HOLDER HAS NOT RELIED ON, ANY EXPRESS OR IMPLIED REPRESENTATION OR WARRANTY BY OR ON BEHALF OF THE COMPANY OR ANY THIRD PARTY OR ANY INFORMATION PROVIDED BY THE COMPANY OR ANY OTHER PERSON. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY MADE IN THIS <u>SECTION 3</u>, NO PERSON MAKES ANY REPRESENTATION OR WARRANTY WITH RESPECT TO THE HOLDER OR ANY OF ITS AFFILIATES OR THEIR RESPECTIVE BUSINESSES, OPERATIONS, ASSETS, LIABILITIES, CONDITION (FINANCIAL OR OTHERWISE) OR PROSPECTS, NOTWITHSTANDING THE DELIVERY OR DISCLOSURE TO THE COMPANY OR ANY OF ITS AFFILIATES OR REPRESENTATIVES OF ANY DOCUMENTATION, FORECASTS, PROJECTIONS OR OTHER INFORMATION WITH RESPECT TO ANY ONE OR MORE OF THE FOREGOING. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES EXPRESSLY MADE IN THIS <u>SECTION 3</u>, ALL OTHER REPRESENTATIONS AND WARRANTIES, WHETHER EXPRESS OR IMPLIED, ARE EXPRESSLY DISCLAIMED BY THE HOLDER. NONE OF THE FOREGOING SHALL LIMIT ANY CLAIM FOR FRAUD.

4. COVENANTS.

4.1 <u>Interim Operating Covenants</u>. Except (i) as otherwise required or expressly contemplated by this Agreement, (ii) as required by Law or (iii) with the prior written consent or waiver of the Holder (such consent not to be unreasonably withheld, conditioned or delayed), the Company hereby covenants and agrees that, between the date hereof and the Closing:

4.1.1 the Company shall not, and shall not permit any Subsidiary to, incur any Indebtedness in an aggregate amount in excess of \$50,000,000 outstanding at any time, except for (i) Indebtedness solely between or among the Company and any of its wholly owned Subsidiaries, (ii) letters of credit issued in the ordinary course of business consistent with past practices and (iii) trade credit or trade payables in the ordinary course of business consistent with past practices;

4.1.2 except as set forth in the applicable Capital Budget, the Company shall not, and shall not permit any Subsidiary to, make or authorize capital expenditures in any calendar year that, individually or in the aggregate, exceeds 50% of such amount set forth in the applicable Capital Budget;

4.1.3 the Company shall not, and shall not permit any Subsidiary to, sell, lease, license or otherwise dispose of or encumber any of its properties or assets that are material, individually or in the aggregate, to the business of the Company and its Subsidiaries or enter into any Contract with respect to the foregoing;

4.1.4 the Company shall not, and shall not permit any Subsidiary to, acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial portion of the assets of, or by any other manner, any business of any Person or division thereof, or otherwise acquire or agree to acquire any assets, in each case, that are material, individually or in the aggregate, to the business of the Company and its Subsidiaries, or enter into any Contract with respect to a joint venture, strategic alliance or partnership;

4.1.5 the Company shall not, and shall not permit any Subsidiary to, amend its or their certificate of incorporation or bylaws or comparable organizational or governing documents, change the size of its board of directors or similar governing body or create an executive committee or similar committee to whom general authority of the board of directors is delegated, <u>provided</u>, <u>however</u>, that nothing contained in this Section 4.1.5 shall prohibit the adoption of a Delaware forum selection bylaw;

4.1.6 the Company shall promptly notify the Holder upon the Company becoming aware of any new Legal Proceeding filed, or threatened to be filed, against the Company or any of its Subsidiaries which could reasonably be expected to have a Material Adverse Effect; or

4.1.7 the Company shall not declare or pay any dividends or make any distributions in respect of its capital stock, except as expressly set forth in this Agreement.

4.2 Further Actions.

4.2.1 Upon the terms and subject to the conditions set forth in this Agreement, the Holder and the Company agree to use reasonable best efforts to take, or cause to be taken, all actions necessary, proper or advisable to consummate, as promptly as reasonably practicable, the Transactions, including using reasonable best efforts to (i) obtain all authorizations, consents, Orders, approvals, licenses, permits and waivers of all Governmental Authorities or third parties that may be or become necessary for its execution and delivery of, and the performance of its obligations pursuant to, this Agreement and the consummation of the Transactions and (ii) provide such other information to any Governmental Authority as such Governmental Authority may lawfully request in connection herewith. Each party shall make its respective filing, if necessary, pursuant to the HSR Act as promptly as reasonably practicable, but in any event no later than ten (10) Business Days, following the date of this Agreement. In addition, each party agrees to make, or cause to be made, as promptly as reasonably practicable and use its reasonable best efforts to make as promptly as practicable after the date of this Agreement (or such other date as may be required by the applicable Antitrust Law) any other filings and notifications pursuant to any other Antitrust Law with respect to the Transactions. The parties shall supply as promptly as reasonably practicable thereafter to the appropriate Governmental Authorities any additional information and documentary material that may be requested related to the Transactions, consents, Orders or approvals, except as otherwise set forth in <u>Section 7.15</u>. In furtherance of the foregoing, the Holder shall, and shall use commercially reasonable efforts to cause each of its Affiliates to, consider in good faith commercially reasonable steps to avoid or eliminate each and every impediment under any Antitrust Law that has been asserted by any Governmental Authority so as to enable the parties hereto to consumm

practicable, and in any event prior to the Outside Date, including considering the sale, divestiture, license or other disposition of such of its assets, properties or businesses, to avoid the entry of, and the commencement of litigation seeking the entry of, or to effect the dissolution of, any injunction, temporary restraining order or other Order in any suit or proceeding that would otherwise have the effect of materially delaying or preventing the consummation of the Transactions. Notwithstanding the foregoing or any other provision of this Agreement to the contrary, in no event shall the Holder or any of its Affiliates be obligated to, sell, divest, license, dispose of, take any other action with respect to, accept any operational modification or restriction with respect to or otherwise encumber the any of the assets, Properties, businesses or investments of the Holder or any of its Affiliates.

4.2.2 Each party shall notify the other party of, and give such other party the opportunity to participate in the defense or settlement of, any Legal Proceeding brought by current or former stockholders of the Company against the Company and/or its directors or officers relating to the Transactions ("*Stockholder Litigation*") that would prevent or materially impede, interfere with, hinder or delay the consummation of the Transactions or which would prevent the consummation of the Exchange prior to the Outside Date. Without limiting the foregoing, each party shall (i) keep the other party reasonably informed on a current basis of the status and material developments in, any such Stockholder Litigation, (ii) give such other party the opportunity to participate in, review and comment (which comments shall be considered in good faith) on all material filings or responses to be made in the defense or settlement of any Stockholder Litigation, and (iii) not compromise, settle, come to an arrangement regarding or agree to compromise, settle or come to an arrangement regarding any Stockholder Litigation, or consent to the same without the prior written consent of such other party.

4.2.3 Subject to applicable Laws, the Company and the Holder each shall use its reasonable best efforts to (i) cooperate in all respects with each other in connection with any filing or submission with a Governmental Authority in connection with the Transactions and in connection with any investigation or other inquiry by or before a Governmental Authority relating thereto and (ii) keep each other reasonably apprised of the content and status of any material communications with, and communications from, any Governmental Authority with respect to the Transactions, including promptly notifying the other parties hereto of any communication it or any of its Affiliates receives from any Governmental Authority relating to any review or investigation of the Transactions under the HSR Act, and shall permit the other parties to review in advance (and to consider in good faith any comments made by the other party in relation to) any proposed material communication by such party to any Governmental Authority relating to such matters. None of the parties to this Agreement shall agree to participate in any meeting, telephone call or discussion with any Governmental Authority in respect of any submissions, filings, investigation (including any settlement of the investigation), litigation or other inquiry relating to the matters that are the subject of this Agreement unless it consults with the other parties a reasonable amount of time in advance and, unless prohibited by such Governmental Authority, gives the other parties the opportunity to attend and participate at such meeting, telephone call or discussion. The parties to this Agreement will coordinate and cooperate fully with each other in exchanging such information and providing such assistance as the other party may reasonably request in connection with the foregoing and in seeking early termination of any applicable waiting periods, including under the HSR Act. The parties to this Agreement shall provide each other with copies of all correspondence, filings or communications between them or any of their Representatives, on the one hand, and any Governmental Authority or members of its staff, on the other hand, with respect to this Agreement and the Transactions; provided, however, that each party may, as it deems advisable and necessary, reasonably designate any competitively sensitive materials provided to the other party as "Outside Counsel Only Material" and may redact the materials as necessary to (i) remove references concerning the valuation of the Company, (ii) comply with contractual arrangements and (iii) address reasonable attorney-client or other privilege or confidentiality concerns.

4.2.4 Notwithstanding anything to the contrary in this Agreement, no party shall be required, and the Company shall not permit its Subsidiaries, to take or commit to take any action with respect to its assets, Properties, business or operations in connection with obtaining the expiration or termination of the applicable waiting periods under, or any approvals under, the HSR Act or any other authorization, unless the effectiveness of such agreement or action is conditioned upon the occurrence of the Closing.

4.2.5 None of the Company, its Subsidiaries or the Holder shall, and each shall cause their respective controlled Affiliates not to, and shall use reasonable best efforts to cause their respective non-controlled Affiliates not to, enter into any Contract or transaction (including any merger or acquisition) that would reasonably be expected to make it materially more difficult, or to materially increase the time required, to: (i) consummate the Transactions and the Closing, (ii) obtain the expiration or termination of the waiting period under the HSR Act, or the authorizations, consents, Orders and approvals required under any other Antitrust Law applicable to the Transactions, (iii) avoid the entry of, avoid the commencement of litigation seeking the entry of, or effect the dissolution of, any injunction, temporary restraining order or other Order that would materially delay or prevent the consummation of the Transactions or (iv) obtain all authorizations, consents, Orders and approvals of Governmental Authorities necessary for the consummation of the Transactions.

4.3 Stockholder Meeting; Proxy Material.

4.3.1 The Company shall cause a meeting of its stockholders (the "Company Stockholder Meeting") to be duly called and held as soon as reasonably practicable following the date of this Agreement (but in any event, no later than sixty (60) days after the date that the Company Proxy Statement is mailed to the Company stockholders) for the purpose of obtaining the Company Stockholder Approval. Subject to any Adverse Recommendation Change, the Board of Directors of the Company and Special Committee of the Company shall each recommend approval of the Company Stockholder Approval to the Company's stockholders, and the Company Proxy Statement shall include a statement to the effect that the Board of Directors of the Company and the Special Committee of the Company have recommended that the Company's stockholders vote in factor of the adoption of this Agreement at the Company Stockholder Meeting. In connection with such meeting, the Company (i) shall promptly prepare and file with the Securities and Exchange Commission (the "SEC"), and in any event within fifteen (15) days after the date of this Agreement, and shall use its reasonable efforts to have cleared by the SEC and shall thereafter mail to its stockholders as promptly as practicable, the Company Proxy Statement and all other proxy materials for such meeting, (ii) shall use its commercially reasonable efforts to obtain the necessary approvals by its stockholders of the Company Stockholder Approval, and (iii) shall otherwise comply with all Laws applicable to such meeting. The Company shall provide the Holder with a copy of the preliminary proxy statement and all modifications and amendments thereto, and shall consult with and provide the Holder with a reasonable opportunity to review and comment thereon, prior to filing or delivery to the SEC. The Company shall consider in good faith any and all comments received from the Holder with respect to the preliminary proxy statement and all modification and amendments thereto for inclusion in such preliminary proxy statement, modification or amendment. The Company shall notify the Holder promptly of the receipt of any comments from the SEC or its staff and of any request by the SEC or its staff for amendments or supplements to the Company Proxy Statement or for additional information and shall supply the Holder with copies of all material correspondence between the Company or any of its Representatives, on the one hand, and the SEC or its staff, on the other hand, with respect to the Company Proxy Statement or the Transactions, and shall consult with the Holder prior to responding to any such comments or requests. If at any time prior to the Company Stockholder Meeting there shall occur any event that should be set forth in an amendment or supplement to the Company Proxy Statement, the Company shall promptly prepare and mail to its stockholders such an amendment or supplement. The Company shall not mail any Company Proxy Statement, or any amendment or supplement thereto, to which the Holder reasonably and timely objects.

4.3.2 The Company shall not adjourn or postpone the Company Stockholder Meeting without the prior written consent of the Holder; <u>provided</u> that the Company may adjourn or postpone the Company Stockholder Meeting (i) to the extent necessary to ensure that any amendment or supplement to the Company Proxy Statement that the Company reasonably determines (following consultation with the Holder) is necessary to comply with applicable Law if provided to the stockholders of the Company within a reasonable amount of time in advance of the Company Stockholder Meeting, (ii) if, as of the time that the Company Stockholder Meeting is originally scheduled, there are insufficient shares of Common Stock represented at such meeting (either in person or by proxy) to constitute a necessary quorum to conduct the business of the Company Stockholder Meeting, or (iii) if, as of the time that the Company Stockholder Meeting is originally scheduled, adjournment or postponement of the Company Stockholder Meeting is necessary to enable the Company to solicit additional proxies required to obtain the Company Stockholder Approval.

4.3.3 The Holder agrees that, unless this Agreement has been terminated pursuant to <u>Section 5</u> prior thereto, at any meeting of the Company's stockholders held to obtain the Company Stockholder Approval, the Holder shall vote and shall cause its Affiliates to vote the outstanding shares of capital stock of the Company entitled to vote thereon beneficially owned by the Holder or its Affiliates in favor of the Company Stockholder Approval (except in the case of any vote applicable exclusively to the Disinterested Stockholders).

4.4 <u>Charter Amendments</u>. Promptly following the receipt of the Company Stockholder Approval by the Company and prior to the Closing, the Company shall cause to be filed with the Delaware Secretary of State, an amendment to the Charter reflecting (i) the Charter Amendment and (ii) the Series E-1 Charter Amendment.

4.5 <u>Dividends</u>. Following the receipt of the Company Stockholder Approval by the Company, and after the effectiveness of the Charter Amendment and the Series E-1 Charter Amendment and prior to the Closing, the Board of Directors of the Company shall declare a dividend payable in shares of Common Stock to holders of the issued and outstanding shares of Common Stock, and to the holders of the Series B 12% Cumulative, Convertible Preferred Stock and the Series D 6% Cumulative, Convertible Preferred Stock, in each case. on an as converted basis, providing for the payment to such holders of 0.3 shares of Common Stock per share of Common Stock outstanding, or on an as converted basis, as applicable, all as contemplated by the Company Proxy Statement (the "*Dividend*"). The Dividend shall be payable to such holders as they appear on the stock register of the Company on the applicable record date, which shall be fixed by the Board of Directors of the Company as a date prior to the Closing Date (the "*Record Date*"). The Board of Directors of the Company as a date prior to the Closing Date (the "*Record Date*").

4.6 Section 382 Standstill. Notwithstanding anything herein to the contrary, the Holder shall not either directly or indirectly (i) purchase any stock or securities of the Company, or (ii) sell any stock or securities of the Company, without obtaining the prior written consent of the Company. Unless the Company shall have affirmatively objected to such purchase or sale in writing within five (5) Business Days of the Holder's request, the Company will be deemed to have given such consent. The Company shall be permitted to object to such purchase or sale only if there is a material risk that such purchase or sale shall (i) cause an ownership shift in the ownership of the Company's equity for purposes of Section 382 of the Internal Revenue Code of 1986, as amended ("Section 382"), and (ii) directly or indirectly cause the Company's cumulative ownership shift for purposes of Section 382 to be equal to or

greater than 44 percentage points, in combination with other transactions that have occurred in the preceding 36-month period or that the Company reasonably expects will occur within the next six (6) months, such percentage change calculated for periods when there are more than one class of equity outstanding for purposes of Section 382 using the Hold Constant Principle described in Notice 2010-50, 2010-27 IRB 12. The Company shall make reasonable assumptions and presumptions in determining whether to object to such purchase or sale, which assumptions and presumptions may be reasonably conservative but which must be generally consistent with the regulations under Section 1.382-2T and any successor or similar provisions. In the event that the Company and the Holder shall disagree regarding the impact of any purchase or sale on the Company's cumulative ownership shift for purposes of Section 382, the Company and the Holder shall mutually agree on a nationally recognized accounting firm (other than an accounting firm that is the Company's auditor or principal outside tax advisor), and such accounting firm shall be permitted to conclusively determine the cumulative ownership shift for purposes of Section 382 taking into account the Holder's proposed sale or purchase. Such determination by the accounting firm shall be binding on both parties; provided, however, that in the event that the accounting firm shall determine that the Company's cumulative ownership shift for purposes of Section 382 taking into account the Holder's proposed sale or purchase exceeds 44%, the Holder shall be permitted to seek a private letter ruling from the Internal Revenue Service in order to determine the cumulative ownership shift for purposes of Section 382 taking into account the Holder's proposed sale or purchase. The Company shall provide reasonable cooperation to the Holder in seeking such ruling. The parties agree that if it is necessary to retain the services of an accounting firm to determine whether the Company's refusal to consent was a reasonable determination, then the costs of such accounting firm shall be shared equally by the parties (50% each), and that both parties shall act expeditiously in seeking such determination and shall urge the accounting firm to render its determination as expeditiously as reasonably possible.

4.7 <u>Related-Party Agreements</u>. On or prior to the Closing Date, the parties shall cause the Related-Party Contracts to be terminated effective as of the Closing.

4.8 <u>NYSE Listing</u>. The Company shall use reasonable best efforts to cause the Issued Shares to be approved for listing on the NYSE, subject to official notice of issuance, prior to the Closing Date.

4.9 <u>Public Disclosure</u>. The Company and the Holder agree that the initial press release to be issued with respect to the execution and delivery of this Agreement shall be in a form agreed to by the parties and that the parties shall consult with each other before issuing any press release or making any public announcement with respect to this Agreement and the Transactions and shall not issue any such press release or make any such public announcement without the prior consent of the other party (which shall not be unreasonably withheld, delayed or conditioned); <u>provided</u>, <u>however</u>, that a party may, without the prior written consent of the other party (but after prior consultation, to the extent practicable in the circumstances) issue such press release or make such public statement to the extent required by applicable Law or the applicable rules of any stock exchange.

4.10 <u>Takeover Statutes</u>. The Company and its Subsidiaries shall not take any action that would cause the Transactions to be subject to requirements imposed by any state takeover statute. If any "moratorium," "control share acquisition," "fair price," "supermajority," "affiliate transactions" or "business combination statute or regulation" or other similar state anti-takeover Laws and regulations may become, or may purport to be, applicable to the Transactions, each of the Company and the Holder shall grant such approvals and take such actions as are reasonably necessary so that the Transactions may be consummated as promptly as practicable under the terms contemplated hereby and otherwise act or eliminate or minimize the effects of such statute or regulation on the Transactions.

4.11 <u>Notification of Certain Matters</u>. The Company shall notify the Holder in writing promptly after obtaining knowledge of any change, event, occurrence, circumstance or effect that is more likely than not to cause any of the conditions to the Closing set forth in <u>Section 1.3</u> not to be timely satisfied. No notification given pursuant to this <u>Section 4.11</u> shall affect the representations, warranties, covenants or other agreements herein or affect the satisfaction or non-satisfaction of any conditions to the obligations of the parties hereto under this Agreement or otherwise limit or affect the remedies available to the Holder.

4.12 <u>Additional Holder Restrictions and Consent Rights</u>. Except (i) as otherwise required or expressly contemplated by this Agreement, (ii) as required by Law or (iii) with the prior written consent or waiver of the Holder (such consent not to be unreasonably withheld, conditioned or delayed), and so long as the Holder and its Affiliates in the aggregate directly or indirectly beneficially own 15% or more of the outstanding Common Stock (or securities convertible or exchangeable into Common Stock), the Company shall not, and shall not permit any Subsidiary to:

4.12.1 incur any Indebtedness in an aggregate amount in excess of \$50,000,000 outstanding at any time, except for (i) Indebtedness solely between or among the Company and any of its wholly owned Subsidiaries, (ii) letters of credit issued in the ordinary course of business consistent with past practices and (iii) trade credit or trade payables in the ordinary course of business consistent with past practices;

4.12.2 except as set forth in the applicable Capital Budget, make or authorize capital expenditures in any calendar year that, individually or in the aggregate, exceeds 50% of the amount set forth in the applicable Capital Budget;

4.12.3 sell, lease, license or otherwise dispose of or encumber any of its properties or assets that are material, individually or in the aggregate, to the business of the Company and its Subsidiaries or enter into any Contract with respect to the foregoing;

4.12.4 acquire or agree to acquire by merging or consolidating with, or by purchasing a substantial portion of the assets of, or by any other manner, any business of any Person or division thereof, or otherwise acquire or agree to acquire any assets, in all cases, for consideration in excess of \$50,000,000 in the aggregate in any calendar year, or enter into any Contract with respect to a joint venture, strategic alliance or partnership; or

4.12.5 amend its or any of its Subsidiaries' certificate of incorporation or bylaws or comparable organizational or governing documents in a manner adverse to the Holder, change the size of its board of directors or similar governing body or create an executive committee or similar committee to whom general authority of the board of directors is delegated, <u>provided</u>, <u>however</u>, that nothing contained in this <u>Section 4.12</u> shall prohibit the adoption of a Delaware forum selection bylaw.

4.13 Holder Preemptive Rights.

4.13.1 Prior to the Company directly or indirectly issuing additional Common Stock, or other equity securities of the Company convertible into, exercisable for or exchangeable into Common Stock ("*Options or Convertible Securities*"), without consideration or for a consideration per share less than the 30-Day VWAP (collectively, the "*New Common Stock*"), such 30-Day VWAP to be proportionately adjusted for any subdivisions, stock splits, combinations or other similar events with respect to the Common Stock, the Holder, so long as the Holder and its Affiliates in the aggregate directly or indirectly beneficially own 20% or more of the outstanding Common Stock (or securities convertible

or exchangeable into Common Stock), shall have the right to purchase the number of shares of New Common Stock in accordance with this <u>Section 4.13</u>. For purposes of the foregoing, the consideration per share received by the Company relating to Options or Convertible Securities, shall be determined by dividing: (i) the total amount, if any, received or receivable by the Company as consideration for the issue of such Options or Convertible Securities, plus the minimum aggregate amount of additional consideration payable to the Company upon the exercise or conversion of such Options or Convertible Securities, by (ii) the maximum number of shares of Common Stock (as set forth in the instruments relating thereto, without regard to any provision contained therein for a subsequent adjustment of such number) issuable upon the exercise or conversion of such Options or Convertible Securities.

4.13.2 So long as the Holder and its Affiliates in the aggregate directly or indirectly beneficially own 20% or more of the outstanding Common Stock (or securities convertible or exchangeable into Common Stock), the Company shall give the Holder at least ten (10) days' prior written notice (the "*Preemptive Right Notice*") of any proposed issuance of New Common Stock, which notice shall set forth in reasonable detail the proposed terms and conditions thereof and shall offer the Holder the opportunity to purchase its Pro Rata Share (which Pro Rata Share shall be calculated as of the date of such notice) of the New Common Stock at the same price, on the same terms and conditions as the New Common Stock is proposed to be issued by the Company, subject to <u>Section 4.13.4</u>. If the Holder wishes to exercise its preemptive rights, it must do so by delivering an irrevocable written notice to the Company within five (5) days after delivery by the Company of the Preemptive Right Notice, which notice shall state the dollar amount of New Common Stock that the Holder would like to purchase up to a maximum amount equal to the Holder's Pro Rata Share of the aggregate offering amount.

4.13.3 Notwithstanding anything in this <u>Section 4.13</u> to the contrary, the provisions of this <u>Section 4.13</u> shall not apply to issuances of New Common Stock by the Company as follows:

(a) Any issuance of New Common Stock upon the conversion or exercise or exchange (including redemption) of any options, warrants or other securities convertible into, or exercisable or exchangeable for, equity securities, <u>provided</u> that (in the case of such other options, warrants or other securities) such issuance is not made in violation of this Agreement, including this <u>Section 4.13</u>;

(b) Any issuance of New Common Stock pursuant to a compensatory plan, agreement or arrangement that has been approved by the Board of Directors of the Company (or the requisite holders of Common Stock (if required by applicable Law));

(c) Any issuance of New Common Stock in connection with any combination, stock split, recapitalization (other than in respect of any securities issued with respect to any new investment made in connection with such recapitalization) or reclassification of the Common Stock;

(d) Any issuance of New Common Stock as a dividend on any existing class or series of capital stock of the Company; or

(e) Any issuance of New Common Stock as consideration in any business combination or acquisition transaction involving the Company or any of its direct or indirect Subsidiaries or in any joint venture, in each case, not in violation of the terms of this Agreement.

4.13.4 If the Holder or the Company reasonably believes, based on the written opinion of legal counsel, that any regulatory or stockholder approval, including under applicable Antitrust Laws or applicable rules and regulations of any national securities exchange or over-the-counter market on which the Common Stock is listed for trading, is required prior to the Holder purchasing the New Common Stock it has elected to purchase pursuant to this <u>Section 4.13</u>, the Holder shall not be required to make such purchase, and the Company shall delay such issuance of the New Common Stock to either the Holder or the applicable third party until such approval has been obtained (or, in the case of applicable Antitrust Laws, the required filings have been completed and any applicable waiting period has expired). The Company shall use reasonable best efforts to comply promptly with all applicable regulatory requirements related to obtaining such approvals. In no event shall the Holder be deemed to have failed to purchase any New Common Stock for purposes of <u>Section 4.13</u> as a result of its inability to make such purchase prior to such compliance in accordance with this <u>Section 4.13.4</u>. If this <u>Section 4.13.4</u> applies, nothing in this <u>Section 4.13</u> shall prevent the Company from proceeding with the issuance of the New Common Stock to the proposed purchasers other than the portion of the New Common Stock that the Holder has elected to purchase pursuant to this <u>Section 4.13</u>.

4.14 Access to Information.

4.14.1 So long as the Holder and its Affiliates in the aggregate directly or indirectly beneficially own 10% or more of the outstanding Common Stock (or securities convertible or exchangeable into Common Stock), the Company shall afford the Holder and its Representatives reasonable access during business hours upon reasonable prior notice to (i) all of the properties, books, Contracts and records of the Company and each Subsidiary and (ii) all other information concerning the business of the Company, financial condition and results of operations of the Company or any Subsidiary as the Holder may reasonably request; <u>provided</u> that the Company may restrict the foregoing access to the extent that compliance with this <u>Section 4.14.1</u> would adversely affect any attorney-client or other legal privilege of the Company. The Company shall cooperate with the Holder from time to time to provide any information requested by the Holder reasonably and in good faith for a bona fide business purpose in connection with the Holder's ownership of the Common Stock.

4.14.2 No information or knowledge obtained in any investigation pursuant to this <u>Section 4.14</u> shall affect or be deemed to modify any representation, warranty, covenant, agreement or conditioned contained herein.

5. <u>TERMINATION</u>. This Agreement may be terminated and the Transactions may be abandoned at any time only as follows:

5.1 by mutual written agreement of the Company and the Holder at any time prior to the Closing; or

5.2 by either the Company or the Holder, if:

5.2.1 the Closing shall not have occurred on or by one minute after 11:59 p.m. (New York City Time) on January 17, 2022 (as such date may be extended pursuant to this Agreement, the "*Outside Date*"); provided, however, that the right to terminate this Agreement pursuant to this <u>Section 5.2.1</u> shall not be available to any party whose material breach of any provision of this Agreement primarily results in the failure of the Closing to occur by the Outside Date; or

5.2.2 any Governmental Authority of competent jurisdiction shall have enacted, issued, promulgated, enforced or entered any applicable Law permanently restraining, enjoining, prohibiting or making illegal prior to the Closing, the consummation of the Exchange, and such applicable Law shall have become final and nonappealable; <u>provided</u>, that the right to terminate this Agreement pursuant to this <u>Section 5.2.2</u> shall not be available to any party whose material breach of any provision of this Agreement primarily results in such applicable Law to be enacted, issued, promulgated, enforced or entered; or

5.2.3 the Company Stockholder Approval is not obtained at the Company Stockholder Meeting or any adjournment or postponement thereof at which a vote is taken thereon.

5.3 by the Holder, if:

5.3.1 at any time prior to the Closing, an Adverse Recommendation Change shall have occurred; or

5.3.2 at any time prior to the Closing, a breach of any representation or warranty or failure to perform any covenant or agreement on the part of the Company set forth in this Agreement shall have occurred that would cause any of the conditions to Closing set forth in <u>Section 1.3.2</u> or <u>Section 1.3.3</u> not to be satisfied, and such condition is incapable of being satisfied by the Outside Date or if capable of being cured in such time frame, shall not have been cured within the earlier of the Outside Date and thirty (30) days of the date the Holder gives the Company notice of such breach or failure to perform; <u>provided</u>, that the right to terminate this Agreement pursuant to this <u>Section 5.3.2</u> shall not be available to the Holder if the Holder is then in material breach of this Agreement; or

5.4 By the Company, if:

5.4.1 at any time prior to obtaining the Company Stockholder Approval, the Board of Directors of the Company and, if applicable, the Special Committee authorizes the Company to enter into a definitive agreement concerning a Superior Proposal and the Company contemporaneously enters into such definitive agreement concerning such Superior Proposal and, in connection with the termination of this Agreement, the Company (i) pays to the Holder in immediately available funds a fee of \$3,500,000 (the "Termination Fee") by wire transfer of same-day funds within one (1) Business Day following the date of termination of this Agreement, (ii) upon the closing of such Superior Proposal transaction (if such closing occurs), causes the Series E-1 Preferred and the Series F-1 Preferred to be redeemed for cash, at the Liquidation Preference (as defined in the Series E-1 Certificate of Designations), including the Participation Rights Value (as defined in the Series E-1 Certificate of Designations), in the case of the Series E-1 Preferred, and the Redemption Price (as defined in the Series F-1 Certificate of Designations) in the case of the Series F-1 Preferred, such redemption to occur contemporaneously with the closing of such Superior Proposal transaction and (iii) if the closing of such Superior Proposal transaction does not occur, pays to the Holder in immediately available funds a fee of \$7,000,000 by wire transfer of same-day funds within one (1) Business Day following the date of termination of the definitive agreement concerning such Superior Proposal or the occurrence of the "outside date", "end date" or similar event in respect thereof; provided, that in the case of this clause (iii), the Holder shall also have the option, at its election, to reinstate this Agreement and effectuate the Exchange at the lower of the 30-Day VWAP or the volume weighted average price of the Common Stock for the 30 trading days immediately prior to such reinstatement on the NYSE as reported by Bloomberg Financial Markets or, if Bloomberg Financial Markets is not then reporting such prices, by a comparable reporting service of national reputation mutually selected by the Company and Holder, and in the event such election is made by the Holder, each of the parties hereto shall cooperate reasonably and in good faith to take, or cause to be taken, all actions necessary or appropriate, and to execute all documents, instruments and agreements, to give effect to the reinstatement of this Agreement and to consummate the transactions contemplated hereby and thereby;

5.4.2 at any time prior to the Closing, a breach of any representation or warranty or failure to perform any covenant or agreement on the part of the Holder set forth in this Agreement shall have occurred that (A) would (x) cause any of the conditions to Closing set forth in <u>Section 1.3.1</u> or <u>Section 1.3.3</u> not to be satisfied or (y) reasonably be expected to prevent, materially delay or materially impede the consummation of the Exchange and (B) is incapable of being satisfied by the Outside Date or if capable of being cured in such time frame, shall not have been cured within the earlier of the Outside Date and thirty (30) days of the date the Company gives the Holder notice of such breach or failure to perform; <u>provided</u>, that the right to terminate this Agreement pursuant to this <u>Section 5.4.2</u> shall not be available to the Company if the Company is then in material breach of this Agreement.

5.5 The party desiring to terminate this Agreement pursuant to this <u>Section 5</u> (other than pursuant to <u>Section 5.1</u>) shall give written notice of such termination to the other party.

5.6 Effect of Termination. If this Agreement is terminated pursuant to this <u>Section 5</u>, this Agreement shall become void and of no effect without liability of any party (or any stockholder, director, officer, employee, agent, consultant or Representative of such party) to the other party hereto; <u>provided</u>, that (a) the provisions of <u>Section 5.4</u>, this <u>Section 5.6</u>, <u>Section 6</u> and <u>Section 7</u> and the Confidentiality Agreement shall survive any termination hereof pursuant to <u>Section 5</u> and (b) neither the Company nor the Holder shall be relieved or released from any liabilities or damages arising out of its fraud or Willful and Material Breach of any provision of this Agreement.

6. INDEMNIFICATION.

6.1 <u>Indemnification by the Company</u>. The Company agrees to indemnify the Holders, its Affiliates and the respective Representatives (the "*Holder Related Parties*") from all costs, losses, liabilities, damages, or expenses of any kind or nature whatsoever (including any expenses reasonably incurred in connection with investigating, defending or preparing to defend any Claim) ("*Losses*"), and hold each of them harmless against, any and all actions, suits, proceedings (including any investigations, litigation or inquiries), demand, and causes of actions ("*Claims*"), and, in connection therewith, and promptly upon demand, pay or reimburse each of them for all Losses, whether or not involving a Third Party Claim, as a result of, arising out of, or in any way related to:

6.1.1 any inaccuracy of any of the representations and warranties made by the Company in this Agreement or in any certificate delivered by or on behalf of the Company pursuant to this Agreement as of the date made or (unless such representation and warranty is made as of a specific date) as of the Closing Date;

6.1.2 the breach or non-fulfillment of any covenant, agreement or obligation of the Company contained herein; or

6.1.3 any Legal Proceeding arising out of the Transactions;

<u>provided</u> that, in the case of <u>Section 6.1.1</u>, such claim for indemnification is made prior to the applicable expiration of such representation or warranty as set forth in <u>Section 7.14</u>; <u>provided</u>, <u>however</u>, that for purposes of determining when an indemnification claim has been made, the date upon which a Holder Related Party shall have given notice (stating in reasonable details the basis of the indemnification) to the Company shall constitute the date upon which such claim has been made.

6.2 <u>Indemnification by the Holder</u>. The Holder agrees to indemnify the Company, its Affiliates and their respective Representatives (collectively, the "*Company Related Parties*") from Losses, and hold each of them harmless against, any and all Claims, and, in connection therewith, and promptly upon demand, pay or reimburse each of them for all Losses, whether or not involving a Third Party Claim, as a result of, arising out of, or in any way related to:

6.2.1 any inaccuracy of the representations and warranties made by the Holder in this Agreement or in any certificate delivered by or on behalf of the Holder pursuant to this Agreement as of the date made or (unless such representation and warranty is made as of a specific date) as of the Closing Date; or

6.2.2 the breach or non-fulfillment of any covenant, agreement or obligations of the Holder contained herein;

<u>provided</u> that, in the case of <u>Section 6.2.1</u>, such claim for indemnification relating to a breach of any representation or warranty is made prior to the expiration of such applicable representation or warranty pursuant to <u>Section 7.14</u>; <u>provided</u>, <u>however</u>, that for purposes of determining when an indemnification claim has been made, the date upon which a Company Related Party shall have given notice (stating in reasonable detail the basis of the claim for indemnification) to the Holder shall constitute the date upon which such claim has been made.

6.3 Limitations on Liability; Materiality.

6.3.1 The Holder Related Parties shall not be entitled to assert any claim for indemnification under <u>Section 6.1.1</u> unless and until (i) with respect to any individual item of Loss, such item is greater than \$100,000 (the "*Per Claim Threshold*") and (ii) the aggregate Losses (excluding any particular Loss that does not exceed the Per Claim Threshold) actually suffered or incurred for which indemnification would otherwise be available to the Holder Related Parties exceed \$3,500,000 (the "*Deductible*"), at which point indemnification will be available to the Holder Related Parties only for those Losses in excess of the Deductible (subject to <u>clause (i)</u>) up to the Cap; <u>provided</u> that the limitations set forth in this sentence shall not apply to (A) claims based on fraud or Willful and Material Breach or (B) any claim pursuant to <u>Section 6.1.1</u> relating to a breach of any of the Company Fundamental Representations.

6.3.2 The Company Related Parties shall not be entitled to assert any claim for indemnification under <u>Section 6.2.1</u> unless and until (i) with respect to any individual item of Loss, such item is greater than the Per Claim Threshold and (ii) the aggregate Losses (excluding any particular Loss that does not exceed the Per Claim Threshold) actually suffered or incurred for which indemnification would otherwise be available to the Company Related Parties exceed the Deductible, at which point indemnification will be available to the Company Related Parties only for those Losses in excess of the Deductible (subject to <u>clause (i)</u>) up to the Cap; <u>provided</u> that the limitations set forth in this sentence shall not apply to (A) claims based on fraud or Willful and Material Breach or (B) any claim pursuant to <u>Section 6.2.1</u> relating to a breach of any of the Holder Fundamental Representations.

6.3.3 Notwithstanding anything to the contrary in this Agreement, for purposes of determining (i) whether there has been a breach of any representation or warranty set forth in this Agreement or in any certificate delivered pursuant to this Agreement and (ii) the amount of Losses for which any Holder Related Party or Company Related Party, as applicable, may be entitled to indemnification under this <u>Section 6</u>, each such representation or warranty shall be deemed to have been made without any qualifications or limitations as to materiality (including any qualifications or limitations made by reference to a Material Adverse Effect or a Holder Material Adverse Effect, as applicable.)

6.4 Indemnification Procedure.

6.4.1 A claim for indemnification for any matter not involving a Third Party Claim must be asserted by notice to the party from whom indemnification is sought; <u>provided</u>, <u>however</u>, that failure to so notify the indemnifying party shall not preclude the indemnified party from any indemnification which it may claim in accordance with this <u>Section 6</u>, except as otherwise provided in <u>Section 6.1</u> and <u>Section 6.2</u>.

6.4.2 Promptly after any Company Related Party or Holder Related Party (hereinafter, the "Indemnified Party") has received notice of any indemnifiable claim hereunder, or the commencement of any Legal Proceeding by a third person, which the Indemnified Party believes is an indemnifiable claim under this Agreement (each a "Third Party Claim"), the Indemnified Party shall give the indemnitor hereunder (the "Indemnifying Party") reasonably prompt written notice of such Third Party Claim, but failure to so notify the Indemnifying Party will not relieve the Indemnifying Party from any liability it may have to such Indemnified Party hereunder except to the extent that the Indemnifying Party is materially prejudiced by such failure. Such notice shall state the nature and the basis of such Third Party Claim to the extent then known. The Indemnifying Party shall have the right to defend and settle, at its own expense and by its own counsel who shall be reasonably acceptable to the Indemnified Party, any such matter. If the Indemnifying Party undertakes to defend or settle, it shall promptly, and in no event later than thirty (30) days, notify the Indemnified Party of its intention to do so, and the Indemnified Party shall cooperate with the Indemnifying Party and its counsel in all commercially reasonable respects in the defense thereof and the settlement thereof. Such cooperation shall include furnishing the Indemnifying Party with any books, records and other information reasonably requested by the Indemnifying Party and in the Indemnified Party's possession or control. Such cooperation of the Indemnified Party shall be at the cost of the Indemnifying Party. After the Indemnifying Party has notified the Indemnified Party of its intention to undertake to defend or settle any such asserted liability, and for so long as the Indemnifying Party uses commercially reasonable efforts to pursue such defense, the Indemnifying Party shall not be liable for any additional legal expenses incurred by the Indemnified Party in connection with any defense or settlement of such asserted liability; provided, however, that the Indemnified Party shall be entitled (i) at its expense, to participate in the defense of such asserted liability and the negotiations of the settlement thereof and (ii) if (A) the Indemnifying Party has, within thirty (30) days of when the Indemnified Party provides written notice of a Third Party Claim, failed (1) to assume the defense or employ counsel reasonably acceptable to the Indemnified Party and (2) notify the Indemnified Party of such assumption or (B) the defendants in any such action include both the Indemnified Party and the Indemnifying Party and counsel to the Indemnified Party shall have concluded that there may be reasonable defenses available to the Indemnified Party that are different from or in addition to those available to the Indemnifying Party or if the interests of the Indemnified Party reasonably may be deemed to conflict with the interests of the Indemnifying Party, then the Indemnified Party shall have the right to select one separate counsel plus one local counsel and to assume such legal defense of such indemnified party and otherwise to participate in the defense of such action, with the reasonable expenses and reasonable fees of such separate counsel and such local counsel, and other reasonable expenses related to such participation to be reimbursed by the Indemnifying Party as incurred, provided that, under such circumstances, the Indemnifying Party and Indemnified Party shall cooperate in good faith to formulate a defense plan which shall allocate work between such counsels so as to avoid duplication of effort and management of legal fees. If the Indemnifying Party elects to control the defense of the Third Party Claim, the Indemnified Party shall not consent to the entry of any judgment or enter into any settlement with respect to the Third Party Claim, or take any voluntary action prejudicial to the determination of the Third Party Claim, without the prior written consent of the Indemnifying Party (such consent not to be unreasonably withheld, conditioned or delayed). Notwithstanding any other provision of this Agreement, the Indemnifying Party shall not settle any indemnified claim without the consent of the Indemnified Party, unless the settlement thereof imposes no liability or obligation on, and includes a complete release from liability (in a form reasonably satisfactory to such Indemnified Party) of, and does not include any admission of wrongdoing or malfeasance by, the Indemnified Party.

6.5 <u>Waiver of Consequential and Punitive Damages</u>. No party shall have any liability to any other party (or to any other Holder Related Party or Company Related Party) for any special, consequential or punitive damages ("*Excluded Claims*") as a result of any claim arising under or relating to the Transactions including, without limitation, any claim under this <u>Section 6</u> and each party hereby waives to the maximum extent permitted by applicable Law any such Excluded Claim; <u>provided</u>, <u>however</u>, that such limitation shall not prevent any Holder Related Party or Company Related Party from recovering under this <u>Section 6</u> for any such damages to the extent such damages are payable to a third party in connection with any Third Party Claims.

6.6 Exclusive Remedy. Other than with respect to claims for fraud or Willful and Material Breach, each party acknowledges and agrees that, following the Closing, subject to <u>Section 7.9</u>, such party's sole and exclusive remedy with respect to any and all claims for any breach of any representation, warranty, covenant, agreement or obligation set forth herein or otherwise relating to the subject matter of this Agreement, shall be pursuant to the indemnification provisions set forth in this <u>Section 6</u>, or in the case of termination, as otherwise expressly provide in <u>Section 5</u>. In furtherance of the foregoing, each party hereby waives, to the fullest extent permitted under Law, any and all rights, claims and causes of action for any breach of any representation, warranty, covenant, agreement or obligation set forth herein or otherwise relating to the subject matter of this Agreement it may have against the other parties hereto and their Affiliates and each of their respective Representatives arising under or based upon any Law, except pursuant to the indemnification provisions set forth in this <u>Section 6</u> or with respect to claims for fraud or Willful and Material Breach. Notwithstanding the foregoing, nothing in this <u>Section 6</u> shall limit any Person's right to seek and obtain specific performance or any other form of equitable relief to which any Person shall be entitled under applicable Law or assert a claim of fraud or Willful and Material Breach.

6.7 <u>Effect of Investigation</u>. Neither the representations, warranties and covenants of the Company, nor the right to indemnification of any Holder Related Party making a claim under this <u>Section 6</u> with respect thereto, shall be affected or deemed waived by reason of any investigation made by or on behalf of a Holder Related Party (including by any of its Representatives) or by reason of the fact that a Holder Related Party or any of its Representatives knew or should have known that any such representation or warranty is, was or might be inaccurate by reason of a Holder Related Party waiver of any condition set forth in <u>Section 1.3</u>.

7. GENERAL PROVISIONS.

7.1 <u>Binding Effect; Successors and Assigns</u>. The terms and conditions of this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Nothing in this Agreement, express or implied, is intended to confer upon any party other than the parties hereto or their respective successors and permitted assigns any rights, remedies, obligations, or liabilities under or by reason of this Agreement, except as expressly provided in this Agreement.

7.2 <u>Governing Law; Jurisdiction</u>. This Agreement shall be governed by and construed in accordance with the internal laws of the State of Delaware without giving effect to any choice or conflict of laws provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of laws of any jurisdictions other than those of the State of Delaware. The parties hereto hereby irrevocably submit to the exclusive jurisdiction of the Court of Chancery of the State of Delaware, New Castle County, or, if that court does not have jurisdiction, a federal court sitting in Wilmington, Delaware, in respect of the interpretation and enforcement of the provisions of this Agreement and of the documents referred to in this Agreement, and in respect of the Transactions and thereby, and hereby waive, and agree not to assert, as a defense in any Legal Proceeding thereof or thereof, that is not subject thereto or that such Legal Proceeding may not be brought or is not maintainable in said courts or that the venue thereof may not be appropriate or that this Agreement or any such document may not be enforced in or by such courts, and the parties hereto irrevocably agree that all claims with respect to such Legal Proceeding shall be heard and determined in such courts. The parties hereto hereby consent to and grant any such court jurisdiction over the person of such parties and over the subject matter of such dispute and agree that mailing of process or other papers in connection with any such action or proceeding in the manner provided in <u>Section 7.6</u> or in such other manner as may be permitted by applicable Law, shall be value and sufficient service thereof.

7.3 WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM (WHETHER BASED ON CONTRACT, TORT, OR OTHERWISE) ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE ACTIONS OF ANY PARTY HERETO IN NEGOTIATION, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HERETO.

7.4 <u>Counterparts; Facsimile</u>. This Agreement may be executed and delivered by facsimile signature and in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Counterparts may be delivered via facsimile, electronic mail (including pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, *e.g.*, www.docusign.com) or other transmission method and any counterpart so delivered shall be deemed to have been duly and validly delivered and be valid and effective for all purposes.

7.5 <u>Titles and Subtitles</u>. The titles and subtitles used in this Agreement are used for convenience only and are not to be considered in construing or interpreting this Agreement.

7.6 <u>Notices</u>. All notices, demands and other communications provided for hereunder shall be in writing and shall be given by registered or certified mail, return receipt requested, electronic mail, return receipt requested, facsimile, air courier guaranteeing overnight delivery or personal delivery to the following addresses:

(a) If to the Holder:

LSB Funding LLC 600 Steamboat Road Greenwich, CT 06830 Attn: Legal Department Facsimile: 646-828-2851 Email: legal@eldridge.com

with a copy to (which shall not constitute notice):

Wilmer Hale LLP 7 World Trade Center 250 Greenwich Street

New York, NY 10007 Attn: Justin Ochs and Glenn Pollner Email: justin.ochs@wilmerhale.com and glenn.pollner@wilmerhale.com

(b) If to the Company:

LSB Industries, Inc. 3503 NW 63rd Street, Suite 500 Oklahoma City, Oklahoma, 73116 Attention: Mark Behrman Facsimile: (405) 235-5067 (with such fax to be confirmed by telephone to (405) 235-4546)

with a copy to:

LSB Industries, Inc. 3503 NW 63rd Street, Suite 500 Oklahoma City, Oklahoma, 73116 Attention: Michael Foster

with a copy to (which shall not constitute notice):

Blank Rome LLP 1271 6th Avenue New York, New York 10020 Attn: Robert Mittman and Barry Genkin Email: RMittman@Blankrome.com and Genkin@Blankrome.com

and

Ropes & Gray LLP 1211 Avenue of the Americas New York, New York 10036 Attention: Leonard Klingbaum; Craig E. Marcus and Robb Tretter Email: Leonard.Klingbaum@ropesgray.com; Craig.Marcus@ropesgray.com; and Robb.Tretter@ropesgray.com

7.7 <u>Amendments and Waivers</u>. Any term of this Agreement may be amended, terminated or waived only with the written consent of each of (i) the Company and (ii) the Holder. Any amendment or waiver effected in accordance with this <u>Section 7.7</u> shall be binding upon the Holder and each transferee of the Issued Shares and the Company. Any amendment, supplement or waiver of any term of this Agreement shall be effective only in the specific instance and for the specific purpose for which such amendment, supplement or waiver is made or given.

7.8 <u>Severability</u>. The invalidity or unenforceability of any provision of this Agreement shall in no way affect the validity or enforceability of any other provision hereof.

7.9 <u>Specific Performance</u>. Each party acknowledges that its obligations hereunder are unique and recognizes and affirms that money damages may be inadequate and that the other party hereto may have no adequate remedy at law. Accordingly, each party agrees that the other party shall have the right, in addition to any other rights and remedies in law or in equity, to enforce such party's rights and the other party's obligations hereunder not only by an action or actions for damages but also by an action or actions for specific performance, injunctive and/or other equitable relief (without posting bond or security).

7.10 No Assignment. No party hereto may assign any of its rights or obligations hereunder without the prior written consent of the other party; provided, however, that the Holder may assign its rights and obligations hereunder to any of its Affiliates without the prior written consent of the Company; provided, further, that such Affiliate of the Holder is reasonable financially capable of performing the obligations of the Holder set forth in this Agreement.

7.11 <u>Waiver</u>. By executing this Agreement, the parties hereto agree to and hereby waive, on a one-time basis with respect to the Exchange, any prohibition on the Exchange, including under (i) the Series E-1 Certificate of Designations, (ii) the Series F-1 Certificate of Designations or (iii) the Purchase Agreement to the extent applicable. For the avoidance of doubt, any such waiver contemplated in the preceding sentence is permitted only to the sole extent necessary to permit the consummation of the Exchange.

7.12 Registration Rights Agreement.

7.12.1 Effective as of the Closing, the Registration Rights Agreement is hereby amended as follows (as amended, the "*Amended Registration Rights Agreement*"):

7.12.2 the defined term "<u>Exchange Agreement</u>" means the Securities Exchange Agreement by and between the Company and Purchaser dated July 19, 2021." is inserted in ARTICLE I prior to the definition of "Holder";

7.12.3 the defined term "<u>Issued Shares</u>" has the meaning specified in the Exchange Agreement." is inserted in ARTICLE I prior to the definition of "<u>Liquidated Damages</u>"; and

7.12.4 the defined term "Registrable Securities" is amended to replace "Agreement and" with "," and the phrase "(iv) and the Common Stock comprising the Issued Shares (provided that such Common Stock shall not be deemed to be "Registrable Securities" solely for purposes of Section 2.01, but such Common Stock shall be "Registrable Securities" for all other purposes, including the definition of "Effectiveness Period" and Section 2.04)" is inserted in front of ", in each case,";

7.12.5 the defined term "Effectiveness Period" is amended to replace the phrase "(x) the tenth anniversary of the date hereof" with the phrase "(x) September 30, 2031"; and

7.12.6 for purposes of Section 2.04 of the Registration Rights Agreement, references to the term "Registration Statement" shall be deemed to refer to one or more registration statements that may be requested by the Holder to be filed by the Company under the Securities Act for purposes of effecting the demand rights contemplated in Section 2.04 thereof, and is not limited to a Registration Statement filed pursuant to Section 2.01 thereof (it being understood and agreed, for the avoidance of doubt, that the parties intend for the Holder to be entitled to the demand rights contemplated by Section 2.04 thereof with respect to its Registrable Securities, in addition to the piggyback rights and other rights contemplated by the Registration Rights Agreement).

7.13 Further Assurances. From time to time after the date hereof, each of the parties hereto shall use its commercially reasonable efforts to take, or cause to be taken, all actions necessary or appropriate to give effect to the transactions contemplated by this Agreement.

7.14 <u>Survival of Representations and Warranties</u>. The Fundamental Representations shall survive the execution and delivery of this Agreement for the applicable statute of limitations period, and the other representations and warranties set forth herein shall survive for a period of twelve (12) months following the Closing Date, regardless of any investigation made by or on behalf of the Company or the Holder, in each case other than with respect to claims based on fraud or Willful and Material Breach. The covenants and agreements in this Agreement shall terminate at the Closing, except that the covenants and agreements that explicitly contemplate performance after the Closing shall survive the Closing until fully performed in accordance with their respective terms. The intended effect of termination of representations and warranties pursuant to this <u>Section 7.14</u> is to bar, from and after the date of termination, any claim or cause of action based on alleged inaccuracy of such representation or breach of such representation and warranty with respect to claims based on fraud or Willful and Material Breach. For the avoidance of doubt, if notice of a claim or cause of action based on alleged inaccuracy of such representation and warranty with respect thereto is delivered to the party from whom indemnification is sought before expiration of the applicable representation and warranty then the applicable representation and warranty shall survive until, but only for purposes of, the resolution of the matter covered by such notice.

7.15 <u>Costs and Expenses</u>. Except where this Agreement or any other Transaction Document expressly provides otherwise, each party shall bear its own fees, costs and expenses arising out of or connected with this Agreement, the other Transaction Documents, and the performance of the Transactions; <u>provided</u>, <u>however</u>, that in the event the Agreement is terminated due to failure of the parties to obtain approval of the Transactions under the HSR Act as set forth in <u>Section 4.2.3</u> and the cause of such failure to obtain approval is reasonably attributable to the Holder, then the fees, costs and expenses involved in seeking such approval shall be borne one-half (50%) by the Holder and one-half (50%) by the Company. Notwithstanding the foregoing, whether or not the Closing occurs, the Company shall promptly pay all reasonable attorney's fees and expenses of the Holder and its Affiliates, and its respective Representatives, relating to this Agreement or the other Transaction Documents, and the performance of the Transactions, by wire of immediately available funds as directed by the Holder in writing; <u>provided</u> that, without limitation of anything set forth in <u>Section 6</u>, such expenses described in this sentence shall not exceed \$550,000 without the Company's prior written consent; <u>provided further</u> that the Company shall not have any obligation to pay such fees and expenses if the Agreement is terminated by the Company pursuant to <u>Section 5.4.2</u> due to a material breach of any representation or warranty or material failure to perform any covenant or agreement on the part of the Holder, which breach or failure to perform.

[Signature pages follow]

IN WITNESS WHEREOF, the parties hereto have duly executed this Securities Exchange Agreement as of the date first set forth above.

LSB INDUSTRIES, INC.

By: /s/ Mark T. Behrman Name: Mark T. Behrman Title: Chief Executive Officer and President

[Signature Page to Securities Exchange Agreement]

IN WITNESS WHEREOF, the parties hereto have duly executed this Securities Exchange Agreement as of the date first set forth above.

LSB FUNDING LLC

By: /s/ Anthony D. Minella Name: Anthony D. Minella

Title: Manager

[Signature Page to Securities Exchange Agreement]



FOR IMMEDIATE RELEASE

LSB INDUSTRIES, INC. ANNOUNCES TRANSACTION TO EXCHANGE OUTSTANDING SERIES E-1 AND F-1 PREFERRED STOCK FOR COMMON STOCK; PLANS TO PAY 0.3:1 SPECIAL COMMON STOCK DIVIDEND TO EXISTING COMMON STOCKHOLDERS

Transformative Transaction to Simplify Capital Structure, Lower Cost of Capital and Provide Greater Financial Flexibility to Pursue Growth Initiatives

Oklahoma City, Oklahoma – July 20th, 2021—LSB Industries, Inc. ("LSB" or "the Company"), (NYSE: LXU), led by a Special Committee of the Board of Directors representing the disinterested stockholders of the Company (the "Special Committee") today announced that it has signed a definitive agreement (the "Exchange Agreement") with LSB Funding LLC, an affiliate of Eldridge, to exchange the shares of LSB Series E-1 and Series F-1 Redeemable Preferred Stock held by Eldridge for shares of LSB common stock. Under the terms of the agreement, LSB would exchange, at the closing, approximately \$300 million of preferred stock held by Eldridge into an equivalent value of LSB common stock based on an exchange price of \$6.16, which is equal to the 30-day volume weighted average price as of the date of the Exchange Agreement. In connection with the transaction, existing unaffiliated LSB common stockholders will receive a special dividend in the form of 0.30 shares of LSB common stock for every share owned as of the record date.

Transaction Highlights:

- Eliminates the current financial impact and repayment of the accrued compounding preferred stock and future accruing dividends at 14.5% (increasing to 16.0% in April 2023) unburdening the Company and unlocking shareholder value.
- The Special Committee, Board of Directors and LSB management believe this could lead to a rating upgrade potentially allowing the Company to refinance its senior secured notes at a lower interest rate and on improved terms, which would reduce its cash interest expense and overall cost of capital.
- Improves the Company's financial flexibility allowing it to pursue organic growth initiatives, including in green ammonia and clean energy and accretive M&A opportunities.
- Preserves the Company's significant tax attributes, including approximately \$620 million of federal net operating losses, thereby protecting potentially significant future cash savings and stockholder value.

Mark Behrman, LSB's President and CEO, stated, "The Special Committee, the Board of Directors and management believe that the exchange of our outstanding Series E-1 and F-1 preferred stock for LSB common stock relieves the Company and our common stockholders from the expensive, compounding burden of the payment-in-kind dividend this preferred stock carries. This measure not only improves our current capital structure but, we believe, combined with the favorable credit markets will allow us to refinance our senior secured notes on more favorable terms than our current senior secured notes and provide us with the financial flexibility needed to grow our business organically and through strategic M&A; while maintaining our significant federal net operating losses which we believe we will start utilizing this year."

"Over the last several years we have evaluated options to reduce or eliminate our preferred stock. However, during that period, the nitrogen chemical market pricing environment has not been helpful as selling prices had been at multi-year lows. That prevented us from generating sufficient free cash flow to make cash dividend payments on the preferred stock and therefore caused the dividends thereon to accrue at a compounded rate, which significantly increased the preferred stock balance. While the nitrogen chemical market pricing environment has significantly improved and we are seeing selling prices that are now at multi-year highs, we are in a commodity business where pricing can be volatile and change quickly. As a result, we believe that now is an opportune time to take these actions especially given our desire to refinance our senior secured notes and our need for flexibility to take advantage of numerous attractive organic growth opportunities, including the emerging blue/green ammonia and clean energy markets. Additionally, we regularly evaluate M&A prospects that we believe could be accretive to earnings as a result of the increased scale and expanded production capabilities that they would provide us. We believe that the exchange of our outstanding Series E-1 and Series F-1 preferred stock and the overall simplification of our capital structure as well as the potential refinancing of our senior secured notes will be a critical step in unlocking our ability to take advantage of these opportunities."

The LSB Board of Directors delegated authority to a committee of its independent and disinterested directors with the mandate to act in the interest of the disinterested holders of LSB's common stock with respect to LSB's evaluation of potential business opportunities, including potential transactions involving Eldridge. The highly qualified, independent and disinterested directors have been acting through the Special Committee and are empowered to act in the interests of the holders of LSB common stock with respect to any business opportunity that would require the holdings of Eldridge to be modified, converted or exchanged other than pursuant to the existing terms of the Securities Purchase Agreement dated December 4, 2015, and other related agreements. Further, any such change or action was irrevocably conditioned on both the approval of the Special Committee and the affirmative vote of the disinterested members of the Board of Directors and the affirmative vote of a majority of the outstanding shares of common stock of the Company held by disinterested stockholders.

The Special Committee is composed of Richard W. Roedel, former Chairman and CEO of BDO Seidman LLP; Lynn F. White, former Vice President, Corporate Development at CF Industries; Diana M. Peninger, President & CEO of Reproductive Solutions; and Richard S. Sanders, Jr. former Vice President of Manufacturing for Terra Industries. Mr. Roedel and Mr. White led the Special Committee as co-chairs.

In conjunction with its comprehensive evaluation of the exchange transaction, the Special Committee:

- Retained independent financial advisor, Houlihan Lokey Capital, Inc. and
- Retained independent legal counsel, Blank Rome

Completion of the exchange transaction is subject to a number of customary closing conditions, including receipt of stockholder approval from the holders of a majority of the shares of our outstanding common stock not held by Eldridge or any of its affiliates. LSB expects to file a preliminary proxy for a Special Meeting of Stockholders and deliver additional information related to the special meeting to stockholders within the next few weeks. Results of the stockholder vote will be tabulated at the Special Meeting of Stockholders expected to be held in the third quarter of 2021.

The Company is represented by its legal counsel, Ropes & Gray LLP, and financial advisor, Jefferies LLC. Eldridge is represented by WilmerHale.

About LSB Industries, Inc.

LSB Industries, Inc., headquartered in Oklahoma City, Oklahoma, manufactures and sells chemical products for the agricultural, mining, and industrial markets. The Company owns and operates facilities in Cherokee, Alabama, El Dorado, Arkansas and Pryor, Oklahoma, and operates a facility for a global chemical company in Baytown, Texas. LSB's products are sold through distributors and directly to end customers primarily throughout the United States. Additional information about the Company can be found on its website at www.lsbindustries.com.

About Eldridge

Eldridge invests in businesses across the Insurance, Asset Management, Technology, Mobility, Sports & Gaming, Media & Music, Real Estate, and Consumer landscapes. The firm seeks to build and grow businesses led by proven management teams that have demonstrated leadership and experience to scale an enterprise. Eldridge is headquartered in Greenwich, Connecticut, with additional offices in Beverly Hills, New York, and London. Additional information about Eldridge can be found on its website at www.eldridge.com

Forward-Looking Statements

Statements in this release that are not historical are forward-looking statements within the meaning of the U.S. Private Securities Litigation Reform Act of 1995. These forward-looking statements, which are subject to known and unknown risks, uncertainties and assumptions about us, may include projections of our future financial performance including the effects of the COVID-19 pandemic and anticipated performance based on our growth and other strategies and anticipated trends in our business. These statements are only predictions based on our current expectations and projections about future events. There are important factors that could cause our actual results, level of activity, performance or actual achievements to differ materially from the results, level of activity, performance or anticipated achievements expressed or implied by the forward-looking statements. Significant risks and uncertainties may relate to, but are not limited to, our ability to consummate the exchange transaction on the terms described herein or at all, business and market disruptions related to the COVID-19 pandemic, market conditions and price volatility for our products and feedstocks, as well as global and regional economic downturns, including as a result of the COVID-19 pandemic, competitive, environmental, political, legal and regulatory factors. These and other risk factors are discussed in the Company's filings with the Securities and Exchange Commission (SEC).

Moreover, we operate in a very competitive and rapidly changing environment. New risks and uncertainties emerge from time to time, and it is not possible for our management to predict all risks and uncertainties, nor can management assess the impact of all factors on our business or the extent to which any factor, or combination of factors, may cause actual results to differ materially from those contained in any forward-looking statements. Although we believe the expectations reflected in the forward-looking statements are reasonable, we cannot guarantee future results, level of activity, performance or achievements. Neither we nor any other person assumes responsibility for the accuracy or completeness of any of these forward-looking statements. You should not rely upon forward-looking statements as predictions of future events. Unless otherwise required by applicable laws, we undertake no obligation to update or revise any forward-looking statements, whether because of new information or future developments.

No Offer or Solicitation

This communication is not intended to and shall not constitute an offer to buy or sell or the solicitation of an offer to buy or sell any securities, or a solicitation of any vote or approval, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made, except by means of a prospectus meeting the requirements of Section 10 of the U.S. Securities Act of 1933, as amended, or an exemption from the registration requirements thereof.

Additional Information about the Exchange Transaction and Where to Find It

In connection with the proposed transaction, LSB intends to file with the Securities and Exchange Commission ("SEC") a proxy statement for the special meeting of LSB stockholders and may also file other relevant documents with the SEC regarding the proposed transaction. This communication is not a substitute for the proxy statement or any other document that LSB may file with the SEC. The definitive proxy statement (if and when available) will be mailed to LSB stockholders. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THE PROXY STATEMENT AND ANY OTHER RELEVANT DOCUMENTS THAT MAY BE FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, CAREFULLY AND IN THEIR ENTIRETY IF AND WHEN THEY BECOME AVAILABLE BECAUSE THEY CONTAIN OR WILL CONTAIN IMPORTANT INFORMATION ABOUT LSB AND THE PROPOSED TRANSACTION. Investors and security holders will be able to obtain free copies of the proxy statement (if and when available) and other documents containing important information about LSB and the proposed transaction, once such documents are filed with the SEC through the website maintained by the SEC at http://www.sec.gov. Copies of the documents are filed with the SEC through the website at www.lsbindustries.com or by contacting Michael Foster, General Counsel and Secretary by email at mfoster@lsbindustries.com or by phone at 405-510-3596

Participants in the Solicitation

LSB and certain of its directors and executive officers may be deemed to be participants in the solicitation of proxies in respect of the proposed transaction. Information about the directors and executive officers of LSB, including a description of their direct or indirect interests, by security holdings or otherwise, is set forth in LSB's proxy statement for its 2021 Annual Meeting of Stockholders, which was filed with the SEC on April 19, 2021, and LSB's Annual Report on Form 10-K for the fiscal year ended December 31, 2020, which was filed with the SEC on February 25, 2021. Other information regarding the participants in the proxy solicitations and a description of their direct and indirect interests, by security holdings or otherwise, will be contained in the proxy statement and other relevant materials to be filed with the SEC regarding the proposed transaction when such materials become available. Investors should read the proxy statement carefully when it becomes available before making any voting or investment decisions. You may obtain free copies of these documents from LSB using the sources indicated above.

Company Contact:

Mark Behrman, President & CEO Cheryl Maguire, Executive Vice President & CFO (405) 235-4546

Investor Contact: The Equity Group Inc.

Fred Buonocore, CFA (212) 836-9607 Mike Gaudreau (212) 836-9620

LSB Industries Transaction Summary and Rationale (NYSE: LXU)

July 2021













Transaction Summary

- LSB Industries, Inc. ("LSB" or the "Company") is seeking to exchange its existing preferred stock held by Eldridge into common stock, with Eldridge, becoming a majority shareholder of LSB with ~ 60% ownership
- Eldridge will be exchanging at a 30-day volume weighted average price (VWAP) common stock price of \$6.16 per share
- As a part of the transaction, existing stockholders will receive meaningful economic consideration, in the form of a special common stock dividend of 0.30 shares for every 1 share owned
- This transaction will help to facilitate the refinancing of LSB's senior secured notes on more attractive terms, placing the Company more in line with its public company peers and lowering the Company's overall cost of capital and leverage profile
- The Transaction is expected to close in early September 2021

L\$B

Key Transaction Rationale

- Eliminates the current financial impact and repayment of the accrued compounding preferred stock and future accruing dividends at 14.5% (increasing to 16.0% in April 2023) unburdening the Company and unlocking stockholder value
- The Special Committee, Board of Directors and LSB management believe this transaction should lead to a ratings upgrade, and expect that this will allow the Company to refinance its senior secured notes at a lower interest rate and on improved terms, which would reduce its annual cash interest expense and overall cost of capital
- Improves the Company's financial flexibility allowing it to pursue organic growth initiatives, including in green ammonia and clean energy, and accretive M&A opportunities
- Preserves the Company's significant tax attributes, including \$620 million of federal net operating losses, thereby protecting potentially significant future cash savings and stockholder value

L\$B

Pro-Forma Capitalization

Pro Forma Capitalization⁽¹⁾

(\$ M)	As of 3/31/21	Adjustments	Pro Forma
Cash , Cash Equivalents & Restricted Cash	\$14	(\$9)	\$5
9.625% Senior Secured Notes due 2023	\$435		\$435
Other Debt	\$44		\$44
Total Debt	\$479		\$479
Preferred Stock	\$289	(\$289)	
Market Capitalization ⁽²⁾	\$188	\$351 ⁽³⁾⁽⁴⁾	\$539 ⁽³⁾⁽⁴⁾
Total Capitalization	\$957		\$1,019
TTM Adj. EBITDA	\$67	Commitm improve TTM	
Total Debt + Preferred / TTM Adj. EBITDA	11.5x	EBITDA to red	duce debt
Net Debt + Preferred / TTM Adj. EBITDA	11.2x		7.1x



igures may not sum correctly due to rounding. Excludes Series II & D preferred stock of \$3.0mm. Market capitalization calculated based on 30-day VM

I based on 30-day VWAP of \$6.16 as of 7/16/21 and 30.6 million shares outstanding, including restricted stock, shareholder equity as a result of special common stock kividend to existing stockholders of 0.30 shares for every one share owne (Stock as Taxoutisian environment of \$1571 2 million is of 7/10/2013).



Reducing Leverage Profile with Target Leverage of 4x or Less

- The balance sheet is immediately simplified with the removal of the preferred stock, and leverage multiples drop to 7.1x⁽²⁾⁽³⁾ (PF Q1 2021) and 4.5x based on the consensus analyst 2021 EBITDA estimates of \$106 million
- LSB is committed to continue strengthening its balance sheet to bolster its credit metrics and reduce its debt leverage as adjusted EBITDA improves throughout the balance of 2021
- No cash consideration paid in the conversion of the preferred stock, avoiding the burden of additional debt and allowing leverage to be accessed for other strategic initiatives



rer may not sum correctly due to nounding. geterm debt, including current portion and excluding OID. Preferred stock balance represents liquidation preference 2) Defined as (Total Debt + Preferred Stock - Cash) / Adjusted EBITDA. Excludes Series B & D prefer includes Series F3 participation rights based on the last day closing share price of each respective period. Excludes 3) See slide 10 for Adjusted (BITDA reconciliation. ies B & D preferred stock of \$3.0mm. 4) Based on consensus analyst EBITDA estimates of \$106 million.

Exchange Provides Financial Flexibility to Handle Commodity Price Volatility

(\$Millions)

Illustrative Adjusted EBITDA Sensitivity Grid at natural gas cost of \$3.00/mmbtu

			Trough Pr	ice			Peak Pri	ce
			Ave	rage UAM	Price/S	T For A	Year	Key Factors:
			\$130	\$150	\$170	\$190	\$210	 Assumes no turnarounds at any facility and the related
th Price	ar a	\$200	\$60	\$70	\$80	\$90	\$100	impact from lost production/sales and
Trough	Ammonia A Year	\$250	\$80	\$90	\$100	\$110	\$120	 additional expenses Average ammonia plant on-
	Tampa / MT For	\$300	\$100	\$110	\$120	\$130	\$140	stream rate of 97%, 95% and 95% at El Dorado, Cherokee
Price	Average Tam Price/MT	\$350	\$120	\$130	\$140	\$150	\$160	and Pryor, respectively • Every \$0.10 change in natural
Peak	Aver	\$400	\$140	\$150	\$160	\$170	\$180	gas cost is ~\$2.0-\$2.5 million (depending on product mix) impact to adjusted EBITDA

- LSB's financial performance is directly linked to commodity cycles, with LSB adjusted EBITDA closely correlating to the ammonia and UAN peak and trough selling prices
- Per the above sensitivity grid, LSB's annual adjusted EBITDA can range between approximately \$60 million to \$180 million at trough and peak pricing environments
- Given the preferred stock has a 14.5% annual dividend, increasing to 16.0% by April 2023, this high dividend rate creates a burden on the Company and limits its opportunities to deliver value to common stockholders as the holder of the preferred stock takes no general economy, nitrogen market or other risks
- In a lower selling price environment, PIK accretion will continue to detract from fundamental equity value and would only accelerate when the dividend rate step-up occurs in April 2023

L\$B

Preferred Stock Balance Only Partially Redeemed at Peak Pricing Cycle

	_		Scenario 1: Trough Pricing Cycles										ycles	Scenario 3: Peak Pricing Cycles					
	LT	M 3/21	Ye	ar 1	Y	ear 2	Y	oar 3	Year 1	Y	ear 2	۲	'ear 3	Ľ	'ear 1	Y	ear 2	Y	ear 3
Adjusted EBITDA			\$	60	\$	70	\$	80	\$ 110	\$	120	\$	130	\$	160	\$	170	\$	180
(-) Interest Expense ⁽¹⁾				(50)		(50)		(50)	(40)		(40)		(40)		(40)		(40)		(40)
(-) Capex				(30)		(30)		(30)	(30)		(30)		(30)		(30)		(30)		(30)
Free Cash Flow			\$	(20)	\$	(10)	\$		\$ 40	\$	50	\$	60	\$	90	\$	100	\$	110
Beginning Preferred Stock Balance ⁽²⁾			\$	289	\$	331	\$	380	\$ 289	\$	307	\$	319	ş	289	s	257	\$	212
(+) Preferred Stock Accretion				42		49		61	42		46		51		42		38		34
(-) Preferred Stock Paydown ⁽³⁾				-		-		•	(24)		(34)		(44)		(74)		(84)		(94)
Ending Preferred Stock Balance	\$	289	\$	331	\$	380	\$	441	\$ 307	\$	319	\$	326	\$	257	\$	212	\$	151

 Under the trough commodity pricing scenarios, LSB financial performance would be negatively impacted, and, as such, the Company may not have enough cash available to pay both the interest on the debt and the preferred stock dividend and, therefore, preferred stock balance would continue to grow

The preferred stock will continue to amplify the capital structure risk, negatively impacting the common
equity value, especially during trough pricing cycles

 In the best case scenario at peak pricing cycle and with lower financing cost on current outstanding debt from a refinance, LSB would not be able to earn its way out of the current capital structure

g in the mid-cycle and peak pricing cycles scenario. pation rights value based on 30 day trailing VWAP of 56.16 as of 7/15/21. Assumes the ne of each available for debt ensumed to 881 availability is enabler than follower.

Next Steps

- Receive approval from shareholders to consummate the exchange of the Eldridge preferred stock into LSB common stock
- Take advantage of issuer friendly credit market environment to refinance the Company's existing debt
- Continue to improve operations within a favorable fertilizer market backdrop and the recovery of industrial demand exiting the pandemic
- Deliver improved financial results that will continue to bolster the Company's credit metrics and achieve target leverage of 4x or below
- Pursue margin enhancing business developments opportunities
- Seek new projects to drive further improvement in operating performance and expand production, storage, and logistics capabilities
- Pursue accretive M&A opportunities

L\$B



LSB

EBITDA Reconciliation

					LTM	Adjustments
		2018	2019	2020	3/31/21	1. Stock-based compensation: Non-cash
	Net Loss	(72)	(63)	(62)	(56)	adjustment for options and restricted stock unit expense recognized in each period
	Interest expense	43	46	51	50	 Noncash loss (gain) on natural gas contracts: Represents non-cash impact of natural gas
	Loss on extinguishment of debt	6				contracts
	Depreciation and amortization	73	70	71	70	 Legal fees (Leidos): Legal fees and expenses related to litigation and recovery efforts against Leidos
	Provision (benefit) for income taxes	2	(21)	(5)	(4)	4. Loss (gain) on disposal of assets: Non-cash
Re	ported EBITDA	51	32	55	60	impact of assets sold or taken out of service during the period
	Stock-based compensation	8	2	2	2	5. FMV adjustment on preferred stock embedded derivatives: Unrealized loss from
2	Noncash loss (gain) on natural gas contracts			1	(1)	embedded derivatives: Unrealized loss from the change in fair value associated with the embedded derivative included in the Series E
3	Legal fees (leidos)	5	10	6	3	Preferred stock
•	Loss (gain) on disposal of assets and other	(2)	11	1	1	 Consulting costs associated with initiatives: Temporary costs related to current operational improvement initiatives
6	FMV adjustment on preferred stock embedded derivatives	(1)	(1)	(0)	1	7. Turnaround Costs: As LSB does not capitalize
6	Consulting costs associated with initiatives	4	1	1	(0)	turnaround costs, represents the effect of amortizing turnaround expenses over the
0	Turnaround costs	10	13	0	0	period of benefit
8	Severance	3	1			 Severance: During 2018 and 2019, the Company incurred costs of approximately \$2.6 million and \$0.6 million respectively,
Ac	justed EBITDA	78	69	65	67	primarily related to corporate restructuring

L\$B

Preferred Stock Reconciliation

				Current				
	12/31/20			3/31/21	6/30/21			7/19/21
Beginning Preferred Balance	\$	268.7	\$	278.0	\$	287.5	\$	297.7
(+) Preferred Stock Periodic Dividend Accrual		9.3		9.5		10.2		2.2
Ending Preferred Stock Balance	\$	278.0	\$	287.5	\$	297.7	\$	299.9
Participation Rights Value ⁽¹⁾		1.9		1.9		1.9		1.9
Total Preferred Stock Balance	\$	279.9	\$	289.4	\$	299.6	\$	301.7

rd stock payments are scheduled to occur on May 1" and November 1" of each year. eriods is calculated using 30-day trailing VWAP of \$6.16 as of 7/16/21 and 303,646 participant shares. Agreement Date

Note: Preferred stock bi 1) For purposes of a

Safe Harbor Statement

Forward Looking Statements

- Statements in this release that are not historical are forward-looking statements within the meaning of the U.S. Private Securities Litigation Reform Act of 1995. These forward-looking statements, which are subject to known and unknown risks, uncertainties and assumptions about us, may include projections of our future financial performance including the effects of the COVID-19 pandemic and anticipated performance based on our growth and other strategies and anticipated trends in our business. These statements are only predictions based on our current expectations and projections about future events. There are important factors that could cause our actual results, level of activity, performance or actual achievements to differ materially from the results, level of activity, performance or anticipated achievements expressed or implied by the forward-looking statements. Significant risks and uncertainties may relate to, but are not limited to, our ability to consummate the exchange transaction on the terms described herein or at all, business and market disruptions related to the COVID-19 pandemic, market conditions and price volatility for our products and feedstocks, as well as global and regional economic downturns, including as a result of the COVID-19 pandemic, that adversely affect the demand for our end-use products; disruptions in production at our manufacturing facilities; and other financial, economic, competitive, environmental, political, legal and regulatory factors. These and other risk factors are discussed in the Company's filings with the Securities and Exchange Commission (SEC).
- Moreover, we operate in a very competitive and rapidly changing environment. New risks and uncertainties emerge from time to time, and it is not possible for our management to predict all risks and uncertainties, nor can management assess the impact of all factors on our business or the extent to which any factor, or combination of factors, may cause actual results to differ materially from those contained in any forward-looking statements. Although we believe the expectations reflected in the forward-looking statements are reasonable, we cannot guarantee future results, level of activity, performance or achievements. Neither we nor any other person assumes responsibility for the accuracy or completeness of any of these forward-looking statements. You should not rely upon forward-looking statements as predictions of future events. Unless otherwise required by applicable laws, we undertake no obligation to update or revise any forward-looking statements, whether because of new information or future developments.

Non-GAAP Financial Measures

This presentation includes certain non-GAAP financial measures, including adjusted EBITDA and adjusted gross profit margin, which are provided to assist in an understanding of our business and its performance. These non-GAAP financial measures should be considered only as supplemental to, and not as superior to, financial measures prepared in accordance with GAAP. Non-GAAP financial measures should be read only in conjunction with consolidated financial statements and other financial information prepared in accordance with GAAP. Reconciliations of non-GAAP measures to the relevant GAAP measures are provided in the appendix of this presentation.

L\$B

Safe Harbor Statement (continued)

No Offer or Solicitation

This presentation is not intended to and shall not constitute an offer to buy or sell or the solicitation of an offer to buy or sell any securities, or a solicitation of any vote or approval, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made, except by means of a prospectus meeting the requirements of Section 10 of the U.S. Securities Act of 1933, as amended, or an exemption from the registration requirements thereof.

Additional Information about the Exchange Transaction and Where to Find It

In connection with the proposed transaction, LSB intends to file with the SEC a proxy statement for the special meeting of LSB stockholders and may also file other relevant documents with the SEC regarding the proposed transaction. This communication is not a substitute for the proxy statement or any other document that LSB may file with the SEC. The definitive proxy statement (if and when available) will be mailed to LSB stockholders. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THE PROXY STATEMENT AND ANY OTHER RELEVANT DOCUMENTS THAT MAY BE FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, CAREFULLY AND IN THEIR ENTIRETY IF AND WHEN THEY BECOME AVAILABLE BECAUSE THEY CONTAIN OR WILL CONTAIN IMPORTANT INFORMATION ABOUT LSB AND THE PROYOSED TRANSACTION. Investors and security holders will be able to obtain free copies of the proxy statement (if and when available) and other documents containing important information about LSB and the proposed transaction, once such documents are filed with the SEC through the website maintained by the SEC at http://www.sec.gov. Copies of the documents field with the SEC through the website maintained by the SEC at http://www.sec.gov. Copies of the documents field with the SEC through the website maintained by the SEC at http://www.sec.gov. Copies of the documents field with the SEC through the website maintained by the SEC at http://www.sec.gov. Copies of the documents field with the SEC through the website maintained by the SEC at http://www.sec.gov. Copies of the documents field with the SEC through the website maintained by the SEC at http://www.sec.gov. Copies of the documents field with the SEC by LSB may be obtained free of charge on LSB's website at www.lsbindustries.com or by contacting Michael Foster, General Counsel and Secretary by email at mfoster@lsbindustries.com or by phone at 405-510-3596

Participants in the Solicitation

LSB and certain of its directors and executive officers may be deemed to be participants in the solicitation of proxies in respect of the proposed transaction. Information about the directors and executive officers of LSB, including a description of their direct or indirect interests, by security holdings or otherwise, is set forth in LSB's proxy statement for its 2021 Annual Meeting of Stockholders, which was filed with the SEC on April 19, 2021, and LSB's Annual Report on Form 10-K for the fiscal year ended December 31, 2020, which was filed with the SEC on February 25, 2021. Other information regarding the participants in the proxy solicitations and a description of their direct and indirect interests, by security holdings or otherwise, will be contained in the proxy statement and other relevant materials to be filed with the SEC regarding the proposed transaction when such materials become available. Investors should read the proxy statement carefully when it becomes available before making any voting or investment decisions. You may obtain free copies of these documents from LSB using the sources indicated above.

